UNITED STATES BANKRUPTCY COURT DISTRICT OF CONNECTICUT BRIDGEPORT DIVISION

In re:	Chapter 11
AFFINITY HEALTHCARE MANAGEMENT, INC., ET AL ¹ Debtors.	Case Nos. 16-30043 through 16-30047 (JAM) Jointly Administered
AFFINITY HEALTHCARE MANAGEMENT,)
INC., ET AL)
Movant)
V.	Re: ECF No. 6
REVENUE MANAGEMENT SOLUTIONS, LLC;)
STATE OF CONNECTICUT DEPARTMENT OF	ý
SOCIAL SERVICES;)
STATE OF CONNECTICUT DEPARTMENT OF)
REVENUE SERVICES; AND	ý
STATE OF CONNECTICUT DEPARTMENT OF)
LABOR)
Respondents	,

FOURTEENTH INTERIM ORDER (A) APPROVING SALE OF PROVIDER ACCOUNTS RECEIVABLE TO REVENUE MANAGEMENT SOLUTIONS, LLC PURSUANT TO §§ 363(B) AND (F) OF THE BANKRUPTCY CODE; (B) GRANTING FIRST-PRIORITY SECURITY INTERESTS ON PURCHASED ACCOUNTS; (C) AUTHORIZING INDEBTEDNESS WITH ADMINISTRATIVE SUPER-PRIORITY AND SECURED BY LIENS ON AND SECURITY INTERESTS IN NON-PURCHASED ACCOUNTS AND ON ALL OTHER ASSETS OF THE PROVIDERS PURSUANT TO §§ 364(C) AND (D) OF THE BANKRUPTCY CODE; (D) AUTHORIZING USE OF CASH COLLATERAL AND (E) GRANTING RELATED RELIEF

Upon the motion (the "Motion") of Affinity Health Care Management, Inc. (the "Lead Debtor"), Health Care Investors, Inc. d/b/a Alexandria Manor, Health Care Alliance, Inc. d/b/a Blair Manor, Health Care Assurance, LLC d/b/a Douglas Manor, and Health Care Reliance, LLC d/b/a Ellis Manor (collectively, the "Debtors"), the debtors and debtors-in-possession, in the

¹ Affinity Healthcare Management, Inc., Case No. 16-30043; Health Care Investors, Inc., Case No. 16-30044; Health Care Alliance, Inc., Case No. 16-30045; Health Care Assurance, LLC, Case No. 16-30046; and Health Care Reliance, LLC, Case No. 16-30047.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 2 of 102

above-referenced chapter 11 cases (collectively, the "Cases"), by their undersigned counsel, pursuant to §§ 105, 361, 362, 363, 364, and 507 of the United States Bankruptcy Code, 11 U.S.C. §§ 101 et seq. ("Bankruptcy Code") and Rules 2002, 4001 and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), seeking, inter alia, entry of an interim order (this "Interim Order") that:

(a) authorizes and obligates the Health Care Investors, Inc. d/b/a Alexandria Manor, Health Care Alliance, Inc. d/b/a Blair Manor, Health Care Assurance, LLC d/b/a Douglas Manor, and Health Care Reliance, LLC d/b/a Ellis Manor (individually and collectively, jointly and severally, the "Providers") to affirm, undertake and continue to perform all of their respective obligations under that certain Purchase Agreement (the "Purchase Agreement," a true and genuine copy of which is attached hereto as **Exhibit A**) made as of May 25, 2012, by and between the Providers and Revenue Managements Solutions, LLC or its permitted successors and assigns² ("RMS"), as "Buyer"³, wherein the four (4) Providers, each of which operate a skilled nursing facility as aforesaid, shall continue, jointly and severally, individually and collectively, as set out in the Purchase Agreement, to offer to sell and RMS may purchase, at its discretion, certain healthcare accounts receivables of the Providers (the "Post-Petition Purchased Accounts") together with all books, records, billing and credit files, an irrevocable paid-up license to use related medical and patient records, chattel paper, documents, instruments and general intangibles related to the Post-Petition Purchased Accounts and certain other assets necessary for RMS or its agents to collect the Post-Petition Purchased Accounts, all as set forth in the Purchase Agreement (collectively, with the Purchased Accounts, the "Post-Petition

² Funding affiliates of RMS (owned directly or indirectly by RMS or its principals) are permitted assigns under the Purchase Agreement.

³ Terms with initial capitalization herein have the same meaning as provided for in the Purchase Agreement, unless otherwise defined herein.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 3 of 102

Conveyed Property"), on the terms and conditions set forth in the Purchase Agreement and the other documents and instruments provided for therein and related thereto (collectively, with the Purchase Agreement, the "Funding Documents"), such purchases and sales being made free and clear of liens, claims, interests and encumbrances, under §§ 363(b) and (f) of the Bankruptcy Code;

- (b) grants RMS a first-priority security interest in the Post-Petition Purchased Accounts and in the proceeds thereof pursuant to § 1-201(35) of the Uniform Commercial Code of the State of Connecticut, Conn. Gen. Stat. § 42a-1-101 et seq., to secure all of its right, title and interest in the Post-Petition Purchased Accounts and the related Post-Petition Conveyed Property (the "Post-Petition Ownership Interests");
- (c) authorizes the Providers to obtain Initial Installments, Subsequent Installments and other financial accommodations and post-petition funding from RMS pursuant to the terms and conditions of the Funding Documents and this Interim Order (the "Post-Petition Funding Facility");
- (d) approves of the terms of this Interim Order approving the Post-Petition Funding Facility in the form and substance satisfactory to RMS, at its sole and absolute discretion;
- (e) authorizes the Providers to grant RMS assurances for the Providers' full and timely payment by and performance of all Obligations owed by the Providers to RMS arising under, out of, or in connection with the Funding Documents, by granting RMS (i) pursuant to § 364(c)(1) of the Bankruptcy Code, a super-priority administrative expense claim allowable under § 503(b) of the Bankruptcy Code having priority over any and all expenses and claims specified in any other section of the Bankruptcy Code including, without limitation, §§ 503(b) and 507(b) of the Bankruptcy Code, subject only to the Carve-Out (as hereinafter defined), and

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 4 of 102

(ii) pursuant to § 364(d) of the Bankruptcy Code, a valid, enforceable, binding and duly perfected continuing first-priority security interest in and to and a first lien upon the Non-Purchased Accounts and the credit balances, and all of the Providers' other rights, property and assets, real, personal, tangible and intangible not sold to RMS under the Purchase Agreement (including, without limitation, inventory, general intangibles, contract rights, purchase orders, documents, instruments, chattel paper, goods, good will, patents, trademarks and other intellectual property, cash, deposit accounts, rights to payment, furniture, fixtures, machinery, equipment, lease rights, land and other real property, causes of action, commercial tort claims, and all other "Collateral" defined in the Security Agreement which definition is incorporated herein by reference) wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof, subject only to the Carve-Out; provided, however, that the DIP Collateral (as defined below) shall not include any avoidance or similar actions of the Providers arising under §§ 542, 544, 545, 547, 548, 550 and 551 of the Bankruptcy Code ("Avoidance Actions");

(f) authorizes and directs the Providers to pay all Obligations of every nature of the Providers now or hereafter existing under or arising out of or in connection with the Purchase Agreement, including, without limitation, the Outstanding Initial Installments, the Funding Fee, Buyer's out-of-pocket costs and expenses, the indemnification obligations, the Lock Box Fees, wire transfer fees, and other amounts payable under the Funding Documents as they become due, all to the extent provided by and in accordance with the terms of the respective Funding Documents;

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 5 of 102

- (g) authorizes the Debtors' continued use of "cash collateral" as that term is defined by Bankruptcy Code § 363(a) ("Cash Collateral") on the terms and conditions set forth in this Interim Order;
- (h) authorizes the Debtors to provide adequate protection of the liens and security interests of RMS and the State of Connecticut Department of Labor ("DOL");
- (i) provides adequate protection to RMS (to the extent any Prepetition Obligations, as defined below, remain outstanding) and the DOL for any Diminution in Value (as defined herein) of their respective interests in the Prepetition Collateral (as defined herein), including the Cash Collateral;
- (j) vacates and modifies the automatic stay imposed by section 362 of the Bankruptcy Code to the extent necessary to implement and effectuate the terms and provisions of the Funding Documents and this Interim Order;
 - (k) provides for other related relief.

The Court having considered the Motion, the exhibits attached thereto, the Funding Documents, and the evidence submitted or adduced and the arguments of counsel made at the interim hearing held on January 11, 2017 (the "Continued Interim Hearing"); and notice of the Continued Interim Hearing having been given in accordance with Bankruptcy Rules 2002, 4001(b), (c) and (d), and 9014; RMS having agreed to provide post-petition funding by the purchase of Accounts and the payment of Initial Installments and Subsequent Installments, and other financial accommodations to the Providers on the terms of the Funding Documents and this Interim Order; and the Continued Interim Hearing to consider the interim relief requested in the Motion having been held and concluded; and all objections, if any, to the interim relief requested in the Motion having been withdrawn, resolved or overruled by the Court; and it appearing to the

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 6 of 102

Court that granting the interim relief requested is necessary and otherwise is fair and reasonable and in the best interests of the Debtors, their estates, and their creditors and equity holders, and is essential for the continued operation of the Debtors' businesses; and after due deliberation and consideration, and for good and sufficient cause appearing therefor:

NOW, THEREFORE, upon the record of the Continued Interim Hearing and after due deliberation, good and sufficient cause appearing therefore, the Court hereby makes the following **FINDINGS OF FACT and CONCLUSIONS OF LAW**:

- A. <u>Petition</u>. On January 13, 2016, (the "Petition Date") the Debtors each filed a voluntary petition for relief under Chapter 11 of title 11 of the United States Code (the "Bankruptcy Code"). The Debtors continue to possess their assets and manage and operate their businesses as debtors-in-possession pursuant to §§ 1107 and 1108 of the Bankruptcy Code.
- B. <u>Jurisdiction</u>. This Court has jurisdiction over this proceeding and the parties and property affected hereby pursuant to 28 U.S.C. §§157 and 1334. The Motion is a "core" matter as defined in 28 U.S.C. §§ 157(b)(2)(A), (D), (G), (K), (M) and (N). Venue is proper in this Court pursuant to 28 U.S.C. §§ 1408 and 1409.
- C. <u>The Providers' Business</u>. The Providers operate four (4) skilled nursing facilities in Connecticut, as aforesaid, and the Lead Debtor manages the business affairs of the Providers.
- D. <u>Statutory Committee</u>. On January 25, 2016, the United States Trustee (the "U.S. Trustee) appointed an official committee of unsecured creditors in these Cases pursuant to section 1102 of the Bankruptcy Code (the "<u>Statutory Committee</u>").
- E. <u>HUD</u>. The United States Department of Housing and Urban Development ("HUD") asserts a first priority security interest in the personal property of the Providers, which the Debtors and RMS dispute.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 7 of 102

- F. <u>Debtors' Stipulations</u>. After consultation with their attorneys and financial advisors, the Debtors (on behalf of and for themselves and their estates) admit, stipulate, acknowledge and agree that (collectively, paragraphs E(i) through E(vii) below are referred to herein as the "Debtors' Stipulations"):
- (i) Prepetition Funding Facility: Pursuant to the Purchase Agreement and other Funding Documents, prior to the Petition Date, the Providers, jointly and severally, individually and collectively, as set out in the Purchase Agreement, offered to sell and RMS purchased certain healthcare accounts receivables of the Providers (the "Prepetition Purchased Accounts," and together with the Post-Petition Purchased Accounts, the "Purchased Accounts") together with all books, records, billing and credit files, an irrevocable paid-up license to use related medical and patient records, chattel paper, documents, instruments and general intangibles related to the Prepetition Purchased Accounts and certain other assets necessary for Buyer or its agents to collect the Prepetition Purchased Accounts, all as set forth in the Purchase Agreement (collectively, with the Prepetition Purchased Accounts, the "Prepetition Conveyed Property," and, together with the Post-Petition Conveyed Property, the "Conveyed Property"), on the terms and conditions set forth in the Purchase Agreement and the other Funding Documents (the "Prepetition Funding Facility," and together with the Post-Petition Funding Facility, the "Funding Facility"):
- (ii) <u>Prepetition Obligations</u>: The Purchase Agreement provided that the aggregate amount of the Outstanding Initial Installments plus any other outstanding Obligations of the Providers shall not (except as permitted by Buyer in its sole discretion) exceed \$2,500,000 (the "Facility Cap"). As of the Petition Date, the aggregate amount of the Outstanding Initial Installments due and payable pursuant to the Prepetition Funding Facility was \$1,450,000

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 8 of 102

(together with any amounts paid, incurred or accrued prior to the Petition Date in accordance with the Funding Documents, any fees, expenses, and disbursements (including, without limitation, attorneys' fees, related expenses and disbursements), indemnification obligations, all obligations of the Lead Debtor as provided in the Lead Debtor Guaranty (as defined below), and other Obligations of the Debtors to RMS of whatever nature, whether or not contingent, whenever arising, due or owing in respect thereof, the "Prepetition Obligations").

(iii) <u>Prepetition Guaranties</u>:

- (a) Pursuant to that certain Guaranty dated May 25, 2012 (as amended, supplemented, restated or otherwise modified prior to the Petition Date, the "Lead Debtor Guaranty"), by and among the Lead Debtor and RMS, the Lead Debtor, *inter alia*, guaranteed the obligations of the Providers under the Purchase Agreement. The Lead Debtor granted a security interest in and lien upon (the "Lead Debtor Liens") all of its assets to RMS to secure it obligations under the Lead Debtor Guaranty including its guarantee of the Providers' obligations under the Purchase Agreement.
- (b) Pursuant to that certain Guaranty dated May 25, 2012 (as amended, supplemented, restated or otherwise modified prior to the Petition Date, the "Fischman Guaranty"), by and among Benjamin Z. Fischman ("Fischman") and RMS, Fischman, *inter alia*, guaranteed the obligations of the Providers under the Purchase Agreement.
- (c) Pursuant to that certain Guaranty dated May 25, 2012 (as amended, supplemented, restated or otherwise modified prior to the Petition Date, the "Strasser Guaranty"), by and among Samuel Strasser ("Strasser") and RMS, Strasser, inter alia, guaranteed the obligations of the Providers under the Purchase Agreement.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 9 of 102

- (iv) Prepetition Liens and Prepetition Collateral. Pursuant to that certain Security Agreement dated as of May 25, 2012 (as amended, supplemented, restated or otherwise modified prior to the Petition Date, the "Security Agreement"), by and between the Buyer and the Providers, and as more fully set forth in the other Funding Documents, to secure all of the Obligations of every nature of the Providers then or thereafter existing under or arising out of or in connection with the Purchase Agreement, the Providers granted to RMS a continuing security interest in and liens upon (the "Prepetition Liens") their respective right, title and interest in all of the properties, assets and rights of the Providers, wherever located, whether now owned or thereafter acquired or arising, and all proceeds and products thereof (the "Prepetition Collateral," and together with the DIP Collateral, the "Collateral"). The Prepetition Liens are validly perfected and prior in right, title and interest to any other lien, encumbrance or interest in the Prepetition Collateral.
- Obligations. The Debtors (for themselves and their estates) and RMS acknowledge and agree that: (a) as of the Petition Date, the Providers had absolutely sold, transferred and conveyed and RMS had absolutely purchased and acquired all right, title and interest in the Prepetition Purchased Accounts and the other Prepetition Conveyed Property, on the terms and conditions set forth in the Purchase Agreement and the other Funding Documents, and that the transfer of ownership and title to the Prepetition Purchased Accounts and the other Prepetition Conveyed Property from the Providers to RMS constituted a "true sale" of the Prepetition Conveyed Property for fair value and is not subject to re-characterization as a loan or a financing; (b) as of the Petition Date, the Prepetition Liens on the Prepetition Collateral are valid, binding, enforceable, non-avoidable and properly perfected; (b) as of the Petition Date, the Prepetition

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 10 of 102

Liens were senior in priority over any and all other liens on the Prepetition Collateral, subject only to certain liens otherwise permitted by the Prepetition Funding Documents (to the extent any such permitted liens are valid, properly perfected, non-avoidable and senior in priority to the Prepetition Liens as of the Petition Date, the "Permitted Prior Liens"); (c) the Prepetition Obligations constitute legal, valid, binding, and non-avoidable obligations of the Debtors; (d) no offsets, challenges, objections, defenses, claims or counterclaims of any kind or nature to any of the Prepetition Liens or Prepetition Obligations exist, and no portion of the Prepetition Liens or Prepetition Obligations is subject to any challenge or defense, including, without limitation, avoidance, disallowance, disgorgement, re-characterization, or subordination (whether equitable or otherwise) pursuant to the Bankruptcy Code or applicable non-bankruptcy law; (f) the Debtors and their estates have no claims, objections, challenges, causes of actions, and/or choses in action, including without limitation, avoidance claims under chapter 5 of the Bankruptcy Code, against RMS or any of its affiliates, agents, attorneys, advisors, professionals, officers, directors and employees; (g) as of the Petition Date, the value of the Prepetition Collateral securing the Prepetition Obligations exceeded the amount of those obligations, and accordingly the Prepetition Obligations are allowed secured claims within the meaning of § 506 of the Bankruptcy Code, in an amount of not less than \$1,450,000 on account of the Outstanding Initial Installments as of the Petition Date together with any and all other amounts of whatever nature owing in respect of such Prepetition Obligations of the Debtors; and (h) any payments made on account of the Prepetition Obligations to or for the benefit of RMS prior to the Petition Date were on account of amounts in respect of which RMS was over-secured, were payments out of the Prepetition Collateral, and therefore such payments did not diminish any property otherwise available for distribution to any other creditors of the Debtors.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 11 of 102

- (vi) <u>Cash Collateral</u>. The Debtors represent that substantially all of the Debtors' cash, including the cash in their deposit accounts, wherever located, whether as original collateral or proceeds of other Prepetition Collateral, constitute the Cash Collateral of RMS; provided, however, that, notwithstanding the forgoing, the Purchased Accounts and their proceeds, wherever located, constitute the exclusive property of RMS pursuant to the terms of the Funding Documents and this Interim Order.
- (vii) <u>Default by the Debtors</u>. The Debtors acknowledge and stipulate that theDebtors are in default of their debts and Obligations under the Funding Documents.

G. Findings Regarding the Post-Petition Funding.

- (i) <u>Post-Petition Funding</u>. The Providers have requested from RMS, and RMS is willing to purchase Accounts and extend to the four (4) Providers, jointly and severally, as one Provider, certain Initial Installments, Subsequent Installments and other financial and funding accommodations, as more particularly described and on the terms and conditions of the Purchase Agreement, the other Funding Documents and this Order.
- (ii) Need for Post-Petition Funding. The Providers do not have sufficient available sources to provide working capital to operate their businesses in the ordinary course without the requested post-petition funding from RMS. The Providers' ability to provide patient services, and maintain their business relationships with vendors, suppliers and employees, and to otherwise fund their operations, are essential to the Providers' viability. There is an immediate need for funding to minimize the disruption of the Providers' business and daily operations, manage and preserve the assets of its bankruptcy estate, provide patient care to existing and future patients and enhance the likelihood of a successful reorganization for the benefit of the Debtors' bankruptcy estates, creditors and other parties-in-interest. Thus, the Post-Petition

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 12 of 102

Funding Facility contemplated by the Motion is necessary, essential and appropriate to prevent immediate and irreparable harm to the Debtors, and their estates, employees and patients. Absent entry of this Interim Order, the Debtors' operations risk being disrupted, resulting in immediate and irreparable harm to the estate and the likelihood of a successful reorganization will be impaired. The sale of the Post-Petition Purchased Accounts to RMS on the terms and conditions set forth in the Funding Documents and this Interim Order is in the best interests of the Debtors' estates, creditors and other parties-in-interest.

The Providers, jointly and severally, individually and (iii) True Sale. collectively, are treated as one "Provider" under the Purchase Agreement. As such, the Provider has agreed to sell, and RMS has agreed to purchase, the Post-Petition Purchased Accounts and the other Post-Petition Conveyed Property, now existing or hereafter arising, together with all proceeds thereof, and to advance funds and provide other financial accommodations to the Provider, upon the entry of this Interim Order, which sales to RMS shall be free and clear of liens, claims, interests and encumbrances under §§ 363(b) and (f), upon the terms and conditions of the Purchase Agreement, attached hereto as **Exhibit A** and the other Funding Documents. To secure all of its right, title and interest in the Post-Petition Purchased Accounts and the related Post-Petition Conveyed Property, and their proceeds, RMS shall be granted a first-priority security interest in the Post-Petition Purchased Accounts and the other Post-Petition Conveyed Property, and in the proceeds thereof, pursuant to § 1-201(35) of the Uniform Commercial Code of the State of Connecticut, Conn. Gen. Stat. § 42a-1-101 et seq. The transfer of ownership and title to the Post-Petition Purchased Accounts and the other Post-Petition Conveyed Property from the Provider to RMS constitutes a "true sale" of the subject property for fair value and is not subject to re-characterization as a loan or a financing.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 13 of 102

- (iv) <u>No Credit Available on More Favorable Terms</u>. In view of the Provider's current financial condition, financing arrangements and capital structure, the Provider has represented to the Court, and the Court finds, that the Provider is unable at this time to obtain sufficient unsecured financing on an administrative priority basis under § 503(b)(1) of the Bankruptcy Code.
- Conditions Precedent. RMS is unwilling to purchase the Post-Petition (v) Purchased Accounts and to provide the Initial Installments and Subsequent Installments on account thereof and any other financial accommodations without being afforded with respect thereto the protections afforded by: (i) a "true sale" finding, as to the Purchased Accounts and the related Conveyed Property; (ii) a sale "free and clear" of liens, claims, interests and encumbrances pursuant to §§ 363(b) and (f), as to the Post-Petition Purchased Accounts and the related Post-Petition Conveyed Property; (iii) the grant, in accordance with § 364(c)(1) of the Bankruptcy Code, of an administrative expense claim allowable under § 503(b) of the Bankruptcy Code with priority over any and all claims specified in any other section of the Bankruptcy Code, including, without limitation, §§ 503(b) and 507(b) of the Bankruptcy Code, subject only to the Carve-Out; and (iv) the grant, in accordance with § 364(d)(1) of the Bankruptcy Code, of a continuing security interest in and lien on the DIP Collateral pledged to RMS by the Providers under the Funding Documents as security for all Obligations arising under the Post-Petition Funding Facility, such lien of RMS being senior in priority to any existing duly perfected liens and security interests in the DIP Collateral of any person (including the DOL) which existed on the Petition Date and any replacement liens they may hold, subject only to the Carve-Out; and (v) the other protections afforded by § 363(m) and § 364(e) of the Bankruptcy Code, and this Interim Order.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 14 of 102

- (vi) Use of Proceeds of the Post-Petition Funding Facility; Payment of Prepetition Obligations. As a condition to RMS' continued funding and the agreement for the use of Cash Collateral, RMS requires, and the Debtors have agreed, that proceeds of the Post-Petition Funding Facility shall be used, in each case in a manner consistent with the terms and conditions of the Funding Documents, and solely for (1) payment of the Prepetition Obligations, (2) working capital and other general corporate purposes, (3) permitted payment of costs of administration of the Cases (subject to the limitations of the Carve Out (as defined herein)), (4) payment of fees and expenses due under the Post-Petition Funding Facility, (5) payment of such prepetition expenses, in addition to the Prepetition Obligations permitted to be so paid in accordance with the consents required under the Funding Documents, and as approved by the Court. Payment of the Prepetition Obligations in accordance with this Interim Order is necessary as RMS will not otherwise consent to providing the Post-Petition Funding Facility, and will not otherwise consent to the use of its Cash Collateral and other Prepetition Collateral or the subordination of their liens to the Carve Out. Such payment will not prejudice the Debtors or their estates, because payment of such amounts is on account of RMS being over-secured.
- (vii) <u>Guaranties</u>. As a condition to RMS' continued funding and agreement for the use Cash Collateral and other Prepetition Collateral, Benjamin Z. Fischman and Samuel Strasser (individually and collectively referred to herein as the "Individual Guarantors") shall consent and agree to the entry of this Interim Order and any further Interim or Final Orders relating to the Post-Petition Funding Facility, and they shall execute and deliver guaranty and reaffirmation agreements to RMS.
- (viii) Adequate Protection. RMS and DOL are each entitled to receive, and they shall receive, adequate protection on account of their respective interests in the Prepetition

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 15 of 102

Collateral, pursuant to sections 361, 362, 363 and 364 of the Bankruptcy Code, for any diminution in the value of their respective interests in the Prepetition Collateral (including Cash Collateral) resulting from, among other things, the subordination to the Carve Out (as defined herein) and to the DIP Liens (as defined herein), the Debtors' use, sale or lease of such Prepetition Collateral, and the imposition of the automatic stay (collectively, and solely to the extent of any such diminution in value, the "Diminution in Value"). Pursuant to sections 361, 363, 364 and 507(b), as adequate protection: (i) RMS will receive (a) adequate protection liens and super-priority claims, (b) current payments fees, costs and expenses and other amounts due under the Funding Documents, and (c) ongoing payment of the reasonable fees, costs and expenses, including, without limitation, legal and other professionals' fees and expenses, of RMS; and (ii) DOL will receive adequate protection liens.

- (ix) <u>Sections 506(c)</u> and <u>552(b)</u>. In light of (i) the Post-Petition Funding Facility and RMS' agreement to subordinate its rights of payment, liens and superpriority/priority claims to the Carve Out; and (ii) DOL's agreement to subordinate its rights of payment, liens and super-priority/priority claims to the Carve Out, DIP Liens and DIP Super-Priority Claims, each of the RMS and DOL are entitled to (a) a waiver of any "equities of the case" claims under section 552(b) of the Bankruptcy Code, and (b) a waiver of the provisions of section 506(c) of the Bankruptcy Code.
- H. <u>Business Judgment and Good Faith</u>. The terms of the Post-Petition Funding Facility taken as a whole are at least as favorable to the Providers as those available from other sources, and are fair, just and reasonable under the circumstances, are ordinary and appropriate for funding of a debtor-in-possession and reflect the Providers' exercise of prudent business judgment, and also as to the Lead Debtor as guarantor of the Providers' obligations and

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 16 of 102

indebtedness to RMS, consistent with the fiduciary duties of the Debtors, supported by reasonably equivalent value and fair consideration, and the Funding Documents are enforceable in accordance with their terms. Consequently, (i) the sale of the Post-Petition Purchased Accounts to RMS and their funding by RMS under §§ 363(b) and (f) of the Bankruptcy Code shall be deemed to have been in good faith, as that term is used in §363(m) of the Bankruptcy Code, and (ii) any other financial accommodations, credit, loans, indebtedness, and liens on the DIP Collateral extended to the Provider by RMS under the terms of this Interim Order and the Funding Documents provided on or after the commencement of these Cases shall be deemed to have been extended in good faith, as that term is defined in § 364(e) of the Bankruptcy Code, in the event that any provision of this Interim Order is vacated, reversed, or modified, on appeal or otherwise.

I. Notice. Notice of the Hearing and of the relief requested in the Motion was given to (i) the United States Trustee for the District of Connecticut; (ii) RMS and its counsel; (iii) the United States Attorney, (iv) the Internal Revenue Service (v) the State of Connecticut Department of Revenue Services ("DRS"); (vi) the State of Connecticut Department of Social Services ("DSS"); (vii) the State of Connecticut Department Labor (viii) the United States Department of Housing and Urban Development; (ix) the Twenty Largest Unsecured Creditors filed in each of the Cases (x) the Office of the Attorney General for the State of Connecticut; and (xi) all other persons entitled to notice under the Federal Rules of Bankruptcy Procedure (collectively, the "Notice Parties"), together with a motion to shorten time. Such notice constitutes good and sufficient notice of the Hearing under the circumstances in accordance with Fed. R. Bankr. P. 4001(b), Fed. R. Bankr. P. 4001(c), Fed. R. Bankr. P. 4001(d) and §102(1) of the Bankruptcy

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 17 of 102

Code, as required by §§ 363(c), 363(e) and 364(e) of the Bankruptcy Code, in light of the emergency nature of the relief requested in the Motion.

BASED UPON the foregoing findings of fact and conclusions of law, and upon the record made before the Court at the Continued Interim Hearing, and good cause therefore appearing, it is hereby **ORDERED**, **ADJUDGED AND DECREED** that:

- 1. <u>Record</u>. The record at the Hearing and the findings of fact and conclusions of law above are incorporated by reference.
- 2. <u>Motion Granted.</u> The Motion is **GRANTED**, and through and including February 11, 2017, the Providers are authorized to obtain funding from RMS under terms of the Purchase Agreement, the other Funding Documents, and this Interim Order, by selling Accounts to RMS and receiving in exchange Initial Installments, Subsequent Installments and other payments and other financial accommodations from RMS on the terms and conditions set forth in the Purchase Agreement and other Funding Documents and this Order.
- 3. <u>Authorization and Approval of Funding Documents; Sale of Accounts, and Indebtedness and other Financial Accommodations Thereunder</u>. The Providers hereby affirm and undertake and shall continue to perform all of their respective obligations under that certain the Purchase Agreement and the other Funding Documents, each of which is hereby approved in all respects. Without limitation thereof:
- (a) The Providers may sell and RMS may purchase Post-Petition Purchased Accounts and associated Post-Petition Conveyed Property, on the terms and conditions of the Purchase Agreement and the other Funding Documents, and RMS may exercise its rights under the Funding Documents, subject to the terms of this Order and of such documents. The sale of Accounts hereunder by the Providers to RMS shall be a "true sale" of the Post-Petition

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 18 of 102

Purchased Accounts which vests all right, title and interest in the Post-Petition Purchased Accounts and the other Post-Petition Conveyed Property sold under this Order, and their proceeds, in RMS and such sale is not subject to re-characterization as a loan or a financing;

- Petition Conveyed Property to RMS under this Interim Order is free and clear of liens, claims, interests and encumbrances pursuant to §§ 363(b), (f) and (m) of the Bankruptcy Code, and RMS shall receive and is hereby granted a first-priority security interest in the Post-Petition Purchased Accounts and in the proceeds thereof pursuant to § 1-201(35) of the Uniform Commercial Code of the State of Connecticut, Conn. Gen. Stat. § 42a-1-101 *et seq.*, to secure all of its right, title and interest in the Post-Petition Purchased Accounts and the related Post-Petition Conveyed Property in the proceeds thereof. Any valid pre-petition liens and security interests on the property sold to RMS at the time of their sale are released by this Interim Order and shall attach to the proceeds of sale, in their respective order of priority;
- (c) The Debtors' are hereby authorized and directed to do and perform all acts and make, execute and deliver all documents that may be necessary or desirable for the Debtors' performance under the Funding Documents, and to execute and deliver and make payments of monies and proceeds to RMS as called for therein;
- (d) The Providers are authorized to incur Obligations and to receive other financial accommodations from RMS under the Funding Documents, and the Lead Debtor is authorized and directed to reaffirm, guarantee and undertake certain other obligations and indebtedness on the terms and conditions set forth in the Lead Debtor Guaranty.
- (e) The Providers may use the DIP Collateral only on the terms and subject to the conditions set forth in this Interim Order and the Funding Documents.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 19 of 102

- (f) As a condition to RMS' continued funding and agreement for the use Cash Collateral and other Prepetition Collateral, the Individual Guarantors have consented and agreed to the entry of this Interim Order relating to the Post-Petition Funding Facility.
- 4. <u>Payments and Transfers to RMS</u>. The Providers are authorized to and shall, consistent with the requirements of the Medicare statutes and regulations, make all payments and transfers of property to RMS or for the benefit of RMS which are contemplated by this Interim Order or may be provided, permitted or required under the Funding Documents. These payments and transfers shall not be subject to avoidance under § 549 or any other provision of the Bankruptcy Code or state law.

5. Lockbox Accounts.

- (a) The Providers and RMS are authorized to maintain and/or establish and operate the Lock Box Accounts required by the Purchase Agreement and the other Funding Documents for the collection of the Provider's Government Receivables and Non-Government Receivables in accordance with the terms and conditions of the Purchase Agreement and the other Funding Documents, and the Providers are authorized and directed to execute, deliver and perform under any and all Lock Box Agreements governing such Lock Box Accounts. The Governmental Lockbox Accounts for the Government Receivables shall be in the name and sole control of the Providers, alone, without limitation of the contractual covenants of the Providers to RMS under the Purchase Agreement.
- (b) Promptly upon entry of this Order, the Providers (and to the extent applicable, RMS) shall take all actions necessary and appropriate to cause all payments on the Providers' Accounts, whether or not purchased by RMS, to be paid into or deposited in the Lock Box Accounts, including, without limitation: (a) giving written notice, as per the terms and

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 20 of 102

conditions of the Purchase Agreement, to all governmental Payors and non-governmental Payors; (b) depositing or transferring into the Lock Box Accounts all cash, checks, or wire transfers representing, in whole or in part, all payments on Purchased Accounts, in accordance with the Funding Documents; and (c) implementing such other actions as are required to effectuate the foregoing.

- 6. <u>Post-Petition Ownership Interests</u>. RMS shall have full right, title and ownership interest in all Post-Petition Purchased Accounts and the other Post-Petition Conveyed Property which are purchased from the Providers under this Interim Order and the Funding Documents with a first priority-perfected security interest therein, as buyer thereof, senior to the ownership and lien rights of the Providers and of any creditor of the Providers. On the terms of the Purchase Agreement and the other Funding Documents, RMS shall receive the proceeds of the Post-Petition Purchased Accounts and the other Post-Petition Conveyed Property, which Post-Petition Ownership Interests of RMS represent the sole property of RMS, as owner, and are not subject to re-characterization as collateral for a loan or a financing.
- 7. <u>Liens And Super-Priority Administrative Claim as Collateral Security</u>. As collateral security for the full and timely performance of the Obligations of the Providers to RMS under the Purchase Agreement and the other Funding Documents, RMS is hereby indefeasibly granted:
- (a) Pursuant to § 364(d) of the Bankruptcy Code, a valid, enforceable, binding and duly perfected continuing first-priority security interest in and to and a first lien upon (the "DIP Liens") the Non-Purchased Accounts and the credit balances, and all of the Providers' other rights, property and assets, real, personal, tangible and intangible not sold to RMS under the Purchase Agreement (including, without limitation, inventory, general intangibles, contract

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 21 of 102

rights, purchase orders, documents, instruments, chattel paper, goods, good will, patents, trademarks and other intellectual property, cash, deposit accounts, rights to payment, furniture, fixtures, machinery, equipment, lease rights, land and other real property, causes of action, commercial tort claims, and all other "Collateral" defined in the Security Agreement which definition is incorporated herein by reference) wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof (collectively, the "DIP Collateral"), subject only to the Carve-Out.

- (b) A super-priority administrative expense claim under § 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses, whether heretofore or hereafter incurred, specified in any other section of the Bankruptcy Code, including, without limitation, §§ 503(b) and 507(b) of the Bankruptcy Code (the "DIP Super-Priority Claim"), subject only to the Carve-Out.
- (c) Nothing in this Order or in the Funding Documents shall be deemed to grant RMS any claims, causes of actions or recoveries from any "avoidance actions" under §§ 542, 544, 545, 547, 548, 550 and 553 of the Bankruptcy Code, and the Avoidance Actions are specifically excluded from the DIP Collateral, and RMS shall not receive any recover on account of its DIP Liens or its DIP Super-Priority Claim from any proceeds of the Avoidance Actions.
- (d) Prior to any Event of Default under the Purchase Agreement, the Providers may pay professionals court-approved compensation and expense reimbursements in the ordinary course and as may be allowed and payable under §§ 328, 330 and 331 of the Bankruptcy Code, and any procedures order from this Court, but only pursuant to pre-approved budgets approved by RMS in writing, at its sole and absolute discretion, and only out of Cash Collateral and out of funding proceeds paid by RMS to the Providers. During an Event of

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 22 of 102

Default, no professional fees can be paid absent an authorizing Order from this Court. RMS's security interest as Buyer in Purchased Accounts and in Conveyed Property cannot be invaded or used to pay professional fees or costs; provided, however, that such preclusion does not apply to the Chapter 11 Quarterly Fees (as defined below).

- (e) Payment on account of the Carve-Out shall not be deemed to reduce the Obligations owed to RMS.
- 8. <u>Validity and Perfection of Ownership and Security Interests and Liens; Automatic</u>
 Perfection.
- (a) Upon entry of this Order, the Post-Petition Ownership Interests, security interests and liens granted to RMS pursuant to this Interim Order shall be valid, enforceable, valid, attached and perfected without further notice and without regard to any other federal, state or local requirements or laws requiring notice, filing, registration, recording or possession of collation or any other act. RMS shall not be required to file any financing statements, mortgages, UCC-1 financing statements, notices of lien of similar instruments in any jurisdiction or filing office, or required to take any other action in order to validate or perfect the Post-Petition Ownership Interests, security interests and liens granted by Provider pursuant to this Interim Order and the Purchase Agreement.
- (b) Notwithstanding the foregoing, RMS may, in its sole discretion, elect to file, record or otherwise record this Order and/or such UCC-1 financing statements, mortgages, deeds of trust and other documents as RMS may deems fit, including, without limitation, a UCC-1 financing statement as to the Purchased Accounts and the Conveyed Property listing RMS as secured party and the Provider as Provider, to secure all of its right, title and interest in the

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 23 of 102

Purchased Accounts and the related Conveyed Property, and a UCC-1 financing statement reflecting RMS's lien on the Collateral.

- (c) Should RMS in its sole discretion, from time to time, choose to file such financing statements, mortgages, deeds of trust, notices of lien or similar instruments, or take any other action to validate or perfect any such Post-Petition Ownership Interests, security interest, lien or mortgage, the Provider and its officers are hereby directed to execute any such documents or instruments as RMS may reasonably request in accordance with the Purchase Agreement and the other Funding Documents, and all such documents and instruments shall be deemed to have been filed or recorded as of the date and time of the entry of this Order.
- (d) A certified copy of this Order may, in the discretion of RMS, be filed with or recorded in filing or recording offices in addition to or in lieu of such financing statements, deeds, mortgages, notices of lien and similar instruments, and all filing offices are hereby directed to accept such certified copy of this Order for filing and recording.
- 9. <u>Validity and Priority of the Post-Petition Ownership Interests</u>. The Post-Petition Ownership Interests granted to RMS herein are fully valid and this Order vests title in all the Conveyed Property in RMS, free and clear of, and senior in priority to, all other competing ownership claims, security interests, liens, mortgages and encumbrance of any kind asserted by the Provider or any third party, whether created consensually, by a Court order or otherwise, including, without limitation, liens and security interests granted by the Provider in favor of third parties under §§ 363, 364, or any other section of the Bankruptcy Code.
- 10. <u>Seniority of RMS Liens on the Collateral</u>. Subject only to the Carve-Out, the liens and security interests granted to RMS herein on the Collateral to secure any Obligations due by the Provider to RMS under the Funding Documents shall be of first-priority and senior to,

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 24 of 102

and shall not be subordinated or made equal to, any lien, security interest, mortgage or other interest in favor of any party by any order of the Court or otherwise, including, without limitation, liens or interests:

- (a) authorized in connection with the use of Cash Collateral under § 363 of the Bankruptcy Code;
- (b) granted in connection with the obtaining of credit or incurring of indebtedness under § 364 of the Bankruptcy Code;
- (c) recognized as senior under § 510 of the Bankruptcy Code, governing subordination; or
- (d) provided for in a plan of reorganization or an order confirming a plan of reorganization in these cases, in these Cases or in any Successor Cases unless:
- (i) RMS has given its prior express written consent, which shall not be implied from any other action, inaction of acquiescence by RMS or (ii) the order providing for such lien, interest or subordination is conditioned on full satisfaction of the Obligations of the Provider to RMS, as determined after all open Batches have first closed (as confirmed in writing by RMS). Without limitation of the foregoing, unless RMS has given its prior written consent, or the Court enters an order, upon proper notice and hearing, requiring that all of the Providers' Obligations to RMS be immediately satisfied in full, as determined after all open Batches have first closed (as confirmed in writing by RMS) and the same has occurred, there shall not be, at any time, in these Cases or any Successor Cases any further order that authorizes (i) under § 363 of the Bankruptcy Code, the use of Collateral in which RMS has an interest or the use, sale or lease, other than in the ordinary course of business, of any other property of the Providers in which RMS has an interest; (ii) under § 364 of the Bankruptcy Code, the obtaining of credit or the

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 25 of 102

incurring of indebtedness that is (A) secured by a security interest on lien in property that constitutes part of the Collateral pledged to RMS or (B) entitled to priority administrative status which is equal or senior to that granted to RMS herein, or (iii) the enforcement by any creditor of the Provider of any alleged judgment, security interest or lien in any Collateral found by the Court to be junior to RMS, nor shall the Provider and/or its creditors take any other action which impairs RMS' Post-Petition Ownership Interests in the Purchased Accounts and the other Conveyed Property.

11. Limitations on the Funding Facility, Cash Collateral, and Carve Out. proceeds of the Funding Facility, Cash Collateral, and Carve Out may not be used: (a) in connection with or to finance in any way any action, suit, arbitration, proceeding, application, motion or other litigation of any type (i) adverse to the interests of RMS or its rights and remedies under the Funding Documents, or this Order, as applicable, including, without limitation, for the payment of any services rendered by the professionals retained by the Debtors or the Statutory Committee in connection with the assertion of or joinder in any claim, counterclaim, action, proceeding, application, motion, objection, defense or other contested matter, the purpose of which is to seek, or the result of which would be to obtain, any order, judgment determination, declaration or similar relief; (ii) invalidating, setting aside, avoiding or subordinating, in whole or in part, the Prepetition Obligations; (iii) for monetary, injunctive or other affirmative relief against any RMS or its Collateral, or (iv) preventing, hindering or otherwise delaying the exercise by RMS of any rights and/or remedies under this Interim Order, the Funding Documents, or applicable law, or the enforcement or realization (whether by foreclosure, credit bid, further order of the Court or otherwise) by RMS upon any of the Collateral; (b) to make any distribution under a plan of reorganization in any of the Cases or any

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 26 of 102

Successor Cases; (c) to make any payment in settlement of any claim, action or proceeding, before any court, arbitrator or other governmental body without the prior written consents required under the Funding Documents; (d) to pay any fees or similar amounts to any person who has proposed or may propose to purchase interests in any of the Debtors without the prior written consents required under the Funding Documents, (e) subject to the limited use of Cash Collateral as provided in this Interim Order, objecting to, contesting, or interfering with in any way the RMS' enforcement or realization upon any of the Collateral once an Event of Default has occurred; (f) using or seeking to use Cash Collateral or selling or otherwise disposing of Collateral without the consents required under the Funding Documents; (g) using or seeking to use any insurance proceeds constituting Collateral without the consents required under the Funding Documents; (h) incurring any debt outside the ordinary course of business without the prior consents required under the Funding Documents; (i) objecting to or challenging in any way the claims, liens, or interests (including interests in the Prepetition Collateral or Collateral) held by or on behalf of RMS; (j) asserting, commencing or prosecuting any claims or causes of action whatsoever, including, without limitation, any actions under chapter 5 of the Bankruptcy Code, against RMS; (k) prosecuting an objection to, contesting in any manner, or raising any defenses to, the validity, extent, amount, perfection, priority, or enforceability of any of the Prepetition Obligations, Prepetition Liens, or any other rights or interests of any of RMS; or (l) preventing, hindering or otherwise delaying the exercise by any RMS of any rights and remedies granted under this Interim Order. Notwithstanding the foregoing, the proceeds of the Post-Petition Funding Facility, Cash Collateral and Carve Out may have been used by the Statutory Committee to investigate and analyze the Prepetition Obligations and Prepetition Liens within the Challenge Period (as defined herein).

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 27 of 102

- 12. <u>No Obligation to Extend Credit.</u> RMS shall not have any obligation to purchase any Accounts or pay any Initial Installments or Subsequent Installments, or extend any other financial accommodation unless and until all of the conditions precedent to the making of such purchase and the payment or extension of credit under the applicable Funding Documents and this Interim Order have been satisfied in full or waived by RMS, in its sole and absolute discretion.
- 13. <u>Use of DIP Facility Proceeds; Payment of Prepetition Obligations</u>. From and after the Petition Date, the Debtors shall use funding received pursuant to the Post-Petition Funding Facility only for the purposes specifically set forth in this Interim Order and the Funding Documents. Notwithstanding any first-day orders entered authorizing the Debtors to pay any prepetition or other expenses, all such payments shall be made in accordance with the Budget (a copy of the Budget is attached to this Interim Order as <u>Exhibit B.</u>)
- 14. Authorization to Use Cash Collateral. Subject to the terms and conditions of this Interim Order and the Funding Documents, and in accordance with the Budget, the Debtors are authorized to use Cash Collateral until the Termination Date; provided, however, that during the seven (7) calendar days after the Termination Date, the Debtors may use Cash Collateral in accordance with the terms and provisions of the Budget solely to meet payroll and to pay expenses critical to the preservation of the Debtors and their estates. Nothing in this Interim Order shall authorize the disposition of any assets of the Debtors or their estates outside the ordinary course of business, or any Debtor's use of any Cash Collateral or other proceeds resulting therefrom, except as permitted in this Interim Order, the Funding Documents, and in accordance with the Budget.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 28 of 102

15. Adequate Protection Liens.

- (a) <u>Prepetition Funding Facility Adequate Protection Liens</u>. Pursuant to sections 361, 363(e) and 364(d) of the Bankruptcy Code, as adequate protection of the interests of RMS in the Prepetition Collateral (to the extent any Prepetition Obligations remain outstanding) against any Diminution in Value of such interests in the Prepetition Collateral, the Debtors hereby grant to RMS continuing valid, binding, enforceable, non-avoidable and automatically perfected post-petition security interests in and liens on the DIP Collateral (the "Prepetition Funding Facility Adequate Protection Liens").
- (b) <u>DOL Adequate Protection Liens</u>. Pursuant to Sections 361, 363(e) and 364(d) of the Bankruptcy Code, as adequate protection of the interests of DOL in the Prepetition Collateral against any Diminution in Value of such interests in the Prepetition Collateral, the Debtors hereby grant to the DOL continuing, valid, binding, enforceable, non-avoidable and automatically perfected post-petition security interests in and liens on the DIP Collateral (the "DOL Adequate Protection Liens," and together with the Prepetition Funding Facility Adequate Protection Liens, the "Adequate Protection Liens").

(c) Priority of Adequate Protection Liens.

- (i) The Prepetition Funding Facility Adequate Protection Liens shall be junior in payment and priority only to the: (A) Carve Out; (B) DIP Liens; and (C) Permitted Prior Liens. The Prepetition Funding Facility Adequate Protection Liens shall otherwise be senior to all other security interests in, liens on, or claims against any of the DIP Collateral.
- (ii) The DOL Adequate Protection Liens shall be junior in payment and priority only to the: (A) Carve Out; (B) DIP Liens; (C) Prepetition Liens; (D) Prepetition Funding Facility Adequate Protection Liens; and (F) Permitted Prior Liens. The DOL Adequate

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 29 of 102

Protection Liens shall be senior to all other security interests in, liens on, or claims against any of the DIP Collateral.

(iii) Except as provided herein, the Adequate Protection Liens shall not be made subject to or *pari passu* with any lien or security interest by any court order heretofore or hereafter entered in the Cases or any case under Chapter 7 of the Bankruptcy Code upon the conversion of any of the Cases, or in any other proceedings superseding or related to any of the foregoing (collectively, "Successor Cases"), and shall be valid and enforceable against any trustee appointed in any of the Cases or any Successor Cases, or upon the dismissal of any of the Cases or Successor Cases. The Adequate Protection Liens shall not be subject to §§ 506(c), 510, 549, or 550 of the Bankruptcy Code. No lien or interest avoided and preserved for the benefit of any estate pursuant to section 551 of the Bankruptcy Code shall be made *pari passu* with or senior to the Adequate Protection Liens.

16. Adequate Protection Super-Priority Claims.

- (a) <u>Prepetition Facility Super-Priority Claims</u>. As further adequate protection of the interests of RMS in the Prepetition Collateral (to the extent any Prepetition Obligations remain outstanding) against any Diminution in Value of such interests in the Prepetition Collateral, RMS is hereby granted as and to the extent provided by sections 503(b) and 507(b) of the Bankruptcy Code an allowed super-priority administrative expense claim in each of the Cases and any Successor Cases (the "Prepetition Facility Super-Priority Claim").
- (b) <u>Priority of Prepetition Facility Super-Priority Claims</u>. The Prepetition Facility Super-Priority Claims shall be junior in payment and priority only to the Carve-Out and DIP Super-Priority Claims. Except as set forth herein, the Prepetition Facility Super-Priority Claims shall have priority over all administrative expense claims and unsecured claims against

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 30 of 102

the Debtors or their estates, now existing or hereafter arising, of any kind or nature whatsoever, including, without limitation, administrative expenses of the kinds specified in or ordered pursuant to §§ 105, 326, 328, 330, 331, 365, 503(a), 503(b), 506(c), 507(a), 507(b), 546(c), 546(d), 726 (to the extent permitted by law), 1113 and 1114 of the Bankruptcy Code.

17. Adequate Protection Payments and Protections.

Prepetition Funding Facility Adequate Protection Payments and Protections. As further adequate protection to RMS, to the extent any Prepetition Obligations remain outstanding, the Debtors are authorized and directed to provide adequate protection to RMS in the form of: (i) payments of fees and other amounts due under the Funding Documents; (ii) ongoing payment of RMS' reasonable legal and other professionals' fees and expenses; and (iii) delivery to RMS of all proceeds of the Prepetition Purchased Accounts and other Prepetition Conveyed Property in accordance with the terms and conditions of the Purchase Agreement and the other Funding Documents (together, the "Prepetition Facility Adequate Protection Payments").

- 18. Reporting Requirement. The Debtors shall provide counsel to RMS, counsel to Pacifica Connect, LLC and counsel to the Statutory Committee with monthly reports (the "Monthly Reporting Period") reconciling actual cash receipts and expenses with the projected cash receipts and expenses (the "Actual/Budget Report"), respectively, in the Budget no later than the 10th of each month following the Monthly Reporting Period, with the first Actual/Budget Report due on October 10th.
- 19. <u>Bar of Challenges and Claims</u>. In accordance with this Court's prior Orders, the Statutory Committee and all other parties in interest have been provided with the right to seek, solely in accordance with the provisions of this paragraph 18 contained in such prior Orders, to assert on behalf of the Debtors or the Debtors' creditors or interest holders claims or challenges

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 31 of 102

against RMS that arose prior to the Petition Date or to otherwise challenge the Debtors' Stipulations, including, but not limited to those in relation to: (a) the sale, transfer and conveyance of the Prepetition Purchased Accounts to RMS; (b) the validity, extent, priority, or perfection of the Prepetition Liens of RMS or any other liens or security interests asserted by RMS against the Prepetition Collateral; (c) the validity, allowability, priority, secured status or amount of the Prepetition Obligations or other debt or obligations claimed by RMS and arising prior to the Petition Date; (d) avoidable transfers that arose prior to the Petition Date; or (e) any liability of RMS with respect to anything arising from or related to the Funding Documents, Prepetition Obligations and/or Prepetition Liens that arose prior to the Petition Date (separately or collectively, a "Challenge"). The Creditors Committee had also been granted standing to prosecute or assert a Challenge. To do so, any party in interest including the Statutory Committee had to commence, as appropriate, a contested matter or adversary proceeding raising such a Challenge, including, without limitation, any claim or cause of action against RMS, no later than April 10, 2016 (the "Challenge Period"). The applicable Challenge Period had not been extended with the written consent of RMS or order of this Court prior to the expiration of the applicable Challenge Period. RMS provided to the Statutory Committee those documents on which it relied no later than February 12, 2016. Only those parties in interest who commenced a Challenge within the Challenge Period were permitted to prosecute such a Challenge. Because all parties in interest, including the Statutory Committee, did not file a Challenge within the Challenge Period: (A) any and all such Challenges by any party (including, without limitation, any Statutory Committee, any chapter 11 trustee, and/or any examiner or other estate representative appointed in these Cases, and any chapter 7 trustee and/or examiner or other estate representative appointed in any Successor Case), shall be deemed to be forever waived and

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 32 of 102

barred, (B) all of the findings, Debtors' Stipulations, waivers, releases, affirmations and other stipulations as to the right, title, interest, priority, extent, and validity as to RMS' property, claims, liens, and interests shall be of full force and effect and forever binding upon the Debtors, the Debtors' Estates and all creditors, interest holders, and other parties in interest in these Cases and any Successor Cases, and (C) any and all claims or causes of action against any of RMS relating in any way to the Debtors shall be released by the Debtors' Estates, all creditors, interest holders and other parties in interest in these Cases and any Successor Cases.

20. Carve-Out.

(a) Carve Out Amounts and Other Provisions. As used in this Interim Order, the "Carve Out" shall encompass the following expenses: (i) allowed fees and reimbursement for disbursements of professionals retained by the Debtors ("Debtors' Professional Fees") in an aggregate amount for all such Professional Fees not to exceed \$540,000.00; (ii) allowed fees and reimbursement for disbursements of professionals retained by the Statutory Committee ("Committee's Professional Fees") in an aggregate amount of all such Committee's Professional Fees not to exceed \$270,000.00 (iii) quarterly fees pursuant to 28 U.S.C. § 1930(a)(6) plus interest accrued pursuant to 31 U.S.C. § 3717 ("Chapter 11 Quarterly Fees"), and any fees payable to the clerk of the Bankruptcy Court; and (iv) amounts due and owing to the Debtors' non-insider employees for post-petition wages. The Carve Out Amount shall be funded, in part, by the Debtors with the proceeds of the Post-Petition Funding Facility on a weekly basis in an amount not less than \$15,000 per week, and segregated in an interest bearing deposit account held in escrow, two-thirds for the benefit of the Debtors' professionals and one-third for the benefit of the Statutory Committee's professionals (the "Designated Amounts") Designated Amounts are sufficient to pay the Debtors' professionals in full and to leave an

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 33 of 102

excess, then such excess shall be available for the Statutory Committee's professionals. Similarly, if the Designated Amounts are sufficient to pay the Statutory Committee's professionals in full and to leave an excess, then such excess shall be available for the Debtors' professionals. An additional \$3,750 will be funded on a weekly basis and remitted directly to the United States Trustee for payment of the Chapter 11 Quarterly Fees. The Carve-Out shall be junior and subordinate to all right, title and interest of RMS in the Purchased Accounts and the other Conveyed Property, and their respective proceeds. The DIP Liens, DIP Super-Priority Claims, the Adequate Protection Liens, and the Prepetition Facility Super-Priority Claims granted pursuant to this Interim Order shall all be subject and subordinate in lien, payment and priority to the amounts payable by the Debtors on account of the Carve Out (collectively, the The Carve-Out amounts specifically allocated for the Debtors' "Carve-Out Amount"). Professional Fees and the Committee's Professional Fees shall be reduced for each dollar for dollar by the aggregate amount of the Debtors' Professional Fees and Committee's Professional Fees respectively, actually paid by the Debtors and their estates during the pendency of the Cases (exclusive of retainers paid to such professionals prior to the Petition Date), respectively.

Object to Fees. RMS shall not be responsible for the direct payment or reimbursement of any fees or disbursements of any professionals retained by the Debtors and/or any Statutory Committee incurred in connection with the Cases. Nothing in this Interim Order or otherwise shall be construed (i) to obligate RMS in any way to pay compensation to or to reimburse expenses of any professionals retained by the Debtors and/or any Statutory Committee appointed, or to guarantee that the Debtors have sufficient funds to pay such compensation or reimbursement; (ii) to increase the Carve Out Amount if actual professional fees exceed the

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 34 of 102

Carve-Out Amount; (iii) as consent to the allowance of any professional fees or expenses of any professionals retained by the Debtors and/or any Statutory Committee appointed; or (iv) to affect the right of RMS to object to the allowance and payment of such fees and expenses.

- 21. Payment of Administrative Expenses. Subject to the terms and conditions of this Interim Order, as long as an Event of Default has not occurred, the Providers shall be permitted to pay, as the same may become due and payable, (i) administrative expenses of the kind specified in § 503(b) of the Bankruptcy Code incurred in the ordinary course of business; (ii) subject to the terms hereof, compensation and reimbursement of expenses of professionals allowed and payable under §§ 330 and 331 of the Bankruptcy Code, and (iii) any other pre-filing or post-filing expenses of the Providers including for adequate protection, all of which are approved by Order of this Court and consented to in writing by RMS, as reflected in RMS's approval of the Providers' budgets, from time to time.
- No Subordination. Except for the Carve-Out on the terms and conditions set forth in this Interim Order, no costs or expenses of administration shall be imposed against RMS, the Purchased Accounts or the Collateral pledged to RMS as security for the Obligations of the Providers to RMS, as determined after all open Batches have first closed (as confirmed by RMS in writing) under any of §§ 105, 506(c), or 552 of the Bankruptcy Code, or otherwise, and, except with respect to the Carve-Out, no other claim or expense having a priority senior to or *pari passu* with that granted to RMS in this Order shall be granted in this Chapter 11 case or any superseding Chapter 7 case while any portion of the Obligations due to RMS by the Providers remain outstanding, calculated after all open Batches have first closed (as confirmed in writing by RMS). Any and all payments realized by RMS upon the sale, liquidation, collection or disposition of the Collateral or the Purchased Accounts shall be paid to RMS free and clear of

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 35 of 102

any claim or charge arising out of or based upon § 506(c) of the Bankruptcy Code or § 552(b) of the Bankruptcy Code. The "equities of the case" exception under § 552(b) of the Bankruptcy Code shall not apply to RMS with respect to proceeds, product, offspring or profits of any of the Prepetition Collateral. No costs or expenses of administration which have or may be incurred in the Debtors' Chapter 11 cases, any conversion of the Debtors' case to Chapter 7 pursuant to § 1112(b) of the Bankruptcy Code, or pursuant to § 506(c) of the Bankruptcy Code, or in any future proceedings or cases related thereto, shall be charged against RMS, the Purchased Accounts or the Collateral, without the prior written consent of RMS, and no such consent shall be implied from any other action, inaction or acquiescence of RMS.

23. Rights and Remedies upon an Event of Default.

- (a) Upon the occurrence of an Event of Default under the Funding Documents or hereunder, RMS shall provide the Providers, the Providers' counsel, counsel for the United States Attorney, and counsel for the Attorney General of the State of Connecticut, and counsel to the Creditors' Committee, and the Office of the United States Trustee with a notice of Event of Default via telephone and in writing, by electronic mail or by overnight, national courier service (a "Default Notice").
- (b) Upon provision of a Default Notice, and pending a Court hearing thereon (the "Default Hearing"), and notwithstanding the automatic stay of § 362 of the Bankruptcy Code, (i) the Providers shall cease offering Accounts to RMS for its discretionary purchase consideration; (ii) RMS shall cease funding to the Providers under the Purchase Agreement; and (iii) RMS shall have all other rights and remedies it holds under the Funding Documents.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 36 of 102

- (c) At the Default Hearing, the only issue that may be raised by any party is opposition thereto shall be whether or not, in fact, and Event of Default has occurred and is continuing.
- (d) RMS may serve a motion for Order granting itself relief from the automatic stay contemporaneously with the Default Notice. Should the Court find, at the Default Hearing, that the Event of Default alleged in the Default Notice has occurred, RMS shall have immediate relief from the automatic stay (the 14-day stay under Fed. R. Bankr. P. 4001(a)(3) being waived) to exercise all of RMS's rights and remedies under the Funding Documents, this Order or applicable law to, among other things, do any or all of the following: (i) terminate the Purchase Agreement; (ii) foreclose any and all Collateral pledged to RMS under the Purchase Agreement and other Funding Documents, whether by direct collection of Non- Purchased Accounts, by public or private sale, or by retention of the Collateral, or as otherwise allowed under the Uniform Commercial Code; (iii) set-off funds of the Providers held by RMS, (iv) accelerate the Obligations due to RMS by the Provider; and/or (v) exercise any other rights and remedies against the Collateral pledged to RMS under the Funding Documents, and exercise any other right or remedy otherwise held by RMS under or in respect of the terms of the Funding Documents, this Order or under applicable law.
- (e) Nothing herein shall impair the right of the Court to order such other and further relief as the Court deems fit at the Default Hearing, or the right of RMS or the Providers or the Lead Debtor to seek other or supplemental relief at the Default Hearing.
- (f) The representations, warranties, covenants, duties, and Obligations of the Providers to RMS under the Funding Documents together with the rights, claims, liens, security

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 37 of 102

interests and ownership interests of RMS under the Funding Documents shall all remain unimpaired and unaffected and shall survive the occurrence of an Event of Default.

- (g) Nothing in this paragraph 22 implies or means that RMS is subject to the automatic stay, as to the Purchased Accounts and the other Conveyed Property which RMS is purchasing hereunder and/or to which RMS holds ownership interests therein.
- 24. <u>Termination Events</u>. The Purchase Agreement shall terminate, notwithstanding the automatic stay of § 362 of the Bankruptcy Code, upon the earliest to occur of the following (the "Termination Date"):
- (a) Upon the occurrence of an Event of Default under the Purchase Agreement and notice thereof which creates rights in RMS to terminate the Purchase Agreement, and RMS's exercise of that right;
- (b) If as to any Provider, it fails to perform or satisfy any of the terms, conditions or covenants of this Order or under the Funding Documents;
- (c) Any Provider files, supports or confirms a Chapter 11 plan of reorganization in this bankruptcy case that does not provide for the prior full satisfaction of the Obligations on or before the "effective date" of the Plan, unless otherwise agreed in writing by RMS (such as, for example, in a written commitment by RMS to convert the financing hereunder into exit financing);
- (d) The factoring called for under the Purchase Agreement terminates by its own terms or by operation of law;
- (e) Any material portion of the Collateral is foreclosed, liquidated, levied, or the subject of a similar act by any person;

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 38 of 102

- (f) Any person terminates their respective forbearances and subordinations to RMS under any inter-creditor or similar agreement related to these Debtors, or any of these persons sent notice of their intent to do the same to any person;
- (g) Any party (including, but not limited to, DSS or DRS) makes any set-offs or recoupments against the Purchased Accounts, the Conveyed Property or the Collateral, beyond limits previously agreed upon with RMS, in writing or as provided for in this Interim Order;
- (h) Any Provider's bankruptcy case is dismissed or converted to a case under Chapter 7 of the Bankruptcy Code;
- (i) There occurs a stay, modification, amendment, vacating or reversal of any terms of this Order or of the Funding Documents, or any of the rights and acknowledgements conferred thereunder, without the written consent of RMS;
- (j) Any Provider commences or continues or voluntary participates in any lawsuit, adversary proceeding or contested matter against RMS, other than with respect to a dispute by the United States Trustee, the Statutory Committee or the Debtors as to any reimbursable expense claimed by RMS;
- (k) Any person commences or continues any non-frivolous action or process with respect to any material portion of the Collateral; and
 - (l) February 11, 2017.
- 25. <u>Effect of Termination</u>. The representations, warranties, duties, covenants, and Obligations of the Providers to RMS under the Funding Documents and this Order, together with the rights, claims, liens and security interests and ownership interests of RMS under the Funding Documents and this Interim Order shall all remain unimpaired and unaffected and shall survive

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 39 of 102

the termination of the Funding Facility and the Termination Date. Without limitation of the foregoing, the Providers shall also, both pre-termination and post-termination, allow access to their respective records and facilities to RMS (including any records sold to RMS hereunder which any Provider still holds, as Primary Servicer for RMS) and the Providers shall otherwise provide RMS with information with respect to and otherwise comply with the terms and conditions and the undertakings in this Order, any subsequent orders or the Interim Order, and the Funding Documents, until the Obligations of the Providers to RMS are satisfied in full, as determined after all open Batches have first closed (as confirmed by RMS in writing).

26. Relief from the Automatic Stay. The automatic stay of § 362 of the Bankruptcy Code is hereby vacated and modified, and the 14-day period of Fed. R. Bank. P. 4001(a)(3) waived, insofar as necessary, and the stay is lifted in favor of RMS to permit RMS and the Providers to: (a) implement the Funding Documents and the Funding Facility pursuant to the terms of such documents and this Order, including, without limitation, by the turnover of monies by the Providers to RMS thereunder; (b) create, validate, evidence, attach and perfect the Post-Petition Ownership Interests, liens and security interests granted to RMS under the Funding Documents and this Order; (c) assess, charge and collect payments, discount fees, and other fees and expenses as provided for by the Funding Documents, and apply such payments to the Obligations owed by the Providers to RMS pursuant to the terms of the Funding Documents; and (d) take any other action authorized or contemplated by this Order the Funding Documents and to carry out the terms thereof.

27. Miscellaneous.

(a) <u>No Impairment of RMS's Rights</u>. If any provision of this Order is hereafter modified, vacated, altered, stayed or reversed by any subsequent order of this or any

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 40 of 102

other court, for any reason, or any failure of the Court to enter a final order acceptable to RMS on the Motion, or if this case is converted to Chapter 7 or dismissed (each a "Subject Event"), such modification, vacation, stay, alteration, reversal, failure, dismissal or conversion shall not impair (i) the validity of any sale of Purchased Accounts or Conveyed Property by the Providers to RMS under this Interim Order and the Funding Documents and the Post-Petition Ownership Interests created in RMS therein, prior to the Subject Event; (ii) the validity or priority of any lien or security interest in the Collateral granted by the Providers to RMS under this Order or the Funding Documents, prior to the Subject Event; (iii) the Obligations created under or due to RMS under this Order and the Funding Documents, prior to the Subject Event; or (iv) the creation of or exercise by RMS of any rights or remedies under the Funding Documents or this Order, prior to the Subject Event, and, without limitation, such rights, remedies, liens, security interests and Post-Petition Ownership Interests shall be governed in all respects by the original provisions of the Funding Documents and this Order and shall all remain in full force and effect, pursuant to §§ 363(m) and 364(e) of the Bankruptcy Code. For purposes of §§ 364(e) and 363(m), the term "appeal" shall include any proceeding for reconsideration, rehearing or reevaluation of this Order by the Court or any other tribunal. RMS is a good faith Buyer, within the meaning of § 363(m) of the Bankruptcy Code. Notwithstanding any such modification, vacatur or stay, any duties, obligations, representations, warranties, or Obligations of the Providers and the Lead Debtor to RMS, as well as any right created in RMS under the Funding Documents or this Interim Order prior to such modification, vacatur or stay, shall be governed in all respects by the original provisions of this Interim Order and the Funding Documents and RMS shall be entitled to all the rights, remedies and privileges granted herein and pursuant to the Funding Documents. The Providers and the Lead Debtor shall give notice to RMS of any

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 41 of 102

motion by any Debtor to modify, amend or vacate this Interim Order. Furthermore, notwithstanding the occurrence of a Subject Event, the rights and interests created pursuant to the Carve-Out shall remain in full force and effect.

- (b) <u>Inconsistencies</u>. In the event of any inconsistency between the terms of this Order and the Funding Documents this Order shall govern.
- (c) <u>No waiver</u>. RMS's failure to seek or exercise any right or remedy under this Order or the Funding Documents shall not constitute a waiver of its rights hereunder or thereunder, or otherwise.
 - (d) <u>Immediate Effect</u>. This Order shall take immediate effect upon its entry.
- (e) <u>Additional Insured</u>. Upon entry of this Order, RMS shall be and is deemed to be, without further action or notice, an additional insured on each insurance policy or policies of the Providers which in any way relates to the Collateral. Each Provider will, at its own expense, keep the Collateral insured to its maximum reasonable insured values against loss, peril and hazards, pay all post-petition taxes, assessments and charges as required by the Funding Documents, and provide proof of payment of the same to RMS, upon its request. Each Provider will further provide RMS with reasonable rights to inspect the Collateral and records related thereto.
- (f) <u>Execution of Documents</u>. The execution by each Debtor's officers of the Funding Documents is ratified, approved and confirmed.
- (g) <u>Right to Factor and Fund on Other Terms</u>. RMS retains the right to provide factoring and other funding services to the Providers on such other, different or additional terms than in this Order, as may be authorized by the Providers and approved by the Court.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 42 of 102

(h) [INTENTIONALLY OMITTED]

- (i) <u>Binding Effect</u>. The provisions of this Order shall be binding upon the Providers, the Lead Debtor, all parties in interest in this bankruptcy case, any examiner, Chapter 7 trustee or Chapter 11 trustee in this bankruptcy case, and upon their respective successors and assigns, provided, however, that RMS shall have no obligation to extend funding or financing to any Chapter 7 trustee, any Chapter 11 trustee or to an examiner, on the terms thereof.
- (j) Right of Recoupment by Medicare. Notwithstanding anything to the contrary herein, nothing in this Order or in any agreement between RMS and any of the Debtors shall (i) impair, modify or affect Medicare's right of recoupment, if any or (ii) be construed, as to the Providers, only, to impair, modify or effect the government's process of administering the Medicare provider agreements to which each Provider is a party, or the Medicare program as it relates to the Providers.
- (k) Right of Recoupment by Medicaid. Notwithstanding anything to the contrary herein, nothing in this Order or in any agreement between RMS and any of the Debtors shall (i) impair, modify or affect State of Connecticut's Medicaid program's right to make a reduction or otherwise withhold any Medicaid receivables arising from services provided by Debtor through recoupment, setoff under Section 553 of the Bankruptcy Code or otherwise, any amounts due to the State of Connecticut's Medicaid program, including, but not limited to, any recoupment on account of prior advances by the State of Connecticut; and (ii) be construed, as to the Providers, only, to impair, modify or effect DSS' process of administering the Medicaid provider agreements to which each Provider is a party, or the Medicaid program as it relates to the Providers. As used herein, "State of Connecticut's Medicaid program" shall be defined to include any agent, carrier, administrator or intermediary of DSS.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 43 of 102

- (1) Recoupments or Set-Offs By The State Of Connecticut. As an inducement to RMS's consent to this Interim Order and to provide the Post-Petition Funding Facility, and DSS' and DRS' consent to this this Interim Order:
- (i) The Providers shall pay DRS \$50,000 per month on or before the 15th day of each month commencing in February, 2016, and continue during the pendency of the Cases under Chapter 11 of the Bankruptcy Code, by check made payable to the Commissioner of Revenue Services and delivered to the Department of Revenue Services, 25 Sigourney Street, Hartford, CT 06106.
- (ii) If and when there is a shortfall in provider taxes due DRS accrued after the Petition Date, amounts otherwise due and payable to the Providers by DSS may be recouped or set-off against, but in an aggregate amount not to exceed \$10,000 per month.
- (iii) Other than to collect payments permitted under this paragraph, DSS shall not, on account of amounts due to DRS by the Debtors, deduct and withhold such amount from amounts otherwise payable to the Providers by DSS.
- (iv) DRS will give RMS contemporaneous notice of any provider taxes due and not timely paid by the Providers. DRS will also give both the Providers and RMS a 10-business day cure period following the delivery to the Providers and RMS of such notice.
- (m) <u>Objections Overruled</u>. All objections to the funding requested by the Provider in the Motion, to the extent not previously resolved or withdrawn, are **OVERRULED**.
- (n) <u>Service of Order</u>. The Provider shall serve a copy of this Order on the Providers and on all creditors of the Debtors and other interested persons within two (2) business days, unless they have been served via the ECF system, which service shall be good service.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 44 of 102

- (o) <u>No Deemed Control</u>. By consenting to this Order, by purchasing receivables and by making advances and other financial accommodations and credit of any type, kind or nature under this Order or the Funding Documents, RMS shall not be deemed to be in control of the operations of the Providers nor shall it be deemed to be a "responsible person", "managing agent", "owner" or "operator" with respect to the Providers.
- (p) <u>No Third-Party Beneficiaries</u>. Other than as expressly set forth above, no rights are created hereunder for the benefit of any third party, any creditor (other than RMS) or any direct, indirect or incidental beneficiary.
- (q) Amendment. The Providers and RMS may modify, amend, supplement or waive any provision of the Funding Documents if they do so in writing and the amendment, modification, supplement or waiver is not, in the good faith business judgment of RMS and the Providers, material, without any need for further approval from the Court or for notice to third parties, provided, however, that the Provider shall provide DSS, DRS, the Statutory Committee, the United States Trustee, and any party requesting notice with prior written notice of any proposed amendment, waiver, modification or supplement, and the Providers shall further file a prior notice of such amendment, waiver, modification or supplement, as agreed to in final form, with the Court. Any material amendment, waiver, modification or supplement must be in writing, executed by both the Providers and RMS, and shall be subject to approval by the Court, on appropriate notice. In addition, the terms and conditions provided for in Section 3.2(f) the Purchase Agreement are hereby modified solely as follows: notwithstanding the provisions of Section 3.2(f), the Buyer may make payments of Subsequent Installments of the Purchase Price of previously Purchased Accounts in amounts not subject to the dollar amount limitation (\$250,000) set forth in subsection (f), and on any day as Buyer determines in its sole and

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 45 of 102

absolution discretion, provided that the conditions set forth in Section 3.2 and otherwise in the Purchase Agreement to the payment of Subsequent Installments to the Providers have been fully satisfied. Furthermore, the "Facility Cap" provided by Section 2.4 of the Purchase Agreement is hereby amended by and substituted with the following: the aggregate amount of the Outstanding Initial Installments plus any outstanding Obligations of the Providers shall not (except as permitted by Buyer in its sole discretion) exceed \$1,875,000.00 (the "Facility Cap").

(r) Post-Petition Professional Fees of RMS. Costs, fees and expenses incurred by RMS in respect of the preparation, negotiation, documentation and execution of documents necessary to implement the Post-Petition Funding Facility and the terms and conditions of this Interim Order, and such further Orders as may hereinafter be entered in connection with the Funding Documents and the Post-Petition Funding Facility, including reasonable legal fees, expenses and costs, as well as fees, expenses and costs, including reasonable legal fees, to enforce, protect and preserve RMS's rights under the Funding Documents and this Interim Order, shall be charged to and paid by the Providers weekly (or at such greater intervals as RMS shall determine at its sole discretion) either directly from the Providers or, if RMS so elects, at its sole discretion, by deducting such amounts from any funding otherwise due to the Providers on account of the Post-Petition Funding Facility; provided, however, that notice of such costs, fees and expenses (including detailed statements evidencing the amounts included) shall be provided in advance of payment to the Office of the United States Trustee, the Statutory Committee and the Debtors, and filed (redacted as appropriate) with this Court and such costs, fees and expenses shall be subject to objection as any other component of RMS' claim in this case.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 46 of 102

- (s) <u>Waiver of Chapter 11 Discharge as to RMS</u>. Without limitation of any right of RMS under the Funding Documents, the Obligations of the Providers and the Lead Debtor to RMS under the Funding Documents shall not be discharged by the entry of an order confirming a Chapter 11 plan of reorganization in the Debtors bankruptcy cases and, pursuant to \$1141(d)(4) of the Bankruptcy Code, unless and until all of the Providers' Obligations to RMS, as determined after all open Batches have first closed (as confirmed in writing by RMS), have been paid or satisfied in full and, before such time, the Providers waive such a discharge and the Providers will not propose a reorganization plan that crams-down on RMS, over its Objection.
- (t) <u>Successors and Assigns</u>. The provisions of this Order shall inure and be binding upon RMS, the Providers and the Lead Debtor, any successor trustee or examiner hereinafter appointed in this case, and all creditors of the Providers and the Lead Debtor, and their respective successors and assigns, including in these Cases or any Successor Cases.
- (u) <u>Immediate Effect</u>. The 14 day stay provided by Fed. R. Bankr. P. 6004(h) is not applicable, so that the parties may immediately comply with this Order, and this Order shall be deemed effective and enforceable immediately.
- (v) <u>Conversion</u>. No order under § 1112 of the Bankruptcy Code or otherwise converting any Debtor's Chapter 11 case to a case under Chapter 7 may be entered unless such Order expressly provides that the priority of RMS's ownership interests, liens and rights as set forth herein and in the Funding Documents are preserved.
- (w) <u>Notice</u>. Notice of the Motion and of the September 13, 2016 hearing thereon was adequate for all purposes of the Bankruptcy Code and the Federal Rules of Bankruptcy Procedure.

Continued Interim Hearing. A hearing to consider the entry of a further (x) Interim Order on the same terms and conditions as this Interim Order is scheduled for February 8, 2017 at 10:00 a.m. (Eastern Time) before the Honorable Julie A. Manning, Chief United States Bankruptcy Judge, at the United States Bankruptcy Court for the District of Connecticut, Bridgeport Division, 915 Lafayette Boulevard, Courtroom 123, Bridgeport, CT 06604. On or before January 16, 2017, the Debtors shall serve, by United States mail, first-class postage prepaid, notice of the entry of this Fourteenth Interim Order, a copy of the Fourteenth Interim Order and notice of the Continued Interim Order Hearing (the "Continued Interim Order Notice") on: (a) the Statutory Committee (b) any party which has filed prior to such date a request for notices with this Court; and (c) counsel for the Statutory Committee. The Continued Interim Hearing Notice shall state that any party in interest objecting to the entry of a Thirteenth Interim Order shall file such objection with the Clerk of the Court no later than on February 3, 2017 at 4:00 p.m. (Eastern Time), and shall serve such objections so as to be received on or before such date and time by: (i) counsel to the Debtors, Pullman & Comley, LLC, 850 Main Street, 8th Floor, PO Box 7006, Bridgeport, CT 06601, Attn: Elizabeth J. Austin; (ii) counsel to Creditors Committee, Neubert, Pepe & Monteith, P.C., 195 Church Street, 13th Floor, New Haven, CT 06510, Attn: Mark I. Fishman; (iii) counsel to RMS, Zeisler & Zeisler, P.C., 10 Middle Street, 15th Floor, Bridgeport, CT 06604, Attn: Stephen M. Kindseth; and (iv) the Office of the United States Trustee for the District of Connecticut, Giaimo Federal Building, 150 Court Street, Room 302, New Haven, CT 06510, Attn: Steven E. Mackey. Unless an objecting party (or its counsel) appears at the Continued Interim Order Hearing to assert the basis for such objection, such objection shall be deemed to have been waived and abandoned by such objecting party. DATED: January 13, 2017

> **Julie A. Manning** Crief United States Bankruptcy Judge District of Connecticut

EXHIBIT A

PURCHASE AGREEMENT

THIS PURCHASE AGREEMENT (the "Agreement") is made and entered into as of the 25th day of May, 2012, by and between Revenue Management Solutions, LLC, a limited liability company organized and existing under the laws of the State of Connecticut (the "Buyer") and Health Care Investors, Inc., d/b/a Alexandria Manor, and Health Care Alliance, Inc., d/b/a Blair Manor, each a corporation, and Health Care Assurance, LLC, d/b/a Douglas Manor, and Health Care Reliance, LLC, d/b/a Ellis Manor, each a limited liability company and all of which are organized under the laws of the State of Connecticut (together, the "Providers").

This Agreement will govern the relationship of the parties as it relates to the purchase and sale of the Providers' Accounts. Once signed by all parties, this Agreement shall be deemed effective as of the date set forth above.

AGREEMENT

1. Definitions.

When used herein, the following terms shall have the meanings set forth below:

"Account" or "Accounts" shall mean individually or collectively all of Providers' (i) accounts (as that term is defined in the UCC), (ii) payment intangibles (as that term is defined in the UCC), and (iii) all other rights of payment, collection or reimbursement (whether owed directly to the Providers or assigned to Providers by a patient or other third party), whenever due, that arose out of, or will arise out of, the rendering whether before or after the date of this Agreement of Healthcare Services, and including, without limitation, all of Providers' rights of payment, collection or reimbursement with respect to such Healthcare Services from any insurer, federal or state government agency or other third party; whether billed on a fee for service, monthly per patient capitation charge or any other basis, whether or not the accounts, payment intangibles, or rights of payment, collection or reimbursement have been invoiced or billed, written off, partially paid, currently assigned to collection agencies or other third party service vendors. Without limiting the foregoing, Accounts shall also include all monies due or to become due to the Providers and obligations to the Providers in any form (whether arising in connection with contracts, contract rights, instruments, or chattel paper) with respect to Healthcare Services, in each case whether or not earned by performance, now or hereafter in existence, and all documents of title or other documents representing any of the foregoing, and all collateral security and guaranties of any kind, now or hereafter in existence, given by any Person with respect to any of the foregoing.

"Affinity" shall mean affinity Health Care Management, Inc.

"Assets" shall mean all Accounts and Related Property of such Accounts as may be Purchased by Buyer pursuant hereto.

"Audit Fees" shall have the meaning set forth 5.1.

"Base Rate" shall mean ten percent (10%) per annum.

"Batch" means all Accounts selected by Buyer to be purchased on a given Closing Date.

"Billing Date" means the day on which the applicable Providers first submitted a duly completed and supported claim or bill to a Payor for payment and collection of an Account.

"Buyer" shall have the meaning set forth in the heading of this Agreement.

"Business Day" means any day that is not a Saturday, Sunday, federal legal holiday or legal holiday in the State of Connecticut.

"Closing" shall have the meaning set forth in Section 2.3.

"Closing Date" shall mean the date upon which the Initial Installment (adjusted as set forth in Section 3.1(c)) is paid with respect to an Account or Batch of Accounts.

"Collateral" means all of the following property now owned or at any time hereafter acquired or created by the Providers, or in which the Providers now have or at any time in the future may acquire any right, title or interest (the "Collateral"): all present and future accounts, including all Accounts, whether or not Purchased by Buyer pursuant to this Agreement, all Related Property of the Accounts, all other personal property and fixtures of the Providers, all machinery and equipment, inventory, general intangibles (including, without limitation, payment intangibles and software), chattel paper, supporting obligations, investment property, instruments, securities, contract rights, equity interests in direct and indirect subsidiaries, deposit accounts (including, without limitation, the Lockbox Accounts), letter-ofcredit rights, intellectual property, copyrights, trademarks, patents, and tradestyles in which the Providers now have or hereafter may acquire any right, title or interest and the proceeds and products thereof (including, without limitation, proceeds of insurance) and all additions, accessions and substitutions thereto and therefor. The priority of the Buyer's security interests shall be as provided in the Intercreditor Agreements. Terms used in the foregoing language of this Section which are defined in the Uniform Commercial Code are used as so defined in the Uniform Commercial Code except as herein expressly provided to the contrary.

"Collections" means with respect to any Account, all cash collections and other amounts required to be credited to the Reserve Account pursuant to Section 3.2(b) hereof for such Account.

"Commercial Lockbox Account" shall have the meaning set forth in Section 5.3(a).

"Contractual Adjustments" means the aggregate of (i) amounts disallowed pursuant to terms of insurance contracts and payment arrangements with third party Payors and (ii) Patient Co-Payments.

"Date of Service" of an Account shall mean the earliest date the healthcare service for which the Account is payable was rendered or the healthcare related equipment, prosthetics, pharmaceuticals or other goods for which the Account is payable were delivered.

"Default Rate" means five percent (5%) in excess of the Base Rate per annum, or such lower rate required by law.

"Eligible Account" means an Account satisfying the following conditions:

- (a) The obligor is a resident of the United States;
- (b) It is denominated and payable in U.S. dollars in the United States;
- (c) It is not payable under workmen's compensation laws or insurance, automobile insurance whether under a no-fault law or otherwise;
- (d) The date the Healthcare Services were provided by the Providers to the applicable patient or healthcare facility (which Healthcare Services gave rise to the Account) is less than 180 days prior to the Closing Date;
- (e) If the Account has been billed, the Billing Date is not more than ten Business Days after the last Date of Service properly included in the invoice;
- (f) If the Account has not been billed, the Account shall be an Eligible Account only to the extent of revenue earned on each day prior to issuance of any invoice in an amount equal to 1/30th of the monthly revenue expected in Buyer's judgment for such Account or, in the case of new [Medicare] patients, at the private pay rate until the [Medicare] rate is set;
- (g) It was generated by the rendering by the Providers of Healthcare Services of the same type and scope and at the same healthcare facilities as the Healthcare Services being rendered by the Providers on the date of this Agreement as disclosed to Buyer; and
- (h) It complies with all of the representations and warranties set forth in Section 4.11 of this Agreement.

"ENR" means the expected net realizable value of an Account or Accounts. The ENR of Accounts shall be the aggregate Gross Value of Accounts minus Contractual

Purchase Agreement, Page 3

Adjustments, as estimated by the Providers and as acceptable to Buyer in its discretion. The ENR shall be subject to further adjustment by Buyer, in its sole discretion, to reflect the payment history of the Accounts and its own estimate of the Contractual Adjustments.

"Event of Default" shall have the meaning set forth in Section 8.1.

"Extended Term" shall have the meaning set forth in Section 9.1.

"Facility Cap" shall have the meaning set forth in Section 2.4.

"financial inability to pay" shall not include a delay in payment by a state Medicaid program (i) pending passage of necessary legislation or (ii) mandated by applicable law.

"Funding Fee" means a fee of Thirty-Five Thousand and 0/100 (\$35,000.00) per month or any portion thereof.

"Governmental Lockbox Account" shall have the meaning set forth in Section 5.3.

"Government Receivable" means any Account which is the obligation of the United States of America, or any State or Territory of the United States of America, and the District of Columbia, or any of their respective agencies, whether under Medicare or Medicaid or otherwise, and whether or not the Account is the primary obligation of such government, agency or agent.

"Gross Value" means the total billing amount of each Account inclusive of the amount of the Patient Co-Payment and the amount payable by a Payor.

"Guaranty" means any of the Guaranties referenced in Section 2.3(b) whether in the form attached hereto as Exhibit E or otherwise.

"HCI" means Health Capital Receivables Funding Administrative Corporation or or any of its affiliates, including without limitation Health Capital Receivables Funding Special Purpose Corporation I, Health Capital Investors, Inc., and Steven B. Nitsberg.

"Healthcare Services" shall mean medical and healthcare services provided by any Person, including, but not limited to, services of physicians, nurses, therapists, or other licensed or unlicensed healthcare personnel, hospital services, skilled nursing facility services, comprehensive outpatient rehabilitation services, home healthcare services, residential and outpatient behavioral healthcare services, the provision of room, board and daily living assistance at licensed healthcare facilities, home care services, transportation to or from healthcare facilities, or the sale, assignment, lease or license whether before or after the date of this Agreement of healthcare related equipment, prosthetics, pharmaceuticals or other goods and any

other medical and healthcare goods and services which are covered by a policy of insurance or by Medicare, Medicaid or any other federal healthcare program, including, without limitation, TRICARE (formerly known as CHAMPUS) and CHAMPVA.

"HUD" means the United States Department of Housing and Urban Development.

"Indemnified Party" shall have the meaning set forth in Section 6.1.

"Initial Installment" shall have the meaning set forth in Section 2.2(a), above.

"Insolvency Proceeding" means the commencement by the Providers of any bankruptcy, insolvency, arrangement, reorganization, receivership or similar proceedings under any federal or state law; or the commencement against any of the Providers or Affinity of any bankruptcy, insolvency, arrangement, reorganization, receivership or similar proceedings under any foreign, federal or state law.

"Intercreditor Agreement" means those certain Intercreditor Agreements of even date by and among each of the Providers and their respective landlords, the Buyer and the United States of America acting by the Secretary of HUD.

"Laws" shall mean any judgment, decree, order or award of any court, governmental body or arbitrator or any federal, state, municipal, local or foreign laws, statute, ordinance, rule or regulation.

"Liens" shall mean liens, mortgages, pledges, security interests, restrictions, prior assignments, encumbrances and charges of any kind or nature whatsoever (including any agreement to give any of the foregoing, any conditional sale or other title retention agreement or any lease in the nature thereof), or any other arrangement pursuant to which title to property is retained by or vested in some other Person for security purposes.

"Lock Box Account" shall have the meaning set forth in Section 5.3(a).

"Lock Box Agreement" shall have the meaning set forth in Section 5.3(a).

"Lock Box Fee" shall have the meaning set forth in Section 5.3(d).

"Material Agreements" means each contract, license or agreement of each of the Providers, whether oral or written, including without limitation, equipment and real property leases, credit, loan or other financing agreement, contracts or sales or purchase orders for the sale or purchase of any asset or service, employment agreements, and licenses or other agreements pertaining to the rights of each of the Providers or a licensee to utilize any patents, trademarks, trade names or know-how, in each case which involve payment or receipt by the Providers in the aggregate over the entire term of the contract exceeds \$250,000.00.

"Medicaid" means, collectively, the health care assistance program established by Title XIX of the Social Security Act (42 USC Sec.1396 et seq.) and any statutes succeeding thereto, and all laws, rules, regulations, manuals, orders, guidelines or requirements pertaining to such program including (a) all federal statutes (whether set forth in Title XIX of the Social Security Act or elsewhere) affecting such program; (b) all state statues and plans for medical assistance enacted in connection with such program and federal rules and regulations promulgated in connection with such program; and (c) all applicable provisions of all rules, regulations, manuals, orders and administrate, reimbursement, guidelines and requirements of all government authorities promulgated in connection with such program (whether or not having the force of law), in each case as the same may be amended, supplemented or otherwise modified from time to time.

"Medicare" means, collectively, the health insurance program for the aged and disabled established by Title XVIII of the Social Security Act (42 USC Sec. 1395 et seq.) and any statutes succeeding thereto, and all laws, rules, regulations, manuals, orders or guidelines pertaining to such program including (a) all federal statutes (whether set forth in Title XVIII of the Social Security Act or elsewhere) affecting such program; and (b) all applicable provisions of all rules, regulations, manuals, orders and administrative, reimbursement, guidelines and requirements of all governmental authorities promulgated in connection with such program (whether or not having the force of law), in each case as the same may be amended, supplemented or otherwise modified from time to time.

"Non-Government Receivables" shall mean all Accounts other than Government Receivables.

"Non-Purchased Account" shall mean any Account of any of the Providers that is not Purchased hereunder.

"Obligations" shall mean any and all payment obligations, including interest accrued thereon at the rate provided in this Agreement, which may at any time be owing by any of the Providers to Buyer, howsoever arising, whether now in existence or incurred by any of the Providers at any time or times hereafter, whether secured by pledge, lien upon or security interest in any assets or property of any of the Providers or the assets or property of any other person, firm, entity or corporation; whether any such payment obligation is absolute or contingent, joint or several, matured or unmatured, direct or indirect and whether any Providers are liable to Buyer as principal, surety, endorser, guarantor or otherwise. Obligations shall include, without limitation, the amounts that now are or which may hereafter become due and payable to Buyer by any of the Providers by reason of the Funding Fee, Buyer's attorneys' fees, Providers' indemnification obligations, the origination fee as specified in Section 2.4, the Lock Box Fees described in Section 5.3(d), the Audit Fees described in Section 5.1, wire transfer fees as described in Section 10.15, and any other fees and Obligations of any of the Providers that are then due and payable. Notwithstanding the foregoing, "Obligations" does not include the repurchase obligation set forth in Section 3.3.

"Outstanding ENR Balance" means the aggregate of the ENR of all Purchased Accounts in a Batch less all amounts withdrawn by Buyer from the Reserve Account pursuant to Section 3.2(d) hereof with respect to such Batch.

"Outstanding Initial Installment" means the aggregate of the Initial Installments with respect to a Batch less all amounts withdrawn by Buyer from the Reserve Account pursuant to Section 3.2(d) hereof with respect to such Batch plus all amounts withdrawn from the Reserve Account pursuant to Section 3.2(c) on account of the Providers' repurchase obligations in connection with such Batch.

"Patient Co-Payment" means the amount of each Account payable by the patient not payable by a Payor.

"Payor" shall mean Persons that are responsible either directly or as a third party payor for the payment of an Account.

"Permits and Licenses" shall have the meaning set forth in Section 4.6.

"Providers" shall have the meaning set forth in the heading of this Agreement.

"Purchase" shall have the meaning set forth in Section 2.1, which meaning shall apply to all forms of this term, including "Purchased" and "Purchases".

"Purchased Account" shall mean an Account of any of the Providers that is Purchased hereunder.

"Purchase Price" shall have the meaning set forth in Section 2.2.

"Related Property" shall mean, with respect to each Account, the following: (i) all books and records of any nature evidencing or related to the Account, including, but not limited to, ledgers, records indicating, summarizing or evidencing an Account and all computer programs, discs, tapes or other electronic files or computer prepared information with respect to the foregoing and any software necessary to operate the same, all contracts, invoices, charges slips, credit memoranda, notes and other instruments and other documents, books, records and other information, (ii) all security interests or liens and property subject thereto from time to time purporting to secure payment of such Account, whether pursuant to the contract related to such Account or otherwise, including all rights of stoppage in transit, replevin, reclamation, supporting obligations and letter of credit rights (as such terms are defined in the Uniform Commercial Code), and all claims of lien filed or held by any of the Providers on personal property; (iii) all rights to any goods whose sale gave rise to such Account, including returned or repossessed goods; (iv) all instruments, documents, chattel paper and general intangibles (each as defined in the Uniform Commercial Code) arising from, related to or evidencing such Account; (v) all UCC financing statements covering any collateral securing payment of such Account; (vi) all guaranties and other agreements or arrangements of whatever character from

time to time supporting or securing payment of such Account whether pursuant to the contract related to such Account or otherwise; and (vii) all proceeds and amounts received or receivable arising from any of the foregoing.

"Repurchase Date" means (a) with respect to an Account Purchased on the initial Closing Date, the 180th day after the initial Closing Date and (b) with respect to any Account Purchased on any date subsequent to the initial Closing Date, the 180th day after the Date of Service with respect to such Account, and (c) with respect to any Account, the date on which Buyer reasonably determines, will not be paid within the periods set forth in clauses (a) and (b) above because of disputes raised by the payor or the payor's proposed exercise of rights of recoupment, set-off or counterclaim.

"Reserve Account" shall have the meaning set forth in Section 3.2.

"Retained Liabilities" shall have the meaning set forth in Section 2.5.

"Servicing Fee" shall have the meaning set forth in Section 5.3(c).

"Subsequent Installment" shall have the meaning set forth in Section 2.2(b) hereof.

"Taxes" shall have the meaning set forth in Section 4.14.

"Term" shall have the meaning set forth in Section 9.1.

"Uniform Commercial Code" shall mean the Uniform Commercial Code, as amended, as presently in effect in the State of Connecticut.

All references to Section numbers throughout this Agreement are references to those Sections of this Agreement, unless otherwise expressly stated to the contrary.

2. Sales and Purchases.

2.1 Obligation to Offer Accounts for Purchase; Purchase of Accounts; Identification of Accounts Purchased. The Providers agrees to offer for sale to Buyer all of the Providers' Accounts on a weekly basis during the Term of this Agreement and during any Extended Term and Buyer agrees to purchase all such accounts selected by the Buyer to be purchased. Such offer of Eligible Accounts by the Providers shall be by submission by each Provider to Buyer each week of a complete list of all of such Providers' Accounts which have not been Purchased by Buyer. Such submission shall be made electronically pursuant to criteria identified from time to time by Buyer. On the terms and subject to the conditions set forth in this Agreement, upon the payment of the Initial Installment (less any offsets, credits or reductions thereto as provided in this Agreement or by applicable law) by Buyer with respect to a Batch of Accounts, all Accounts in such Batch will be conveyed, transferred, assigned, sold

and delivered to Buyer and Buyer will have acquired, accepted and purchased from the Providers, all such Accounts. The specific Accounts Purchased on any Closing Date are those identified as Purchased by Buyer. Upon the transfer of the Accounts to Buyer, the Related Property of such Accounts shall also be transferred to Buyer automatically and without further action by either party hereto. Each transaction described in this Section shall be referred to as a "Purchase" for all purposes hereunder without regard to whether a court or other authority determines, contrary to the intention of the parties and the terms of this Agreement, that such transactions, in fact, constitute loans of funds by Buyer to the Providers. The representations and warranties made by the Providers in this Agreement or in any document delivered hereunder or in connection with any Purchase, shall survive the Closing Date of each and every Purchase of Accounts. The closing of any Purchase of Accounts shall not be deemed a waiver of any failure of the Providers to comply with any condition, covenant or agreement contained herein.

- 2.2 <u>Purchase Price for the Assets.</u> Except for the purchase referred to in (a) below, the purchase price (the "Purchase Price") of Accounts shall be determined separately for each Batch of Accounts purchased and shall be equal to the aggregate ENR of all Eligible Accounts in such Batch. The Purchase Price shall be paid as follows:
- (a) Upon the Purchase of Accounts, Buyer shall pay to the Providers an amount (the "Initial Installment") equal to the lesser of (a) eighty (80%) percent of the Purchase Price of all of the Eligible Accounts in a Batch or (b) the amount requested by Providers.
- All remaining installments of the Purchase Price subsequent to the (b) Initial Installment ("Subsequent Installments") with respect to all Purchased Accounts shall be aggregated and paid with respect to all Purchased Accounts as a group, regardless of the date such Accounts were or are Purchased, in accordance with the provisions of this paragraph (b) and such payment of all remaining amounts of the Purchase Price shall be subject to the terms of this Agreement and applicable legal requirements. The funds from which the Subsequent Installments of the Purchase Price are to be paid shall be limited to the funds in the Reserve Account from time to time pursuant to the terms of this Agreement. Funds shall be released from the Reserve Account on account of the Subsequent Installments of the Purchase Price, in accordance with the provisions of Section 3.2 hereof. All funds in the Reserve Account shall, until released to the Providers pursuant to the terms of this Agreement, be the sole property of Buyer and the Providers shall have no legal or equitable interest therein. Buyer shall have no responsibility for payment of any installment of the Purchase Price subsequent to the Initial Installment except from funds available in the Reserve Account pursuant to the terms of this Agreement. If the funds in the Reserve Account are insufficient to pay the full Purchase Price, Buyer shall have no independent obligation to pay any remaining portion of the Purchase Price subsequent to the Initial Installment.

2.3 Closing.

Each purchase and sale of Accounts and the Related Property with respect to such Accounts pursuant to this Agreement shall be referred to herein as a "Closing". The date upon which the Initial Installment is paid with respect to an Account or Batch of Accounts is referred to herein as the "Closing Date". Prior to the first Closing Date hereunder, Providers shall deliver to Buyer the following documents, executed by the Buyer and any third parties to such documents:

- (a) This Purchase Agreement;
- (b) A Validity Guaranty in the form attached hereto as **Exhibit E** from Affinity, Benjamin Z. Fischman and Samuel Strasser;
- (c) A Landlord's Waiver and Consent in the form attached hereto as **Exhibit F** hereto with respect to each facility occupied by the Providers and their management company;
 - (d) An IRS Form 8821 for all Providers;
- (e) The Lockbox Agreements referenced in Section 5.3 hereof at People's Bank, including arrangements for the transfer thereto, on June 14, 2012, of all balances in other deposit accounts at Bank of America or otherwise where Accounts have been deposited except for amounts needed to fulfill obligations to HCI or meet current operating expenses;
- (f) Multiple copies (as instructed by Buyer) of the Notice to Payors in the form attached hereto as **Exhibit B**;
- (g) The Intercreditor Agreement satisfactory in form and substance to the Buyer in its sole discretion;
- (h) A Subordination, Non-Disturbance and Attornment Agreement with the Secretary of United States Department of Housing and Urban Development and a DSS letter limiting offsets and recoupments against the Accounts in each case satisfactory in form and substance to the Buyer in its sole discretion (the "DSS Letter");
- (i) A termination agreement from HCI and related UCC filing termination statements, releases, and termination of account control agreements satisfactory in form and substance to the Buyer in its sole discretion;
- (j) Proof of authorization by Providers of the execution of the foregoing documents (which generally shall be by way of a certified resolution of the governing board of the Providers, containing the authorizing provisions set forth in **Exhibit D** hereto);
- (k) An opinion of counsel regarding the due execution and legal, valid and binding effect of the documents executed by the Providers and its affiliates; and

(l) Such additional documents as Buyer or its counsel may reasonably require, including without limitation arrangements satisfactory to the Buyer in its sole discretion regarding the resolution of all claims of HCI, including the termination of all liens and deposit account control agreement rights of HCI.

It is possible that this Agreement will be executed before the foregoing provisions are satisfied. For the avoidance of doubt, all agreements set forth in this Agreement regarding the Lockbox Accounts, and the rights of Buyer to amounts deposited therein shall be immediately effective prior to the first Closing. If a deposit account control agreement is not yet in effect with regard to any deposit account into which Accounts are deposited, whether such Accounts are Purchased Accounts or Non-Purchased Accounts, the Providers shall issue such instructions to the bank as Buyer may direct, including without limitation any instruction to sweep funds into a separate account of Buyer or make payments to Buyer of amounts due to Buyer under this Agreement. The Providers agree to take such actions and, at the request of Buyer, execute such amendments to this Agreement and the related documents as Buyer may deem necessary to effectuate the transactions herein contemplated, including without limitation the establishment of appropriate deposit control accounts and sweep orders and the obtaining of appropriate assurances from HUD and DSS and the recording of appropriate Uniform Commercial Code filings. Each Closing hereunder prior to the complete satisfaction of items (a) through and including (1) above will occur only when Buyer is satisfied in its sole discretion with the appropriateness and sufficiency of the interim arrangements regarding such matters.

2.4 Facility Cap.

The aggregate amount of the Outstanding Initial Installments plus any outstanding Obligations of the Providers shall not (except as permitted by Buyer in its sole discretion) exceed \$2,500,000.00 (the "Facility Cap").

- 2.5 <u>Liabilities Not Assumed by Buyer</u>. The Providers covenant, represent and warrant that the Buyer shall not be deemed by anything contained in this Agreement to have assumed liabilities relating to, or arising out of, the Assets or any other business, property or assets of the Providers, including, without limitation, the following (hereinafter collectively referred to as "Retained Liabilities"):
 - (a) Any liability of any of the Providers to any person or entity;
- (b) Any liability of any of the Providers for any federal, state, municipal, local or foreign taxes, assessments, additions to tax, interest, penalties, deficiencies, duties, fees and other government charges or impositions of each and every kind or description, whether measured by properties, assets, wages, payroll, purchases, value added, payments, sales, use, business, capital stock, surplus or income with respect to ownership of the Assets up to and including the related Closing Date with respect to

- (c) Any liability or obligation (contingent or otherwise) of any of the Providers to any person or entity arising out of any litigation, claim, arbitration or other proceeding;
- (d) Any liabilities or obligations of any kind whatsoever relating to any action or inaction by any person or entity, including, without limitation, any of the Providers' officers, directors, shareholders, employees, agents, representatives or independent contractors, relating in any way to the Healthcare Services rendered or provided by any of them in connection with the Accounts or the servicing of any of the Accounts in the case of such servicing up to and including the Closing Date;
- (e) Any liability or obligation (contingent or otherwise) of any of the Providers arising out of defects in or mislabeling of, or damages to persons or property arising out of defects or mislabeling of, products (including, without limitation, prescription medications) manufactured, sold, or prescribed by such Providers in connection with any of the Accounts;
- (f) Any claim by a third party payor for refund or rebate of amounts paid to the Providers, Buyer or otherwise with respect to any Account created or generated by the Providers, including any claim for repayment of funds paid in error;
- (g) Any liability or obligation of the Providers to compensate any person or entity, including, without limitation, any agent, licensor, supplier, distributor or customer of the Providers, in respect of any rendered or provided in connection with the Accounts; and
- (h) Any recapture, set-off, or other claim made by any third party payor against the Accounts.

3. Fees; Reserve Account; Security

3.1 Funding Fee; Deductions of Fees from Purchase Price.

- (a) The Providers shall pay to Buyer the Funding Fee monthly until all Obligations are paid in full at the time of the processing of the final billing file or pursuant to 3.1(b), but in any event by the end of each month.
- (b) At Buyer's option, there shall be deducted from the Initial Installment of the Purchase Price payable at the Initial Closing Date and, if applicable, at each subsequent Closing, the amount of the Funding Fee (which will thereafter not be payable again at the end of May), Buyer's out-of-pocket costs and expenses as described in Section 10.13, the indemnification obligations set forth in Section 6.1, the Lock Box Fees described in Section 5.3(d), wire transfer fees as described in Section 10.15 and any other fees and Obligations of the Providers that are then due and payable and any collection expenses, including collection agency

expenses ("Expenses"). In addition, Audit Fees described in Section 5.1, and the Servicing Fee described in Section 5.3(c) to the extent accrued, may at the discretion of the Buyer be deducted from the Initial Installment of the Purchase Price. In the event that the foregoing amounts result in the Initial Installment being reduced to zero or in the event that Buyer in its sole discretion determines to defer payment of such amounts, Buyer may deduct such amounts from any subsequent installment of Purchase Price that may be due hereunder. The foregoing shall not limit Buyer's rights to demand payment of such fees, Expenses and Obligations at any other time or in any other manner.

3.2 The Reserve Account.

- (a) The Buyer will establish a reserve account (the "Reserve Account"). The Reserve Account will be owned, maintained, managed and controlled by the Buyer and the Providers shall have no legal or equitable interest therein. The funds in the Reserve Account may be commingled with other funds of Buyer so long as Buyer separately accounts for transactions in the Reserve Account pursuant to this Section 3.2.
- (b) The Reserve Account shall be accounted for on a Batch-by-Batch basis and will be credited with all cash received by Buyer with respect to the Providers' Accounts in accordance with the terms and conditions of this Section 3.2, including any amounts paid by the Providers as a result of the Providers' repurchase obligations, as set forth in Section 3.3(b)(iii), below.
- (c) The Buyer may debit or withdraw at any time and from time to time from the Reserve Account all Expenses and other amounts due to Buyer hereunder, including without limitation amounts with respect to any Obligations and repurchase obligations then owed by the Providers to the Buyer under this Agreement and may allocate such amounts among the various Batches as deemed advisable or appropriate in the sole discretion of the Buyer (except that Expenses or other obligations not attributable to a specific Batch s shall not be allocated to Batches).
- (d) The Buyer may debit or withdraw at any time and from time to time from the Reserve Account pursuant to this Section 3.2(d) all amounts paid by third party Payors and others with respect to the Accounts Purchased from the Providers by Buyer; provided, however, Buyer shall not debit or withdraw from the Reserve Account pursuant this Section 3.2(d) an aggregate amount with respect to any Batch in excess of the aggregate ENR of the Accounts in such Batch as of the Purchase Date.
- (e) Buyer shall account monthly to the Providers with a statement of the Outstanding Initial Installments, the Outstanding ENR Balance, the Reserve Account, and charges and payments made pursuant to this Agreement, and in the absence of manifest error, any such accounting rendered by Buyer shall be deemed final, binding and conclusive on Provider unless Buyer is notified by the Providers in writing to the contrary within 30 calendar

days of receipt of each accounting, which notice shall be deemed an objection only to items specifically objected to therein.

- If no Event of Default has occurred, and after all amounts have been debited or withdrawn from the Reserve Account pursuant to Section 3.2(c) and (d) above and Section 3.3 below, the Buyer will, after establishing such reserves as in its discretion are required as a reserve against future withrawals from the Reserve Account permitted by this agreement, including without limitation Sections 3.2 or 3.3 hereof, or against Obligations and repurchase obligations of Providers that may arise under this Agreement, make payments of Subsequent Installments of the Purchase Price of previously Purchased Accounts (allocable to such Batches as Buyer may in its discretion determine) on the 28th day of each month, in the amount of (i) \$250,000.00 plus (ii) any additional then remaining in the Reserve Account after such \$250,000 payment as Buyer determines in good faith will exceed eighty percent (80%) of the Outstanding ENR Balance of all Accounts previously purchased by Buyer. Payments to the Providers under this Section 3.2(f) shall not exceed an amount equal to the aggregate Purchase Price of Purchased Accounts minus the aggregate Initial Installments with respect thereto. Upon termination of this Agreement in accordance with Section 9, balance remaining in the Reserve Account after all withdrawals permitted by this agreement, including without limitation Sections 3.2 and 3.3 will be distributed to the Providers up to an amount equal to the then unpaid Purchase Price of all Accounts purchased hereunder and any excess funds then remaining in the Reserve Account shall be debited to or withdrawn by Buyer.
- (g) Notwithstanding the provisions of this Section 3.2, in the event the Buyer determines that the Providers have breached any of their representations, warranties or covenants set forth in this Agreement, Buyer shall have no obligation to release any further amounts from the Reserve Account on account of Subsequent Installments of the Purchase Price until the Providers has cured such breach to Buyer's satisfaction.
- (h) On or before the 10th day of each month, Providers will send to Buyer, in a format approved by Buyer, a collection activity report with regard to the Accounts for the preceding month's collection activities which shall reflect the collection status of the Accounts as at the end of the preceding month.

3.3 Repurchase Obligation; Security Interest.

(a) If the Outstanding Initial Installment for an Account has not been reduced to zero by the Repurchase Date, the Providers shall repurchase such unpaid or partially paid Account on the Repurchase Date for an amount equal to the Outstanding Initial Installment as of the Repurchase Date, less any portion of the Initial Installment with respect to an Account that the Providers establish to the reasonable satisfaction of the Buyer prior to the Repurchase Date has not been paid due to the bankruptcy, insolvency or financial inability to pay of the Payor with respect thereto.

- (b) At Buyer's option, the Buyer (i) may deduct the repurchase price from the Initial Installment otherwise payable for Accounts Purchased by Buyer, (ii) may deduct the repurchase price from funds otherwise available in the Reserve Account or the Lockbox Accounts, or (iii) may require the Providers to pay the repurchase price by delivery of a certified or bank check, which check shall be deposited in the Reserve Account pursuant to Section 3.2(b).
- It is the intention of the parties hereto that each payment by the (c) Buyer to the Providers with respect to Purchased Accounts to be made hereunder shall constitute part of the purchase and sale of such Purchased Accounts and not a loan. In the event, however, that a court of competent jurisdiction were to hold that the transactions evidenced hereby constitute loans and not purchases and sales, it is the intention of the parties hereto that this Agreement shall constitute a security agreement under the UCC and any other applicable law and that the Providers shall be deemed to have granted and does hereby grant to the Buyer a first priority perfected security interest in all of the Providers' right, title and interest in, to and under, whether now owned and existing or hereafter acquired or arising, the Purchased Accounts; the Lockbox Accounts; all payments of principal of or interest on such Purchased Accounts; all Related Property of the Purchased Account; all other rights relating to and payments made in respect of this Agreement; and all proceeds of any of the foregoing. The Providers do hereby also grant to the Buyer a perfected security interest in all of the Providers' right, title and interest in, to and under, whether now owned and existing or hereafter acquired or arising, the Non-Purchased Accounts, all payments of principal of or interest on such Non-Purchased Accounts; all Related Property of the Non-Purchased Accounts; all other rights relating to and payments made in respect of this Agreement; and all proceeds of any of the foregoing. hereby authorizes the Buyer to file any and all financing statements, amendments, assignments, and continuation statements that the Buyer deems necessary or appropriate to memorialize the sale of the Purchased Accounts and the lien on the Non-Purchased Accounts under this Agreement, perfect or maintain the perfection of the security interest(s) granted by the Providers to the Buyer under this Agreement or otherwise evidence the transactions contemplated in this The Providers shall take such action, including execution of such financing statements and other documentation as Buyer may request, from time to time in order to perfect and protect Buyer's security interest in the Accounts and other Assets and to ensure that it is a first lien on the Accounts and other Assets.
- 4. Representations and Warranties of the Providers. The Providers represents and warrants to Buyer as of the date hereof and as of each Closing Date hereafter, as follows:
- 4.1 <u>Organization and Good Standing</u>. The Providers are either corporations or limited liability companies, duly organized and validly existing under the laws of the State of Connecticut, with full power to carry on their respective businesses as they are now operated, to own their respective assets and to convey good and marketable title and ownership of the Accounts to Buyer. The Providers are qualified to do business and in good standing in each

jurisdiction in which the nature of their businesses or the character of their properties makes such qualification necessary.

- 4.2 <u>Proper Authority</u>. The Providers have all requisite power and authority to enter into this Agreement and to consummate the transactions contemplated hereby. This Agreement and all other Agreements and instruments to be executed by the Providers in connection herewith have been (or upon execution will have been) duly executed and delivered by the Providers, have been effectively authorized by all necessary action, corporate or otherwise, and constitute (or upon execution will constitute) legal, valid and binding obligations of the Providers in accordance with their respective terms.
- Assets, and all Assets are free and clear of all Liens, other than as set forth on **Exhibit A** hereto. The execution and delivery to Buyer of this Agreement and, if applicable, the instruments of transfer of ownership contemplated by this Agreement will vest good and marketable title to the Assets in Buyer free and clear of all Liens and shall constitute a true sale of the Accounts under applicable law (including for tax and accounting purposes) and not a secured financing. Without limiting the generality of the foregoing, except as set forth in the Intercreditor Agreement, none of the Assets is subject to any contract, agreement or understanding (other than this Agreement), whether oral or written, or any indenture or other instrument to which any of the Providers are a party or by which any of the Providers are bound which subjects the Assets to any Lien or prohibits or restricts the conveyance of any interest therein.
- 4.4 <u>Material Agreements and Laws</u>. The execution and delivery of this Agreement by the Providers and the consummation of the transactions contemplated hereby will not result in a breach of any of the terms and provisions of, or constitute a default under, or conflict with, any Material Agreement, Certificates of Incorporation, Articles of Organization, Operating Agreements or Bylaws, as the case may be, of the Providers, or any Laws applicable to any of the Providers. Without limiting the foregoing, no event of default, or event with which the giving of notice or passage of time would constitute an event of default, has occurred under this Agreement or any other Material Agreement.

4.5 Absence of Litigation; No Set-Off.

(a) Except as set forth in <u>Schedule 4.5(a)</u> attached hereto, there are no actions, suits, proceedings or investigations pending or, to the best knowledge of the Providers, threatened by or against any of the Providers before any court, governmental agency or other tribunal, which could materially or adversely affect its ability to perform under this Agreement or which could materially or adversely affect the conduct of its business. Without limiting the foregoing, there are no claims, disputes, actions, proceedings or investigations of any nature pending or, to the best knowledge of the Providers, threatened, against or involving any of the Providers or any of their employees or persons who provide Healthcare Services under agreements with the Providers that relate in any way to any of the Assets or to the Healthcare Services rendered or provided in connection therewith. Further, there are no injunctions, writs,

restraining orders or other orders of any nature against any of the Providers that adversely affect such Providers' performance of the agreements and transactions contemplated in this Agreement, and there are no proceedings or investigations pending or threatened which adversely affect the payment or enforceability of the Accounts.

(b) Except as set forth in Schedule 4.5(b) attached hereto, there are no set-offs, recoupments, allowances, discounts, deductions, counterclaims, or disputes with respect to any Account, either at the time it is accepted by Buyer for purchase or any time prior to the date it is to be paid. "Dispute," as used in the last preceding sentence, shall mean any claim by of any kind whatsoever that is asserted by the Payor as a basis for refusing to pay an Account either in whole or in part, other than its own bankruptcy, insolvency or financial inability to pay.

4.6 Licenses and Permits; Regulatory Approval; Compliance With Law.

- (a) The Providers and their employees and persons who provide Healthcare Services under agreements with the Providers have all permits, licenses, accreditations, certifications, authorizations, approvals, consents and agreements of all Payors, governmental agencies and instrumentalities, accreditation agencies and any other person (the "Permits and Licenses") necessary or required for each of the Providers to own the assets that it now owns, to carry on its business as now conducted, and to execute, deliver, perform and consummate the transactions contemplated by this Agreement; and none of the Providers have been notified by any such Payor, governmental agency or instrumentality, accreditation agency or any other person, during the immediately preceding 24-month period, that such person has rescinded or not renewed, or intends to rescind or not renew, any such permit, license, accreditation, certification, authorization, approval, consent or agreement granted by it to the Providers or their employees or persons who provide Healthcare Services under agreements with the Providers or to which it and the Providers are parties.
- (b) With respect to the Assets and the Healthcare Services rendered or provided in connection therewith, neither any of the Providers nor, to the best knowledge of the Providers, any of their employees or persons who provide Healthcare Services under agreements with the Providers have violated, and on the date hereof does not violate, in any respect, any law, rule or regulation. Neither the Providers nor, to the best knowledge of the Providers, any of their employees or persons who provide Healthcare Services under agreements with the Providers have received any notice of any such violation.
- (c) Neither the Providers nor, to the best knowledge of the Providers, any of Providers' employees or persons who provide Healthcare Services under agreements with the Providers have engaged in any activities which are prohibited, or have received any notice that they are engaged in activities that are prohibited, under federal Medicare and Medicaid statutes, regulations promulgated pursuant to such statutes or related state or local statutes or regulations or which are prohibited by rules of professional conduct, including but not limited to:

 (A) knowingly and willfully making or causing to be made a false statement or representation of a material fact in any application for any benefit or payment; (B) knowingly and willfully

making or causing to be made any false statement or representation of a material fact for use in determining rights to any benefit or payment; (C) failing to disclose knowledge by a claimant of the occurrence of any event affecting the initial or continued right to any benefit or payment on its own behalf or on behalf of another; (D) knowingly and willfully soliciting or receiving any remuneration (including any kickback, bribe or rebate), directly or indirectly, overtly or covertly, in cash or in kind or offering to pay such remuneration: (x) in return for referring an individual to a person for the furnishing or arranging for the furnishing of any item or service for which payment may be made in whole or in part by Medicare, Medicaid or any federal healthcare program or (y) in return for purchasing, leasing or ordering or arranging for or recommending the purchasing, leasing or ordering of any good, facility, service or item for which payment may be made in whole or in part by Medicare, Medicaid or any federal healthcare program; or (z) making prohibited referrals.

- 4.7 <u>Disclosure</u>. The information provided and to be provided by the Providers to the Buyer under or in connection with this Agreement and in any Exhibit or Schedule hereto and thereto, or in any other writing in connection herewith, does not and will not contain any untrue statement of a material fact or omits or will omit to state a material fact required to be stated herein or therein or necessary to make the statements and facts contained herein or therein, in light of the circumstances in which they are made, not false or misleading. Copies of all documents heretofore or hereafter delivered or made available to Buyer pursuant hereto were or will be complete and accurate records of such documents.
- 4.8 No Prior Collections on Accounts. Other than Patient Co-Payments paid on the Date of Service (as to which no payments have been received from payors), no monies have been collected by or on behalf of any of the Providers as of the date of Purchase in respect of any Account, or, if any monies have been collected with respect to the Accounts, such funds have been turned over to Buyer.
- 4.9 Sale of Accounts Conveys Valid Enforceable Claims Against Payors. Providers have fully enforceable rights to collect each Account. Each and every step has been taken by the Providers pursuant to the terms of this Agreement or otherwise to assign to Buyer, all of the Providers' rights to collect and enforce payment of the Account by the respective Payor and the assignment thereof shall transfer fully enforceable rights to Buyer against each Payor of the respective Account to collect the full amount of each of the Accounts from such Payor.
- 4.10 <u>Solvency of Payor</u>. The Payor with respect to each Account is not, as of the date such Account is purchased, the subject of any bankruptcy, insolvency or receivership proceeding.
- 4.11 Representations, Warranties and Covenants of Providers Relating to the Accounts. In addition to the other representations, warranties and covenants of the Providers set forth herein, Providers hereby represent, warrant, and covenant with regard to each of the Accounts as of the Closing Date for the Purchase of such Accounts, that:

- (a) The services stated in, and covered by, the Accounts are Healthcare Services that were actually rendered or provided;
- (b) The patient consent form signed in connection with all Accounts is sufficient to allow the Providers to deliver to the Buyer and any servicer with respect to each such Account, and for the Buyer and any servicer to deliver to the Payor with respect to each such Account, information or documents necessary for the performance of the servicing obligations with respect to such Accounts without violating any patients' privacy rights;
- (c) The private health insurance or other contractual coverage by the responsible Payor was effective at the time the Healthcare Services were rendered or provided and each Provider has pre-verified such coverage prior to the rendering or provision of the Healthcare Services;
- (d) All supporting documentation necessary to verify a claim has been submitted to the Payor by the Providers;
- (e) To the best of Providers' knowledge, all information with respect to the Accounts offered for sale by Providers to Buyer is accurate and complete;
- (f) All billing in respect of the Account was, if billed as of the Closing Date with respect to such Account, or will be, if not billed by the Closing Date with respect to such Account, completed and submitted on a timely basis, accurately and free of errors, and in accordance with all applicable Medicare, Medicaid, other Laws, professional standards and contracts between the Providers and the Payor, including any Payor's rules, procedures or standards incorporated therein;
- (g) No other person or entity participated in the rendering of the Healthcare Services covered by the Account or is entitled to any payment whatsoever in respect of such Services;
- (h) The Healthcare Services covered by the Account were rendered in full in the ordinary course of business, in accordance with the prevailing standards of the practice of medicine in the County in which the Providers operate and are medically necessary under the Payor's standards;
- (i) The amounts charged for the Healthcare Services covered by the Account as set forth in the Accounts represent the standard amounts billed by the Providers for the same or similar services in the ordinary course of business at the date on which such Services were rendered; and
- (j) Each Account is the legal, valid and binding obligation of the respective Payor enforceable in accordance with its terms and is not subject to any dispute,

offset, counterclaim or encumbrance of any kind and Providers have no reason to believe that such Account will not be paid in the ordinary course of the business of the respective Payors.

- 4.12 Financial Condition. The Providers have furnished to Buyer true and complete copies of the Providers' unaudited balance sheets and unaudited income and cash flow statements. The balance sheets and income and cash flow statements have been prepared in accordance with generally accepted accounting principles and the Providers' regular business practices and fairly and accurately set forth the assets and liabilities and results of operations of the Providers as of the date of the balance sheets and for the periods covered by the income and cash flow statements. Since the date of the most recent balance sheet delivered to Buyer, the Providers have not incurred any liabilities or obligations except in the normal course of its business and no event has occurred which adversely affects Providers' operations, including its ability to perform the transactions contemplated by this Agreement.
- 4.13 Solvency; No Fraud against Creditors. Except as disclosed in the financial information referred to in Section 4.12 above, the Providers are not insolvent nor will any of them be rendered insolvent as a result of the sale and transfer of the Assets pursuant to this Agreement or as a result of the transactions contemplated by this Agreement. The Providers' tangible assets are in excess of the total amount of its direct or indirect, fixed or contingent, secured or unsecured, disputed or undisputed liabilities, whether or not such liabilities are reflected on a balance sheet prepared in accordance with generally accepted accounting principles. The sale of the Accounts pursuant to this Agreement is made in good faith and without actual intent to hinder, delay or defraud present or future creditors of the Providers.
- 4.14 Tax Returns and Taxes. Except as set forth on Schedule 4.14 attached hereto, all federal, state and local income, ad valorem, excise, sales, use, real or personal property, premium and other taxes and assessments of any kind, including, without limitation, amounts withheld from payroll for the payment of income tax and other governmental charges ("Taxes") which have become due and payable by the Providers have been properly computed, duly reported, fully paid and discharged (together with all interest and penalties thereon) and there are no unpaid Taxes which are or could become a lien on the properties and assets of the Providers except for current Taxes not yet due and payable. No waivers of federal or state statutes of limitations are outstanding. There are no pending assessments or proposed adjustments or audits presently in progress with respect to any returns of the Providers. The Providers have furnished Buyer with true and correct copies of their federal, state and local income tax returns for the preceding two fiscal years and copies of any extensions filed with respect to any such fiscal years and of their Forms 941 or any other report with respect to their payments to the applicable government agency of amounts withheld from payroll.
- 4.15 <u>Providers' Pension and Profit Sharing Plans</u>. Except as set forth in <u>Schedule 4.15</u> attached hereto, the Providers' (and any of Providers' consolidated subsidiaries) pension or profit sharing plans have been fully funded in accordance with Providers' applicable

obligations. The Providers have not received notice of non-compliance, of a pending audit or investigation or of any charge or deficiency with respect to any such plan. None of Providers' plans are multi-employer plans.

- 4.16 <u>Providers' Principal Place of Business</u>. The Providers' principal place of business and chief executive offices are located at the address set forth on the signature page of this Agreement, and have been located at such address for the past three months. Except as otherwise disclosed to Buyer in writing on or prior to the date of this Agreement, the Providers each conduct business only in the location set forth on the signature page of this Agreement.
- 4.17 <u>Identity of Providers</u>. The full and correct legal name and jurisdiction of incorporation or organization of the Providers are as set forth in the first paragraph of this Agreement; none of the Providers have changed its name in the last two years; and the Providers have no trade names, fictitious names, assumed names or "doing business as" names except as have previously been disclosed in writing to the Buyer.
- 4.18 <u>Financing Statements or Control Agreements</u>. There are no (i) financing statements now on file in any public office governing any property of any kind, real or personal, in which any of the Providers is named in or has signed as the debtor, or (ii) account control agreements applicable to deposit accounts of any of the Providers except the financing statement or statements filed or to be filed in respect of this Agreement or those statements now on file that are listed on <u>Exhibit A</u> hereto.

4.19 Absence of Noncompliance.

None of the Providers is in default under any obligation owed by the Providers to DRS, DSS, DPH, HUD, or any other governmental entity or any creditor or third party of any material threatened action, investigation or penalty made by any governmental entity or any creditor, (except as stated on Schedule).

5. Certain Understandings and Agreements of the Parties.

5.1 Access and Cooperation. From and after the date hereof (i) Buyer and any other authorized agents and appointed representatives of Buyer shall have reasonable access during normal business hours to all account records and any and all other documentation relating to the Accounts, including, without limitation, patient records and information, to the extent permitted by law, and all other information and documents relating to the Providers' financial condition and business, (ii) the Providers shall promptly furnish or cause to be furnished to Buyer all information (including turning over originals or copies of such information) requested by Buyer or any of its agents relating to the Assets and Providers' financial condition and business, (iii) the Providers shall provide Buyer with all information, account numbers and passwords necessary to allow Buyer to view on the Internet all deposits to

and withdrawals from each of the Providers' bank accounts, and (iv) the Providers shall provide Buyer with all information, account numbers and passwords necessary to allow Buyer to view on the Internet all accounts the Providers have with Payors. All costs, fees and expenses incurred by Buyer in conducting any such review or audit ("Audit Fees") shall be paid by Providers to Buyer, upon demand. Providers will give to Buyer and its Processors any information necessary to verify the eligibility and validity of the Accounts sold to Buyer. Providers hereby grant to Buyer or its Processor the right to verify the eligibility and validity of the Accounts sold to the Buyer by contacting any Payor. Further, Providers will instruct Payors to provide any assistance to Buyer and its Processor to conduct this process. After the Closing Date for each Purchase the Providers shall continue to cooperate fully with Buyer and Buyer's agents in any and all matters related to any Accounts, including, without limitation, matters relating to the collection of any Account. It is further understood and agreed that, to the extent permitted by law, Buyer and its agents shall have the right at any time to communicate with and seek the assistance of Payors, patients and relatives or guardians of patients of the Providers for the purpose of facilitating the servicing and collection of the Accounts.

5.2 <u>Handling of Accounts</u>. Subject to Sections 5.3, 7.5 and 8.2, below, from and after the date hereof the Accounts and Related Property shall be handled by the Providers in the ordinary course of its business, and no act shall be done or omitted to be done by or on behalf of any of the Providers, which act or omission could jeopardize collection of payment on any Account. The Providers shall make all appropriate entries in their computers or other billing system for billing or re-billing of Accounts which require that all payments be forwarded directly from the Payors of such Accounts to the appropriate Lockbox Account described in Section 5.3(a) hereof. Buyer shall refer any inquiries it receives concerning the Accounts to Providers.

5.3 Pay-Over of Receivables and Lockbox Accounts.

Prior to the consummation of the transactions contemplated by this (a) Agreement, the Providers shall (i) establish and maintain at the Providers' expense (A) an account in the name of each of the Providers with a depository institution satisfactory to the Buyer (the "Governmental Lockbox Account") into which all collections in respect of Medicaid, Medicare, Title V Maternal and Child Health Services Block Grant Program, Title XX Social Services Block Grant Program, TRICARE (formerly known as CHAMPUS) and CHAMPVA Accounts shall be deposited and (B) an account in the name of the Buyer with a depository institution satisfactory to the Buyer into which all Collections in respect of other Accounts shall be deposited (the "Commercial Lockbox Account"). (The Governmental Lockbox Account and the Commercial Lockbox Account are referred to collectively in this Agreement as the "Lockbox Account"). The Providers hereby agree to direct each Payor of an Account to remit all payments with respect to such Account for deposit in the Commercial Lockbox Account (other than the Payers of Medicaid, Medicare, Title V Maternal and Child Health Services Block Grant Program, Title XX Social Services Block Grant Program, TRICARE and CHAMPVA Accounts which shall be directed to remit all payments with respect

to such Receivables for deposit in the Governmental Lockbox Account) by (A) delivering to such Payor a notice attached as Exhibit B hereto and (B) identifying the Commercial Lockbox Account as the "pay to" address on all bills sent to Payors of Non-Governmental Receivables. The Providers further agree not to change such directive to Payors without the prior written consent of the Buyer. The Providers agree not to terminate the Governmental Lockbox Account or the Commercial Lockbox Account without first providing the Buyer with written notice at least 30 days prior to the effective date of such termination. The Providers shall give standing instructions to the bank at which the Governmental Lockbox Account shall be maintained to transfer any balances in the Governmental Lockbox Account into the Commercial Lockbox Account on a daily basis. Such instructions shall be irrevocable except on 45 days prior written notice to Buyer and the Providers hereby agree not to change or direct the custodian thereof to modify such sweep order nor to provide any other or additional instructions to the custodian thereof. In the event any Provider terminates the Governmental Lockbox Account or the Commercial Lockbox Account, changes the sweep order with respect to the Governmental Lockbox Account or the Payors receive any instruction whatsoever from such Provider indicating that Collections with respect to the Purchased Accounts should be sent to any location other than the Lockbox Account, the Providers hereby acknowledge and agree that such actions would be an express violation of this Agreement, would cause irreparable harm to the Buyer for which there would be no adequate remedy at law, and agrees and consents to entry of an order by a court of competent jurisdiction granting the Buyer specific performance of the terms and provisions of this Agreement as to the Providers.

So long as Collections with respect to Purchased Accounts are in the possession of the Providers, the Providers will (i) hold such Collections in trust for the sole and exclusive benefit of Buyer, (ii) segregate such Collections from other funds of the Providers, (iii) forward such Collections within one business day of receipt thereof to the appropriate Lockbox Account, and (iv) forward immediately to the Lockbox Account at the address stated herein, any Collections received by any Provider or its agents after the date hereof in payment or partial payment of any Accounts (including any payments made to the Governmental Lockbox Account or any other payments in respect of Government Receivables which the Providers will continue to receive notwithstanding their sale and assignment to Buyer). The Providers shall enter into such agreements with Buyer and Buyer's bank with respect to the Governmental Lockbox Account and the Commercial Lockbox Account as Buyer and its bank shall request (the "Lockbox Agreements"), including, without limitation, agreements that perfect Buyer's security interest in such accounts. The Providers further agree that, at Buyer's request, the Providers shall render to the Buyer a full and complete accounting of all Collections received in payment or partial payment of any Accounts by or on behalf of any Provider or any other person or entity other than Buyer, of which such Provider, after due investigation, is aware. Contemporaneously with each remittance of Collections to either Lockbox Account, the Providers shall deliver to Buyer a report identifying the related Account and the related Payor. All funds received into the Commercial Lockbox Account shall be property of the Buyer and the Buyer can direct to movement of funds from the Commercial Lockbox Account to any other account as Buyer shall designate in its sole discretion, regardless of whether the amounts on deposit therein represent

Purchased Accounts or Non-Purchased Accounts. For accounting purposes, all funds received into the Commercial Lockbox Account shall be credited to the Reserve Account and shall be accounted for as set forth in Section 3.2.

- (b) The Providers shall not pay, settle, compromise any Account or release any Payor from its obligation to pay the Purchased Account without the prior written consent of Buyer. The Providers agree that in connection with the Purchase of the Accounts by Buyer, at Buyer's request, the Providers shall indicate in their computer files, in the manner requested by Buyer, that the Purchased Accounts have been sold to the Buyer and that Buyer has a security interest in all other Accounts.
- (c) The Providers shall directly pay bank fees and charges for the maintenance of the Lockbox Accounts incurred by Buyer (the "Lock Box Fees"). In the event Buyer pays any such fees, Providers shall reimburse Buyer immediately. Such fees shall be due and payable as and when incurred and, at Buyer's option, may be deducted from the payment of the Initial Installment pursuant to Section 3.1(c) or from the Reserve Account or the Lockbox Accounts pursuant to Section 3.2(c) or may be required to be paid directly to Buyer.
- (d) Subject to Sections 7.5 and 8.2, the Providers shall diligently bill, service and collect all Accounts, and shall, additionally, comply with the following procedures: (i) Payors will be billed by electronic submission; (ii) all bills shall indicate conspicuously on their face and in BOX 33 of HCFA 1500 forms that the Accounts represented thereby should be paid directly, with respect to Government Receivables, to the Governmental Lockbox Account, and with respect to Commercial Receivables, to the Commercial Lockbox Account; (iii) at Buyer's request from time to time the Providers shall notify Payors designated by Buyer, that the Accounts have been purchased by Buyer; (iv) the Providers shall post all collections with respect to the Accounts as of the date such collections are deposited in the Lockbox Account, (v) any collections not deposited in the Lockbox Account in violation of this Agreement shall be separately accounted for and not posted to any account until such posting is authorized by Buyer; and (vi) the Providers shall institute any additional billing, servicing or collection procedures with respect to the Accounts as Buyer may request. In the event any Provider fails to bill and service all Accounts in accordance with this Section 5.3(e), Buyer may elect to perform such billing or servicing and such Provider shall pay Buyer's customary rates therefor.
- 5.4 <u>Software Interface</u>. Upon request of the Buyer, Providers will provide sufficient specifications to Buyer to enable Buyer to build an electronic data interface compatible with Providers' patient accounting system so that Buyer can receive data from Providers. The Providers acknowledge that inaccuracies and or omissions in the information provided for the interface specifications will affect the funding provided for the Providers pursuant to this Agreement. Buyer will not be liable for any consequences of inaccuracies and or omission of information, regarding the interface specifications. The Providers will be given the opportunity to review the specifications that will be used to develop the interface. The Buyer will make a reasonable effort to ensure that the information being sent by the Providers is

interpreted accurately and timely. In the event that a processing error occurs or that a correction is required of the interface, it is understood by all parties that:

- (a) the Buyer is only responsible for correcting the errors that it made to the extent that the remedy is reasonable and that no further liability exists;
- (b) any required correction to the interface or its operation that is the responsibility of the Buyer will be corrected, to the extent that the remedy is reasonable, at the Buyer's expense;
- (c) any correction to the interface or its operation that is due to incorrect interface specifications and or omissions in the interface specifications given to the Buyer by any of the Providers will be corrected, to the extent that the remedy is reasonable, at the Providers' expense;

5.5 Power of Attorney.

- (a) The Providers hereby grant to Buyer and its officers, employees and agents a power of attorney which is irrevocable and coupled with an interest to do any and all of the following:
 - (i) to endorse and cash any checks, instrument or other payments in respect of any of the Accounts and Related Property made payable or endorsed to the Providers or their order, whether received by Buyer or received by any other party and delivered to Buyer pursuant to this Agreement;
 - (ii) receive, open and deal with any mail addressed to the Providers and put Buyer's address or the address of the appropriate Lockbox Account on any statements mailed to Payors;
 - (iii) pay, settle, compromise, prosecute or defend any action, claim, conditional waiver and release, or proceeding relating to Accounts;
 - (iv) upon the occurrence of an Event of Default, notify in the name of the Providers the Post Office to change the address for delivery of mail addressed to the Providers to such address as Buyer may designate. Buyer shall turn over to the Providers all such mail not relating to the Accounts but may retain copies for Buyer's files;
 - (v) execute and file on behalf of the Providers any financing statement deemed necessary or appropriate by Buyer to protect its interest in and to the Accounts, including the Accounts and Related Property or in the Collateral.

- (vi) to do all things necessary and proper in order to carry out this Agreement.
- 5.6 <u>Future Encumbrances</u>. It is understood and agreed that the Providers shall not, at any time, for any reason or under any circumstances, cause or permit any of the Assets to become subject to any Liens other than the lien of Buyer or its assigns in the Accounts, other than the Liens listed on <u>Exhibit A</u>. It is further understood and agreed that, as of the Closing Date for each Purchase, the Providers shall have no right, title or interest in or to the Assets related thereto, and shall not, at any time, for any reason or under any circumstances, hold themselves out to third parties as having any right, title or interest in or to such Assets.
- 5.7 <u>Cooperation in Litigation</u>. The Providers shall fully cooperate with Buyer in the defense or prosecution of any litigation or proceeding which may be instituted hereafter against Buyer or Buyer's assigns on account of enforcement of Buyer's ownership of the Assets and the enforcement of payment from the related Payor thereof, and the Providers shall indemnify Buyer and its agents or assigns for any loss or expense including their reasonable attorneys' fees incurred by such parties relating to or arising out of the Providers' billing, administration or handling of the Assets prior to or after the Closing Date related thereto.
- 5.8 Delivery of Financial Statements and Tax Returns. So long as this Agreement is in effect, each Provider shall deliver to the Buyer: (i) within 45 days after the end of each fiscal quarter, the Providers' financial statements for such period and for that portion of their fiscal year through the end of such period; (ii) within 150 days after the end of the Providers' fiscal year, the Providers' annual financial statements for such year certified by the Providers' chief financial officers; (iii) when filed, copies of all federal, state tax returns, request for extension or any other returns and, when received, copies of all notices from any federal, state or local agencies with respect to Taxes and pension or profit-sharing plans, (iv) when received by the Providers, all Medicare and Medicaid cost reports and audits, (v) when received by the Providers, any other financial or quality of professional service audits or surveys, (vi) weekly reports on financial condition of the type available to management of the Providers, and (vii) promptly upon request, such other information concerning the Providers as the Buyer may from time to time request. All financial statements delivered to Buyer shall be prepared in accordance with generally accepted accounting principles and on a basis consistent with those previously submitted to the Buyer. The Providers shall file all federal, state and local tax returns and pay all Taxes when due.
- 5.9 No Change of Address. The Providers shall not change their names, mailing addresses, chief executive offices, principal places of business or places where such records are maintained without 30 days prior written notice to Buyer.
- 5.10 Sale of Accounts to be Reflected on Providers' Books and Records. The Providers will reflect on all of their books, records, tax filings and financial statements, and in all its dealings with the Payors of such Accounts, that it has sold the Accounts and related

Assets to Buyer and shall treat and characterize all Purchases as sales of the Accounts and related Assets for accounting and tax purposes. Each of the Providers hereby affirms that they have valid business reasons for selling the Accounts to the Buyer as contemplated by this Agreement rather than obtaining a loan with the Accounts being utilized as collateral therefor.

- 5.11 <u>Proceeds</u>. The proceeds of the sale of the Accounts will be used for the business and commercial purposes of the Providers.
- 5.12 No Proceedings. Each of the Providers hereby agrees that it will not institute against Buyer or join any other person or entity in instituting against Buyer, any bankruptcy, reorganization, arrangement, insolvency or liquidation proceeding, or other proceeding under any federal or state bankruptcy or similar law.

5.13 Conduct of Business and Maintenance of Existence, Permits, Licenses, Compliance with Law and Material Agreements.

- (a) The Providers will continue to engage in business of the same general type as they now conduct and will preserve, renew and keep in full force and effect their existence and their respective rights, privileges and franchises necessary or desirable in the normal conduct of business. The Providers shall not merge, consolidate or sell all or substantially all of their assets without the express prior written consent of Buyer. The Providers shall not change their respective jurisdictions of incorporation or organization without the express prior written consent of Buyer.
- (b) The Providers will comply with all requirements of all applicable Medicare and Medicaid laws, ordinances, rules, regulations, and requirements and all other Laws. The Providers will keep all of their Permits and Licenses in full force and effect and obtain any renewals of such Permits and Licenses and any additional Permits and Licenses that are necessary or desirable for the continued conduct of Providers' businesses as presently conducted. The Providers will not permit an event of default to occur under any of their Material Agreements.
- (c) The Providers will immediately forward to Buyer any notices or correspondence received by the Providers with respect to its failure to comply with any Laws, the limitation, restriction, forfeiture or untimely lapse of any Permit or License and its default under any Material Agreement.

5.14 Notice of Noncompliance.

The Providers will provide immediate written notice to Buyer of any default in any covenant set forth in this article 5 and of any delinquency in any obligation owed by the Providers to DRS, DSS, DPH, HUD, or any other governmental entity or any creditor or third party of any material threatened action, investigation or penalty made by any governmental entity or any creditor.

5.15 Additional Information.

The Providers will promptly provide to Buyer such additional information regarding their business, operations and financial condition as Buyer may reasonably request. Providers will further provide notice to Buyer of any pending controversies or claims involving HCI.

6. Indemnification and Claims.

- demand jointly and severally indemnify and hold harmless Buyer and its officers, directors, shareholders, employees, agents and assigns (each an "Indemnified Party") from and against any and all claims, losses, damages, liabilities and expenses (including, without limitation, settlement costs and any legal, accounting and other expenses for enforcing Buyer's rights and remedies hereunder, or for investigating, prosecuting or defending any actions or threatened actions) (hereinafter "Indemnified Claims") awarded or incurred by any of them arising out of or relating to this Agreement or the transactions contemplated hereby or the use of proceeds therefrom, together with interest on cash disbursements in connection therewith at the Default Rate from the date cash disbursements were made or incurred by any Indemnified Party until paid in full by the Providers, including without limitation Indemnified Claims relating to each and all of the following:
- (a) Any breach or alleged breach of any representation or warranty made by any of the Providers in this Agreement or any Exhibit to this Agreement, with respect to any Account, or in any document delivered in connection herewith;
- (b) Any breach or alleged breach of any covenant, agreement or obligation of any of the Providers contained in this Agreement, any Exhibit to this Agreement or document provided in connection herewith (including any list of Accounts submitted to Buyer by Providers pursuant to Section 2.1), or any other instrument contemplated by this Agreement;
- (c) Any misrepresentation or alleged misrepresentation contained in any statement or certificate furnished by any of the Providers pursuant to this Agreement (including any list of Accounts submitted to Buyer by Providers pursuant to Section 2.1), or in connection with the transactions contemplated by this Agreement;
 - (d) Any Retained Liabilities;
- (e) Any fees and expense of any broker, investment banker or finder with whom it has dealt in connection with the purchase and sale of the Accounts hereunder or the transactions contemplated hereby; and
- 6.2 <u>Claims for Indemnification</u>. Whenever any claim shall arise for indemnification hereunder, the Buyer shall promptly notify the Providers of the claim and, when known, the facts constituting the basis for such claim. In the event of any claim for

indemnification hereunder resulting from or in connection with any claim or legal proceedings by a third party, the notice to the Providers shall specify, if known, the amount or an estimate of the amount of the liability arising therefrom. The Buyer shall not settle or compromise any claim by a third party for which it is entitled to indemnification hereunder, without the prior written consent of the Providers (which shall not be unreasonably withheld) unless the Providers shall have failed to pay indemnification obligations as they accrue or in the event that suit shall have been instituted against them the Providers shall not have taken control of such suit after notification thereof as provided in Section 6.3 of this Agreement.

- Defense by Indemnifying Party. In connection with any claim giving 6.3 rise to indemnity hereunder resulting from or arising out of any claim or legal proceeding by a person who is not a party to this Agreement against the Buyer, the Providers at their sole cost and expense may, upon written notice to the Buyer, assume the defense of any such claim or legal proceeding if it acknowledges to the Buyer in writing its obligations to indemnify the Buyer or any other Indemnified Party with respect to all elements of such claim. The Buyer shall be entitled to participate in the defense of any such action, with its counsel and at its own expense. If the Providers shall not have retained counsel to have charge of the defense of any such claim or litigation resulting therefrom, or if the Buyer or any Indemnified Party shall reasonably conclude that there may be defenses available to it that are different from or additional to those available to the Providers, the Providers shall not have the right to direct the defense of action on behalf of the Buyer or any such Indemnified Party and the Buyer or any such Indemnified Party shall have the right to retain separate counsel in any such action and to participate in the defense thereof in such manner as it may deem appropriate (including but not limited to, settling such claim or litigation, after giving notice of the same to the Providers, on such terms as the Buyer or such Indemnified Party, as the case may be, may deem appropriate) and the reasonable fees and expenses of such counsel shall be assumed by the Providers. If the Buyer or such Indemnified Party defends or settles such third party claim, the Providers shall have the burden to prove by a preponderance of the evidence that the Buyer or such Indemnified Party did not defend or settled such third party claim in a reasonable manner.
- 6.4 Manner of Indemnification. All amounts payable with respect to the Providers' indemnification obligations hereunder shall be due upon demand by the Indemnified Party and shall, at the option of the Buyer, be paid by (i) cash or delivery of a certified or official bank check by one or more of the Providers in the amount of the indemnification obligation, (ii) deduction from the Initial Installment pursuant to Section 3.1(c), or (iii) from the Reserve Account pursuant to Section 3.2(c).

7. Additional Security; Rights of Buyer as Secured Party.

As security for the Providers' payment and performance of all Obligations and its repurchase obligations under Sections 3.3(c) and 8.2, the Providers hereby grant to the Buyer a security interest in the Collateral. The Providers shall execute such financing statements and other documentation as Buyer may request from time to time in order to perfect and protect

Buyer's security interest in the Collateral. The security interests in the Collateral under this Section 7 and under Sections 3.3(c) and 8.2 shall be on the following terms and conditions:

- 7.1 Recourse to security shall not at any time be required and the Providers shall at all times remain liable for the repayment upon demand of all Obligations and its repurchase obligations under Sections 3.3(c) and 8.2. The Providers represent, warrant and covenant that Buyer shall have a first priority security interest in all of the Accounts and Related Property of the Accounts and Collateral in which a security interest is being granted hereunder, subject to the permitted liens in favor of HUD set forth in the Intercreditor Agreement and/or listed in **Exhibit A**. During the term of this Agreement, the Providers shall not sell or assign, negotiate, pledge or grant any security interest in, or permit any Liens to be placed upon, any of the Accounts and Related Property of the Accounts and Collateral to anyone other than Buyer.
- 7.2 The security interest shall secure any and all of the Providers' Obligations and their repurchase obligations under Section 2.3(a) and (b) above. The Providers hereby waive any and all suretyship defenses available to the Providers under applicable law.
- 7.3 Providers hereby authorize Buyer to file UCC-1 Financing Statements with respect to the Collateral, and any amendments or continuations relating thereto, without the signatures of Providers and hereby ratify, confirm and consents to any such filings made by Buyer prior to the date hereof. Providers shall not allow any financing statement (other than that filed by or on behalf of Buyer) to be on file in any public office covering any Collateral or the proceeds thereof, except as set forth on Schedule 7.3. Providers will do all reasonable lawful acts which Buyer deems necessary or desirable to protect the security interest or otherwise to carry out the provisions of this Agreement, including, but not limited to, the execution of all documents, instruments and agreements in form necessary to Buyer. The Providers irrevocably appoints Buyer as their attorney in fact during the term of this Agreement to do all acts which they may be required to do in connection with the creation and perfection of Buyer's security interest under this Agreement, such appointment being deemed to be a power coupled with an interest.
- 7.4 Notwithstanding the creation of the above security interest, the relationship of the parties shall be that of seller and purchaser of Account and not that of lender and borrower. Nothing in this Section shall be construed to limit the rights that Buyer has as purchaser of the Accounts and Related Property pursuant to the other provisions of this Agreement or at law.
- 7.5 Without limiting any other rights of Buyer in the Accounts, Related Property of the Accounts and Collateral, immediately upon the occurrence of any Event of Default, Buyer may exercise any and all rights of a secured party under the Uniform Commercial Code other applicable law and, in addition, to the extent permitted by law: may (a) remove from any premises where same may be located any and all documents, instruments, files and records, and any receptacles or cabinets containing same, relating to the Accounts, or Buyer

may use, at Providers' expense, such of Providers' personnel, supplies or space at Providers' places of business or otherwise, as may be necessary to properly administer and control the Accounts or the handling of collections and realizations thereon; (b) receive, take, endorse, assign, deliver, accept and deposit, in the name of Buyer or the Providers, any and all cash, checks, commercial paper, drafts, remittances and other instruments and documents relating to the Accounts or the proceeds thereof; (c) notify any Payor obligated with respect to any Accounts, that such Accounts have been assigned to Buyer by the Providers and that payment thereof is to be made to the order of and directly and solely to Buyer, (d) communicate directly with Payor to verify the amount and validity of any Accounts created by the Providers; (e) bring suit, in the name of any of the Providers, and generally shall have all other rights respecting said Accounts, including without limitation the right to accelerate or extend the time of payment, settle, compromise, release in whole or in part any amounts owing on any Accounts and issue credits in the name of the Providers; (f) sell, assign and deliver the Accounts and any returned, reclaimed or repossessed merchandise, with or without advertisement, at public or private sale, for cash, on credit or otherwise, at Buyer's sole option and discretion, and Buyer may bid or become a purchaser at any such sale, free from any right of redemption, which right is hereby expressly waived by the Providers; (g) foreclose the security interests created herein by any available judicial procedure, or take possession of any or all of the inventory without judicial process, and enter any premises where any inventory may be located for the purpose of taking possession of or removing the same and (h) exercise any other rights and remedies provided in law, in equity, by contract or otherwise. Buyer shall have the right, without notice or advertisement, to sell, lease, or otherwise dispose of all or any part of the Collateral whether in its then condition or after further preparation or processing, in the name of the Providers or Buyer, or in the name of such other party Buyer may designate, either at public or private sale or at any broker's board, in lots or in bulk, for cash or for credit, with or without warranties or representations, and upon such other terms and conditions as Buyer in its sole discretion may deem advisable, and Buyer shall have the right to purchase at any such sale. If any equipment or merchandise repossessed by Buyer shall require rebuilding, repairing, maintenance or preparation, Buyer shall have the right, at its option, to do such of the aforesaid as is necessary, for the purpose of putting the equipment and merchandise in such saleable form as Buyer shall deem appropriate. If notice of intended disposition of any Collateral is required by law, it is agreed that ten (10) days' notice shall constitute reasonable notification and full compliance with the law. The net cash proceeds resulting from the Buyer's exercise of any of the foregoing rights, (after deducting all charges, costs and expenses, including reasonable attorneys' fees) shall be applied by Buyer to the payment of Providers' Obligations and their repurchase obligations under Sections 3.3(c) and 8.2, whether due or to become due, in such order as Buyer may elect, and Providers shall remain liable to Buyer for any deficiencies, and Buyer in turn agrees to remit to Providers or their successors and assigns, any surplus resulting therefrom. The enumeration of the foregoing rights is not intended to be exhaustive and the exercise of any right shall not preclude the exercise of any other rights, all of which shall be cumulative. Nothing in this Section shall be construed to limit the Providers' exercise of their rights as set forth elsewhere in this Agreement prior to the occurrence of any Event of Default in, and with respect to, any Accounts.

- 7.6 The security interest created under this Section 7 in the Accounts and Related Property shall not be merged or extinguished upon the sale of the Accounts to Buyer hereunder.
- 7.7 Nothing herein shall limit the exercise by Buyer of any other legal or equitable rights that Buyer may have as a secured party.

8. Events of Default and Remedies

- 8.1 Notwithstanding anything hereinabove to the contrary, Buyer may exercise any or all of the remedies set forth in Section 8.2, immediately upon the occurrence of any of the following (herein "Events of Default"):
- (a) Cessation of any material part of the business of any of the Providers or the calling of a meeting of the creditors of any of the Providers for purposes of compromising the debts and obligations of the Providers;
- (b) The occurrence of an Insolvency Proceeding with respect to any of the Providers or any affiliate of the Providers;
- (c) The breach by any of the Providers of any representation, warranty, or covenant contained herein (other than those referred to below) or in any other written agreement between the Providers and Buyer, provided that such breach by the Providers of any of the representations, warranties or covenants referred in this clause (iv) shall not be deemed to be an Event of Default unless and until such breach shall remain uncured or unremedied (if capable of being remedied) to Buyer's satisfaction for a period of ten (10) days from the date Buyer gives the Providers notice of such breach, except that no notice of breach or cure period shall apply to any failure by Providers to provide required notices to Buyer;
 - (d) Any Event of Default occurs under the terms of any Guaranty;
- (e) Any default occurs, which is not cured within any applicable grace period or cure period or waived, (A) in the payment of any amount with respect to any Indebtedness (other than the Obligations) of the Providers in excess of \$100,000.00; or (B) in the performance, observance or fulfillment of any provision contained in any agreement, contract, document or instrument to which any of the Providers is a party or to which any of their properties or assets are subject or bound under or pursuant to which any Indebtedness was issued, created, assumed, guaranteed or secured and such default continues for more than any applicable grace period or permits the holder of any Indebtedness to accelerate the maturity thereof;
- (f) Any Indebtedness of any of the Providers is declared to be due and payable or is required to be prepaid (other than by a regularly scheduled payment) prior to the stated maturity thereof, or any obligation of such Person for the payment of Indebtedness (other

than the Obligations) is not paid when due or within any applicable grace period, or any such obligation becomes or is declared to be due and payable before the expressed maturity thereof, or there occurs an event which, with the giving of notice or lapse of time, or both, would cause any such obligation to become, or allow any such obligation to be declared to be, due and payable;

- (g) Any of the Providers shall, directly or indirectly, instruct any Payor (except as otherwise expressly set forth in this Agreement) to mail or deliver payment on any Accounts to any address or bank account other than Buyer's Lock Box Account, or, in the case of Government Receivables, the Governmental Lockbox Account;
- (h) Any of the Providers fails to remit to Buyer any check, replacement fund, or other payment it receives with respect to an Account within three (3) Business Days of receipt or the discovery of same;
- (i) There has been a breach of representations or warranties affecting Accounts with an aggregate Outstanding ENR Balance of more than \$50,000;
- (j) There has been a change of control of any of the Providers, which shall include, without limitation, a direct or indirect sale or transfer of 50% or more of the capital stock, membership interests or other economic or voting interests in such Providers;
- (k) Failure on the part of any Provider to promptly remit any sums payable by it hereunder when due, including without limitation any repurchase obligation
- The operating license, Medicare provider number or provider agreement, or Medicaid provider number or provider agreement of any Provider is suspended or terminated;
- (m) the Cut-Off Time occurs under the Intercreditor Agreement among the Providers and the United States of America acting through the Secretary of Housing and Urban Development, or such agreement is terminated or modified without the prior written consent of the Buyer,
- (n) any right of set-off or recoupment is exercised or notice of proposed exercise is given with respect to Accounts by the DSS or any other payor or third party (other than as set forth in the DSS Letter with respect to setoffs up to a monthly limit of \$100,000)
- (o) any Lockbox Account or account control agreement with respect to any of the Accounts is terminated or modified without the prior written consent of the Buyer.

Upon the occurrence of an Event of Default, at the option of Buyer (i) all 8.2 Obligations or the Obligations designated by Buyer shall become immediately due and payable; (ii) Interest shall accrue on all outstanding or thereafter accrued Obligations at the Base Rate and Buyer may elect to charge Providers interest at the Default Rate rather than at the Base Rate on all then outstanding or thereafter incurred Obligations; (iii) Providers shall be required to repurchase all outstanding Accounts (other than Accounts as to which the Providers establish to the satisfaction of Buyer are unpaid as a result of the bankruptcy, insolvency or financial inability to pay of the third party Payor with respect thereto) at a repurchase price equal to (A) the aggregate Outstanding ENR Balance of all Batches of Accounts, plus (B) the amount of all fees, costs and expenses, that have been paid out of the proceeds of the Accounts, plus (C) the amount of any other outstanding Obligations, plus (D) the amount of any reasonable reserve requested by Buyer, plus (E) interest at the Base Rate, or at Buyer's election, at the Default Rate on the Outstanding Initial Installment with respect to such Accounts from the date such repurchase obligation arose; (iv) Buyer may perform, at Providers' cost and expense, which shall be immediately due and payable, any or all of the covenants and agreements of Providers contained in this Agreement, (v) Buyer may bill, service, or collect any and all of the Accounts and Providers shall pay Buyer's customary rates therefor, or (vi) Buyer may immediately terminate its obligations under this Agreement. The exercise of any option is not exclusive of any other option or remedy whether pursuant to this Agreement or otherwise which may be exercised at any time or at the same time by Buyer.

9. Term and Termination.

- 9.1 The Providers' obligations to submit Eligible Accounts for purchase hereunder shall be for a period of one (1) year after the initial Closing Date (the "Term") and during any Extended Term. The initial Term shall be automatically extended for successive one year periods (each, an "Extended Term") unless the Providers notify Buyer, not less than ninety (90) days prior to the end of the then current Term or Extended Term, as the case may be, that it does not wish to extend such Term or Extended Term. All other terms and provisions of this Agreement shall remain in full force and effect until this Agreement is terminated. This Agreement shall be terminated as follows:
- (a) Upon the execution and delivery of the express written consent of the Providers and Buyer to such termination.
- (b) Subject to Section 9.2, below, by either the Providers or Buyer at any time after (i) the aggregate Outstanding Initial Installments of all Batches shall have been reduced to zero; (ii) all of the Providers' Obligations under this Agreement, including any interest accrued thereon, shall have been paid to Buyer in full; (iii) there is no Insolvency Proceeding pending or threatened against Buyer, and (iv) there are no pending or threatened claims against Buyer arising out of this Agreement or Buyer's purchase of Accounts hereunder.

9.2 In the event this Agreement is terminated prior to the end of the Term or any Extended Term for any reason other than voluntary termination by Buyer alone, Providers shall pay to Buyer an early termination fee equal to the greater of (A) three percent (3)% of the Facility Cap as of the proposed date of termination or (B) the aggregate of (i) all Funding Fees, Servicing Fees and Origination Fees payable hereunder from and after the Closing Date to the proposed date of termination, divided by (ii) the number of calendar days that have elapsed since the Closing Date to the proposed date of termination, and then multiplied by (iii) the number of calendar days left from the proposed termination date to the end of the Term or the end of the then-current Extended Term. In the event Buyer exercises any of the remedies set forth in Section 8.2 following an Event of Default, Providers shall pay to Buyer a default termination fee in the amount of five percent (5)% of the Facility Cap as of the date of such Event of Default.

10. Miscellaneous.

- 10.1 Notices. Any notice or request under this Agreement shall be given to any party to this Agreement at such party's address set forth beneath its signature on the signature page to this Agreement, or at such other address as such party may hereafter specify in a notice given in the manner required under this Section 10.1. Any notice or request hereunder shall be given only by, and shall be deemed to have been received upon: (i) registered or certified mail, return receipt requested, on the date on which such received as indicated in such return receipt, (ii) delivery by a nationally recognized overnight courier, one (1) Business Day after deposit with such courier, or (iii) facsimile or electronic transmission, in each case upon telephone or further electronic communication from the recipient acknowledging receipt (whether automatic or manual from recipient), as applicable.
- 10.2 Assignability; Parties in Interest; Participations. The parties hereto acknowledge and agree that this Agreement, including all rights and obligations contained herein, may be sold, assigned or otherwise transferred, in whole or in part, by Buyer without the consent of the Providers. This Agreement is not assignable by the Providers. This Agreement shall inure to the benefit of and be binding upon Buyer and the Providers, and their respective permitted successors and assigns. This Agreement shall not benefit or create any right or cause of action in any or on behalf of any person or entity other than the parties hereto and their respective permitted successors and assigns. This Agreement shall not benefit or create any right or cause of action in any or on behalf of any person or entity other than the parties hereto and their respective permitted successors and assigns.
- 10.3 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, including by facsimile, each of which shall be deemed an original, but all of which shall constitute but one and the same Agreement.
- 10.4 <u>Severability</u>. Any provision of this Agreement which is invalid, illegal, or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of

such invalidity, illegality, or unenforceability, without affecting in any way the remaining provisions hereof in such jurisdiction or rendering that or any other provision of this Agreement invalid, illegal, or unenforceable in any other jurisdiction.

- 10.5 <u>Due Diligence Investigation</u>. All representations and warranties contained herein which are made to the best knowledge of a party shall require that such party make reasonable investigation and inquiry with respect thereto to ascertain the correctness and validity thereof. No investigation or inquiry made by or on behalf of Buyer shall in any way affect or lessen Buyer's right to rely (or the reasonableness of Buyer's reliance) on the representations, warranties and covenants made and entered into by the Providers hereunder.
- 10.6 <u>Construction</u>. This Agreement shall, in all cases, be construed simply, according to its fair meaning, and not strictly for or against either party. Any section and paragraph headings contained in this Agreement are for convenience of the reference only and shall not affect the construction or interpretation of this Agreement.
- 10.7 <u>Survival of Representations and Warranties</u>. All representations, warranties, covenants and indemnities made by the parties in this Agreement or in any instrument or document furnished in connection herewith shall survive the initial Closing and any subsequent Closing for the sale of Accounts to Buyer.
- 10.8 Applicable Law; Jurisdiction. This Agreement shall be governed by and construed in accordance with the internal substantive law, and not the choice of law rules, of the State of Connecticut. The Providers hereby consent to, and agrees to submit to the jurisdiction of the Superior Court of the State of Connecticut and the United States District Court for the District of Connecticut, and agrees, at the Buyer's election, that any legal action or proceeding arising out of or with respect to this Agreement shall be brought by the Providers and the Buyer in the Superior Court of the State of Connecticut or in the United States District Court for the District of Connecticut, in each case, that are located in New Haven County, as the Buyer shall elect. Each of the parties hereto irrevocably consents to the service of any and all process in any such action or proceeding brought in the Superior Court of the State of Connecticut or the United States District Court for the District of Connecticut by the delivery of copies of such process to such party at its address specified below its signature.
- Provider hereby (a) irrevocably waives, to the maximum extent not prohibited by law, any right it may have to a trial by jury in respect of any litigation directly or indirectly at any time arising out of, under or in connection with this Agreement or any transaction contemplated hereby or associated herewith; (b) irrevocably waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any such litigation any special, exemplary, punitive or consequential damages, or damages other than, or in addition to, actual damages; and (c) certifies that no party hereto nor any representative or agent or counsel for any party hereto has represented, expressly or otherwise, or implied that such party would not, in the event of litigation, seek to enforce the foregoing waivers.

- 10.10 <u>Complete Agreement</u>. This Agreement, any list of Accounts submitted by Providers to Buyer pursuant to Section 2.1, the exhibits and schedules hereto and the documents delivered or to be delivered pursuant to this Agreement set forth the entire understanding and agreement of the parties hereto with respect to the transactions contemplated herein and may not be contradicted by evidence of prior, contemporaneous, or subsequent oral agreements of the parties, including, without limitation, any term sheet or indication of interest provided by Buyer to Providers. No modification or amendment of or supplement to this Agreement or such other documents shall be valid or effective unless the same is in writing and signed by the party against whom it is sought to be enforced.
- 10.11 Equitable Relief. In the event any of the Providers commits any act or omissions which (i) prevents or unreasonably interferes with: (a) Buyer's exercise of the rights and privileges arising under the power of attorney granted in Section 5.7 of this Agreement; or (b) Buyer's perfection of or levy upon the ownership or security interest granted in the Accounts, including any seizure of any Account, or (ii) constitutes a breach of any of its representations, warranties or covenants hereunder, such conduct will cause immediate, severe, incalculable and irreparable harm and injury, and shall constitute sufficient grounds to entitle Buyer to an injunction, writ of possession, or other applicable relief in equity, and to make such application for such relief in any court of competent jurisdiction, without any prior notice to such Provider. The Providers agree that they shall not defend against any such relief on the basis that Buyer has adequate remedies at law or for damages.
- 10.12 <u>Cumulative Rights</u>. All rights, remedies and powers granted to Buyer in this Agreement, or in any other instrument or agreement given by the Providers to Buyer, are cumulative and may be exercised singularly or concurrently with such other rights as Buyer may have. These rights may be exercised from time to time as to all or any part of the Accounts purchased hereunder as Buyer in its discretion may determine. Buyer may not waive its rights and remedies unless the waiver is in writing and signed by Buyer. A waiver by Buyer of a right or remedy under this Agreement on one occasion is not a waiver of the right or remedy on any subsequent occasion.
- Providers agree to reimburse Buyer upon demand for all reasonable attorney's fees, court costs and other expenses incurred by Buyer in negotiating, administering, terminating or enforcing this Agreement and protecting or enforcing its interest in the Accounts or the Assets, in collecting the Accounts or the Assets, or in the representation of Buyer in connection with any bankruptcy case or insolvency proceeding involving the Providers, the Assets, any Payor, or any Account and in connection with any matters relating to HCI. The Providers shall also be responsible for all reasonable out-of-pocket costs and expenses incurred by Buyer arising out of or related to (i) the documentation of the transactions contemplated by the Agreement including any amendment, supplement or modification of such documentation, which costs and expenses may include due diligence costs and legal fees and expenses; (ii) the transfer of the Assets to Buyer pursuant to the terms of this Agreement, including, without limitation, all sales and

transfer taxes, together with all other transfer or recordation fees and expenses and any legal fees and costs associated therewith, lien searches, and filing and recording fees; and (iii) the administration of the transactions contemplated hereby, including but not limited to, internal and outside legal fees, and all other out-of-pocket expenses.

- 10.14 <u>Interest</u>. If any Obligation of the Providers hereunder is not paid when due, such Obligation shall bear interest at the Base Rate (or if the Buyer so elects, at the Default Rate) until payment in full of such obligation to Buyer. Any amounts uncollected by the Buyer on a Purchased Account thirty days after purchase, by reason of interception by the prior factor or otherwise, shall be immediately repurchased by the Providers. In furtherance thereof the parties stipulate and agree that none of the terms and provisions contained in this Agreement shall ever be construed to provide for payment of interest by Providers in excess of the maximum amount of interest permitted to be charged by applicable law from time to time in effect. Providers shall not ever become liable for payment of any Obligation hereunder, shall ever be liable for accrued interest thereon, or shall ever be required to pay interest thereon in excess of the maximum amount that may be lawfully charged under applicable law from time to time in effect and the provisions of this Section 10.14 shall control over all other provisions of this Agreement. Any interest that may be collected that is in excess of the maximum amount of interest permitted to be charged by applicable law from time to time in effect shall be credited to the Reserve Account established pursuant to Section 3.2 for disposition as therein provided.
- 10.15 <u>Wire Transfer Fees.</u> Payment of each Initial Installment and all Subsequent Installments shall be made by wire transfer in accordance with the wire instructions set forth in <u>Exhibit C</u> hereto. Providers shall pay a wire transfer fee of \$25.00 for each wire transfer of funds to the Providers; provided, however, if any of Providers request wire transfers more frequently than once a week, Providers shall pay a wire transfer fee of \$25.00 for each such additional wire transfer.
- 10.16 Further Assurances. From time to time on and after the Closing Date, the Providers shall immediately execute and deliver to Buyer such instruments of sale, transfer, conveyance, assignment and delivery consents, assurances, powers of attorney and other instruments as may be requested by Buyer in order to vest in Buyer all right, title and interest of the Providers in and to the Assets and otherwise in order to carry out the purpose and intent of this Agreement.

IN WITNESS WHEREOF, the undersigned have executed this Agreement as of the date first above written.

PROVIDERS:

HEALTH CARE INVESTORS, INC. d/b/a Alexandria Manor

By: Name:

Title:

HEALTH CARE ALLIANCE, INC.

d/b/a Blair Manor

By:

Name:

HEALTH CARE ASSURANCE, LLC

d/b/a Douglas Manor

By:

Name: Title:

HEALTH CARE RELIANCE, LLC

d/b/a Ellis Manor

By:

Namé: Title:

BUYER:

REVENUE MANAGEMENT

SOLUTIONS, LLC

By:

Name: Granfonguo GAILUZZ

Title:

Address:

Purchase Agreement, Page 39

SCHEDULE OF EXHIBITS

EXHIBIT A: List of Outstanding Liens, including all Financing Statements and Deposit Account Control Agreements

EXHIBIT B: Form of Notice to Payors

EXHIBIT C: Wire Instructions

EXHIBIT D: Form of Board Resolutions

EXHIBIT E: Form of Guaranty

EXHIBIT F: Form of Landlord Waiver and Consent

EXHIBIT G: IRS Form 8821

Exhibit 8

Medicaid Billing & Collections Authorization Letter

Letterhead of Provider

DATE

COMPANY NAME ADDRESS CITY, ST ZIP

To whom it may concern:

Ladies and gentlemen, I hereby authorize Revenue Management Solutions, LLC, a Connecticut limited liability company ("RMS"), and its agents, to submit billings and claims to you, and receive collections in respect of, all accounts receivable arising from all medical and healthcare services rendered by [Provider's Name], Medicaid Provider Number _____ under your contract with us.

This authorization is irrevocable, absent the written consent of RMS.

Sincerely,		
[Provider	's Name]	
A	[corporation]	
Ву:		
	ipal's Name]	
Ite: T	NT0	

Exhibit D C

Provider's Wire Instructions for Disbursement of all Payments

Revenue Management Solutions, LLC, a Connecticut limited liability company ("RMS"), is hereby instructed to deliver all payments to Provider, in the form of wire transfer of funds to:

Affinity Health Care Management, Inc, a New York corporation	
Actual Account Name	
in the care of the following bank:	
Bank Name: Proples United Bank	
Bank Branch Address: 850 Main Street	
Street Address	
Bridgeport CT 6664 City State Zip Code	7.0
And the second s	
Bank Branch Phone Number: 203-338-4603	
ABA: 221172186	(
Account Number: 6500292274 (Affinity Health Care - MAS	ika)
Name of Bank Officer to contact: Kon Khong Kham	
Telephone number of Bank Officer: 263-338-4603	
*	
E Company of the Comp	
Provider: [Name of Provider]	
Ву:	
[Principal's Name] Its: Title	



People's United Bank accepts many different kinds of electronic payments from both domestic and international Banks. The method that your client will pay you may vary and require different instructions for them to route your payment to People's United without any unnecessary delays. On this product sheet we have listed the payment instructions for the different payment types. Please provide this sheet to your client as a guide to sending payments.

Wire Transfer Instructions - Domestic

Receiving Bank ABA/Routing Number:

221172186

Receiving Bank Name:

People's United Bank

Receiving Bank Address:

Bridgeport, CT

Beneficiary Account Number: Beneficiary Name:

<Your account Number>

<Company name>

Wire Transfer Instructions - International

People's United Bank as established a relationship with the Bank of New York as our intermediary bank in order to receive payments via SWIFT.

You may instruct your customers requesting a SWIFT address to send the payment as follows:

SWIFT Address:

IRVTUS3N

The "account with institution" A/C# and Name:

8900684356 People's United Bank

Your account number:

(exactly as it reads at People's United Bank)

Your company name:

(exactly as it reads at People's United Bank)

Note: This swift address belongs to our intermediary Bank, The Bank of New York/Chase

SWIFT formatted fields:

(56) IRVTUS3N

57} 8900684356

Peoples United Bank

(59)

Bene a/c # and Bene exact name

Automated Clearing House (NACHA, ACH, EFT, Vendor Payments)

Receiving Bank ABA/Routing Number:

221172186

Receiving Bank Name:

People's United Bank

Receiving Bank Address:

Bridgeport, CT

Account Number:

<Your account number>

Company Name:

<Company name>

Note: People's United can accept electronic payments in the following formats: CCD, CCD+, CTX, CTP for corporate to corporate payments. However, PPD and PPD+ are accepted, but should only be used by the sending party for consumer transactions.

Kon Khongkham Vice President Treasury Management Officer Treasury Management Services



850 Main Street 12th Floor - RC 285 Bridgeport, Connecticut 06604

T: 203,338.4603 F: 203,615.9154 E: kon.khongkham@peoples.com Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 93 of 102

Internal Revenue Service Instead, use Form 4506 or Form 4506-T. 1 Taxpayer information. Taxpayer(s) must sign and date this form on line 7. Taxpayer name(s) and address (type or print) Health Care Assurance, LLC. d/b/a Douglas Manor 103 North Road Windham, CT 06280 2 Appointee. If you wish to name more than one appointee, attach a list to this form.							
Do not sign this form urses of applicable lines have been completed. Do not use his form to request a copy or transcript of your tax return. Do not use his form to request a copy or transcript of your tax return. Do not use his form to request a copy or transcript of your tax return. Do not use his form to request a copy or transcript of your tax return. Do not use his form to request a copy or transcript of your tax return. Do not use his form to request a copy or transcript of your tax return. Do not use his form to file of the property of the proper	8821		Tax Information	n Auth	orization		For IRS Use Only
Department of the Treasury Teappyer information. Taxpayer(s) must sign and date this form on line 7. Taxpayer information. Taxpayer(s) must sign and date this form on line 7. Taxpayer information. Taxpayer(s) must sign and date this form on line 7. Taxpayer information number 06-1478014 Department place to the Treasury Department of the Treasu	CENTRAL RESIDENCE SERVICES						
Department of the Trawary Internation (the Trawary) Internation (the T	(Rev. October 2011)						0.0000
1 Tappyer information. Taxpayer(s) must sign and date this form on line 7. Tappyer information. Taxpayer(s) must sign and date this form on line 7. Tappyer information. Taxpayer(s) must sign and date this form on line 7. Tappyer information. Taxpayer(s) must sign and date this form on line 7. Tappyer information number 68-1479014 Daytime legiption enumber 880-423-4388 2 Appointee. If you wish to name more than one appointee, attach a list to this form. Name and address CAF No. PTIN Telephone No. Fax No. Check if new: Address Telephone No. Telepho	Department of the Treasury	- 50					March March March
Texpayer identification number Designation Designation							Date
Health Care Assurance, L.C. dib/a Douglas Manor 103 North Road Windham, CT 06280 2. Appointee. If you wish to name more than one appointee, attach a list to this form. CAF No.			er(s) must sign and date this fo	rm on line			
Desyrtes Nethoptone number (# applicable) 2 Appointee. If you wish to name more than one appointee, attach a list to this form. Name and address CAF No. PTIN Telephone No. Fax No. Check if new: Address	Taxpayer name(s) and address	ss (type or print)			Taxpayer identification num	per	
Specific use not recorded on Centralized Authorization File (CAF). If the tax information authorization is for a specific use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6. Specific use not recorded on CAF, check this box. Ship they would not want any copies of notices or communications and other related materials with the notices. Signature of tax payer(s). If at ax matters you listed on line 3 above unless you checked this box. Signature of tax payer(s). If at ax matters poles to a loint return, either husband or wife must sign. If signature in each of the same tax matters you listed on line 3 above unless you checked this box. To revoke this tax information authorization, you must atter applies to a joint return, either husband or wife must sign. If signed by a corporate office, partner, guardian, executor, receiver, administrator, trustee, or pay other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. Print Name Title (fi applicable) Title (fi		, LLC. d/b/a Dou	glas Manor		0	6-1479014	
2 Appointee. If you wish to name more than one appointee, attach a list to this form. CAF No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Telephone No. Fax No. Telephone No. T					Daytime telephone number	Plan n	umber (If applicable)
CAF No. Trib Telephone No. Fax No. Telephone No. Fax No. Telephone No. Fax No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Telephone No. Telephone No. Fax No. Telephone No. Fax No. Telephone No. Fax No. Telephone No. Fax No. Telephone No. Telephone No. Fax No. Telephone No.	windnam, CT 06280				860-423-4636		
Telephone No. Fax No. Check if new: Address Telephone No. Telephone No. Fax No. Check if new: Address Telephone No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Check if new: Address Telephone No.	2 Appointee. If you	wish to name	more than one appointee, atta	ch a list t	o this form.		
Telephone No. Fax No. Check if new: Address Telephone No. Telephone No. Fax No. Check if new: Address Telephone No. Telephone No. Fax No. Check if new: Address Telephone No. Fax No. Check if new: Address Telephone No.	Name and address			CAF	lo.		
Fax No. Check if new: Address Telephone No. Fax No.				PTIN	********	***********	*************
Fax No. Check if new: Address Telephone No. Fax No.				Telepi	one No.		
Check if new: Address				LESV N	0		
3 Tax matters. The appointee is authorized to Inspect and/or receive confidential tax information in any office of the IRS for the tax matters listed on this line. Do not use Form 8821 to request copies of tax returns. (c) (d) (e) (d) (f) (income, Employment, Excise, etc.) (r) (r) (r) (r) (r) (r) (r) (Chaol	o.	alanhana I	Na CI FayNa C
tax matters listed on this line. Do not use Form 8821 to request copies of tax returns. (a) Type of Tax (income, Employment, Excise, etc.) (income, Employment, Excise, Ex	O Tay matters The						
Tax Form Number (Income, Employment, Excise, etc.) Tax Form Number (1040, 941, 720, etc.) Tax Form Number (1040, 941, 720, etc.) Tax Form Number (1040, 941, 720, etc.) Specific use not recorded on Centralized Authorization File (CAF). If the tax information authorization is for a specific use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 . Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attend a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, pariner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. Pitron Signed AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Date Title (if applicable) Title (if applicable) Title (if applicable) Title (if applicable)	tax matters listed	on this line. Do	not use Form 8821 to reques	t copies o	of tax returns.	n any οπις	[]
(Income, Employment, Exclae, etc.) (1040, 941, 720, etc.) (see the instructions for line 3) 4 Specific use not recorded on Centralized Authorization File (CAF). If the tax information authorization is for a specific use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 . ▶ □ 5 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization. This tax information authorization authorization spou want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. ▶ IF NOT SIGNED AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. ▶ DO NOT SIGN THIS FORM IF IT IS BLANK OR INCOMPLETE. Signature Title (if applicable) Title (if applicable) Title (if applicable)	Type of Ta	ıx				Specific	
4 Specific use not recorded on Centralized Authorization File (CAF). If the tax information authorization is for a specific use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6	(Income, Employment,	Excise, etc.)		(see		Specific	rax matters (see instr.)
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed AND DATED, This TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Date Date Frint Name Title (if applicable) Title (if applicable) Title (if applicable)	or Civil Pena	alty	(1040, 041, 120, 610.)	1000	and managements for into dy		
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed AND DATED, This TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Date Date Frint Name Title (if applicable) Title (if applicable) Title (if applicable)		4					
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed AND DATED, This TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Date Date Frint Name Title (if applicable) Title (if applicable) Title (if applicable)							
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed AND DATED, This TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Date Date Frint Name Title (if applicable) Title (if applicable) Title (if applicable)							
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed AND DATED, This TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Date Date Frint Name Title (if applicable) Title (if applicable) Title (if applicable)							
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed and Dated, This tax information authorization will be returned. Date Signature Date Print Name Title (if applicable) Title (if applicable)							
use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a if you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. In Not signed and Dated, This tax information authorization will be returned. Date Signature Date Print Name Title (if applicable) Title (if applicable)							
a If you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box	4 Specific use not use not recorded	recorded on on CAF, check	Centralized Authorization Fit this box. See the Instructions	le (CAF). on page	If the tax information au 4. If you check this box,	thorization skip lines 5	ls for a specific and 6 ▶ □
a If you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing basis, check this box	5 Disclosure of tay	information (rou must shock a box on line	5a or 5h i	inless the boy on line 4 is	o chocked)	
basis, check this box Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box 6 Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. b IF NOT SIGNED AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. b DO NOT SIGN THIS FORM IF IT IS BLANK OR INCOMPLETE. Date Signature Date Title (if applicable) Title (if applicable) Title (if applicable)							
Note. Appointees will no longer receive forms, publications and other related materials with the notices. b if you do not want any copies of notices or communications sent to your appointee, check this box							
6 Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. If NOT SIGNED AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Do NOT SIGN THIS FORM IF IT IS BLANK OR INCOMPLETE. Signature Date Title (if applicable) Title (if applicable) Title (if applicable) Title (if applicable)	맛이 어린 하 하는 물리를 받았다. 얼마를 맞는 것이 없다.						
6 Retention/revocation of tax Information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box To revoke this tax information authorization, see the instructions on page 4. 7 Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. If NOT SIGNED AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. Do NOT SIGN THIS FORM IF IT IS BLANK OR INCOMPLETE. Signature Date Signature Title (if applicable) Title (if applicable) Title (if applicable)							
authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box	b If you do not want	t any copies of	notices or communications se	nt to your	appointee, check this b	ox	▶ 🛚
authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect and check this box							
corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. IF NOT SIGNED AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. DO NOT SIGN THIS FORM IF IT IS BLANK OR INCOMPLETE. Signature Date Signature Date Print Name Title (if applicable) Title (if applicable)	authorizations for to revoke a prior t and check this bo	the same tax natax information	natters you listed on line 3 abo authorization, you must attac	ove unless th a copy	s you checked the box of any authorizations you	n line 4. If ou want to	you do not want remain in effect
Signature Date Signature Date Date Date Signature Date Date Date Title (if applicable) Date Date Date Date Date	corporate officer,	partner, guardia	an, executor, receiver, adminis	trator, tru	stee, or party other than	the taxpay	
Signature Date Signature Date Date Date Signature Title (if applicable) Date Date Title (if applicable)	► IF NOT SIGNED	O AND DATED	, THIS TAX INFORMATION A	UTHORIZ	ZATION WILL BE RETU	RNED.	
Signature Date Signature Date Date Date Print Name Title (if applicable) Date Date Date Date Date	NO NOT CION	THIS CORM IS	IT IS BUANK OR INCOME.	TE			
Benjamin FISCHMAN MANAGING Print Name Title (if applicable) MEMBER Print Name Title (if applicable)	PDO NOT SIGN	THIS FURIN IF	II IS BLANK OH INCOMPLE	I C.			
Benjamin FISCHMAN MANAGING Print Name Title (if applicable) MEMBER Print Name Title (if applicable)	1	2011	105/25/1-	7 -			1
Benjamin FISCHMAN MANAGING Print Name Title (if applicable) MEMBER Print Name Title (if applicable)	- 0	W		_			
Print Name Title (if applicable) MYM DV Print Name Title (if applicable)	Signature *	/ _	Date	Sig	nature		Date
Print Name Title (if applicable) MYM DV Print Name Title (if applicable)	B ~ .	1-10	Change Manage				
	DENTAM	IN LIZ					
	Print Name		Title (If applicable) MY	Y Del Prin	it Name		Title (if applicable)
		PIN nurr				PIN number fo	or electronic signature

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 94 of 102

Form **8821**

Tax Information Authorization

OMB No. 1545-1165 For IRS Use Only

		Telephone			
Internal Revenue Service					
1 Taxpayer Information. Taxpayer(s) must sign and date this		Data			
Taxpayer name(s) and address (type or print)	Taxpayer identification number	ər			
Affinity Health Care Management, Inc	harranana arang ar	-3680070			
1781 Highland Avenue, Suite 206 Cheshire, CT 06410	Daytime telephone number 203-250-2030	Plan number (if applicable)			
2 Appointee. If you wish to name more than one appointee, a		_1			
Name and address	CAF No.	-13			
	PTIN	***************************************			
	Telephone No.				
	Check If new: Address Te	lephone No. 🗌 Fax No. 🔲			
3 Tax matters. The appointee is authorized to inspect and/or	receive confidential tax information in	any office of the IRS for the			
tax matters listed on this line. Do not use Form 8821 to requ					
(a) (b) Type of Tax (Income, Employment, Excise, etc.) or Civil Penalty (1040, 941, 720, etc.)	(c) Year(s) or Period(s) (see the instructions for line 3)	(d) Specific Tax Matters (see instr.)			
4 Specific use not recorded on Centralized Authorization use not recorded on CAF, check this box. See the instruction 5 Disclosure of tax information (you must check a box on illn a lif you want copies of tax information, notices, and other basis, check this box	ns on page 4. If you check this box, sine 5a or 5b unless the box on line 4 is written communications sent to the	checked): appointee on an ongoing			
Retention/revocation of tax information authorizations. To authorizations for the same tax matters you listed on line 3 at to revoke a prior tax information authorization, you must at and check this box. To revoke this tax information authorization, see the instruction. Signature of taxpayer(s). If a tax matter applies to a joint research.	above unless you checked the box on tach a copy of any authorizations you	line 4. If you do not want u want to remain in effect			
corporate of lazypyet(a). If a lax matter applies to a point re- corporate officer, partner, guardian, executor, receiver, admit that I have the authority to execute this form with respect to	nistrator, trustee, or party other than t	he taxpayer, I certify			
► IF NOT SIGNED AND DATED, THIS TAX INFORMATION	AUTHORIZATION WILL BE RETUR	NED.			
► DO NOT SIGN THIS FORM IF IT IS BLANK OR INCOMP	LETE.				
De la de	h	i i			
1/2/10					
Sighature	Signature	Date			
Benjamin Fischman Pres					
Print Name Fiscandin Pres Title (if applicable)	Print Name	Title (if applicable)			

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 95 of 102

Form 8821 (Rev. October 2011) Department of the Treasury Internal Revenue Service	-9	OMB No. 1545-1165 For IRS Use Only Received by: Name Telephone Function Date						
1 Taxpayer informations 1 Taxpayer name(s) and address) must sign and date this for	The second residence of the second se	er identification num	har			
Health Care Investors, In		a Manor	raxpay		3-3646976			
55 Tunxis Avenue Bloomfield, CT 06002	ic. d/b/a Alexandii	a Warto	Daytime	s telephone number 860-242-0703		ımber (if applicable)		
2 Appointee. If you	wish to name mo	ore than one appointee, attac			-	+:		
Name and address 3 Tax matters. The	appointee Is auti	norized to inspect and/or rec	Telephone No Fax No. Check If new:	: Address 🔲 T	elephone N	No. Fax No. [
tax matters listed	on this line. Do n	ot use Form 8821 to request	copies of tax re	turns.				
(a) Type of Ta: (Income, Employment, or Clvll Pena	Excise, etc.)	(b) Tax Form Number (1040, 941, 720, etc.)		(c) or Period(s) uctions for line 3)	Specific 7	(d) pecific Tax Matters (see Instr.)		
use not recorded of 5 Disclosure of tax a if you want copie basis, check this b Note. Appointees b if you do not want 6 Retention/revoca authorizations for to revoke a prior t and check this box To revoke this tax	Information (you so of tax information will no longer reconstruction of tax information at the same tax main ax information authors.	intralized Authorization Files box. See the instructions of must check a box on line stion, notices, and other writers or communications and tices or communications. This titers you listed on line 3 about horization, you must attact or prization, see the instructions.	on page 4. If you can be a communicated in the related ment to your appoint to your appoint to your appoint a copy of any can be a copy of	the box on line 4 leations sent to the attended to the box of authorizations you are the common to the c	skip lines 5 s checked): appointee notices. ox tomatically in line 4. If	and 6 ▶ [on an ongoing ▶ [revokes all prior you do not want remain in effect ▶ [
corporate officer, p that I have the auti	partner, guardian, hority to execute AND DATED, T	natter applies to a joint return executor, receiver, administhis form with respect to the HIS TAX INFORMATION AND IS BLANK OR INCOMPLE	trator, trustee, o tax matters/per	or party other than rlods on line 3 abo	the taxpay	d by a ver, I certify	_	
Print Nasho	PIN numbe	Title (if applicable)	Print Name		PIN number fo	Title (If applicable)		

For Privacy Act and Paperwork Reduction Act Notice, see page 4.

Form 8821 (Rev. 10-2011)

Cat. No. 11596P

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 96 of 102

Form 8821		Tax Information				OMB No. 1545-1165 For IRS Use Only Received by: Name
(Rev. October 2011)	1333000000	Do not sign this form unless all applicable lines have been completed.				
Department of the Treasury	► Do	▶ Do not use this form to request a copy or transcript of your tax return. Instead, use Form 4506 or Form 4506-T.				Telephone
Internal Revenue Service						Date
		er(s) must sign and date this form	n on line			
Taxpayer name(s) and addre				Taxpayer identification nun		
Health Care Alliance in 612 Hazard Avenue	ic. d/b/a Blair Ma	nor			06-1401064	
Enfield, CT 06082				Daytime telephone number 860-749-8388	Plan	number (if applicable)
2 Appointed If yo	u wish to name	more than one appointee, attac	h a liet i			
Name and address	d Wish to Haine	more than one appointee, attac	CAF	lo		
Ivallie and address			PTIN			
				none No.		
			Fax N	***************************************		
			1.5.255000	*************************	T-1	No. Cl. For No. Cl
O Tay and the Th				if new: Address		
tax matters listed	e appointee is a d on this line. Do	authorized to inspect and/or rece o not use Form 8821 to request	copies	of tax returns.	in any ome	se of the IRS for the
(a)		(b)		(c)		(d)
Type of T (Income, Employment or Civil Pen	t, Excise, etc.)	Tax Form Number (1040, 941, 720, etc.)	(966	Year(s) or Period(s) the Instructions for line 3)	Specific	Tax Matters (see Instr.)
use not recorded	on CAF, check	Centralized Authorization File this box. See the instructions o	n page a or 5b	4. If you check this box, unless the box on line 4	skip Ilnes Is checked	5 and 6 ▶ □
basis, check this Note. Appointees	box s will no longer i	nation, notices, and other writt	 other re		notices.	ee on an ongoing
authorizations for to revoke a prior and check this be	r the same tax n tax information ox	ormation authorizations. This natters you listed on line 3 above authorization, you must attack thorization, see the instructions	ve unles n a copy	s you checked the box of any authorizations y	on line 4. li	f you do not want
corporate officer,	partner, guardi	c matter applies to a Joint return an, executor, receiver, administr te this form with respect to the	rator, tru	istee, or party other that	n the taxpa	ed by a lyer, I certify
► IF NOT SIGNE	D AND DATED	, THIS TAX INFORMATION AU	THORE	ZATION WILL BE RETU	JRNED.	
DO NOT SIGN	THIS FORM IF	IT IS BLANK OR INCOMPLET	E.			
1	7	1 noch to	7			1
- (//	//	1/19/1	<u> </u>		_	
Signature Signature	./	Date	Sig	nature		Date
Kenjamin F	schman	man President	_			
Print Name	— — .	Title (if applicable)	Prl	nt Name		Title (If applicable)
		nber for electronic signature				for electronic signature
or Privacy Act and Pap	erwork Reductio	n Act Notice, see page 4.		Cat. No. 11596P		Form 8821 (Rev. 10-2011)

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 97 of 102

OMB No. 1545-1165 Form 882 Tax Information Authorization For IRS Use Only > Do not sign this form unless all applicable lines have been completed. (Rev. October 2011) Telephone Do not use this form to request a copy or transcript of your tax return. Department of the Treasury Instead, use Form 4506 or Form 4506-T. Internal Revenue Service 1 Taxpayer information. Taxpayer(s) must sign and date this form on line 7. Taxpayer identification number Taxpayer name(s) and address (type or print) Health Care Rellance LLC d/b/a Ellis Manor 210 George Street Daytime telephone number Plan number (If applicable) Hartford, CT 06114 860-296-9166 2 Appointee. If you wish to name more than one appointee, attach a list to this form. CAF No. Name and address PTIN Telephone No. Fax No. Check if new: Address ☐ Telephone No. ☐ Fax No. ☐ Tax matters. The appointee is authorized to inspect and/or receive confidential tax information in any office of the IRS for the tax matters listed on this line. Do not use Form 8821 to request copies of tax returns. (a) Type of Tax (Income, Employment, Excise, etc.) or Civil Penalty (b) Tax Form Number Year(s) or Period(s) Specific Tax Matters (see instr.) (1040, 941, 720, etc.) (see the instructions for line 3) Specific use not recorded on Centralized Authorization File (CAF). If the tax information authorization is for a specific use not recorded on CAF, check this box. See the instructions on page 4. If you check this box, skip lines 5 and 6 5 Disclosure of tax information (you must check a box on line 5a or 5b unless the box on line 4 is checked): a If you want copies of tax information, notices, and other written communications sent to the appointee on an ongoing Note. Appointees will no longer receive forms, publications and other related materials with the notices. b If you do not want any copies of notices or communications sent to your appointee, check this box Retention/revocation of tax information authorizations. This tax information authorization automatically revokes all prior authorizations for the same tax matters you listed on line 3 above unless you checked the box on line 4. If you do not want to revoke a prior tax information authorization, you must attach a copy of any authorizations you want to remain in effect To revoke this tax Information authorization, see the instructions on page 4. Signature of taxpayer(s). If a tax matter applies to a joint return, either husband or wife must sign. If signed by a corporate officer, partner, guardian, executor, receiver, administrator, trustee, or party other than the taxpayer, I certify that I have the authority to execute this form with respect to the tax matters/periods on line 3 above. ▶ IF NOT SIGNED AND DATED, THIS TAX INFORMATION AUTHORIZATION WILL BE RETURNED. THIS FORM IF IT IS BLANK OR INCOMPLETE Date Signature Print Name Title (if applicable) Title (If applicable) Print Name

PIN number for electronic signature

PIN number for electronic signature

Exhibit B

Affinity Health Care Mgt Cash Flow Schedule Actual

(1) Beginning Month Cash On-Hand	1/14/2017	1/21/2017	1/28/2017	2/4/2017	2/11/2017
Weekly Fundings from Factor/Financing Company:				31743	
Weekly	314,000	318,000	320,000	320,000	320,000
Total Weekly Funding	314,000	318,000	320,000	320,000	320,000
Total Weekly Fullding	314,000	310,000	320,000	320,000	320,000
Insuarance Release 20% Release	- 255,000	87,500 -	100,000	:	165,000
DSS Grant of Pending Claim minus State of Ct Recoupments		-		-	-
Total Funding from Factor/Finance Co.	569,000	405,500	420,000	320,000	485,000
Misc Deposits	750	750	750	750	750
**Private Pay	120,000	100,000	120,000	100,000	120,000
Applied Income	15,000	•		99,000	15,000
(2) Total Receipts	704,750	506,250	540,750	519,750	620,750
	704,750	506,250	540,750	519,750	620,750
Cash Disbursements:					
Salary and Wages-Gross	233,000	236,000	243,000	243,000	233,000
Taxes	17,672	18,054	17,672	17,672	17,672
Accounts Payable Checks	34,000	41,500	42,500	36,000	34,000
Therapy Vendor	38,000	34,500	38,000	34,500	38,000
Pharmacy payments via wire	15,938	15,938	15,938	15,938	15,938
Provider tax	33,000	33,000	33,000	33,000	33,000
Health Insurance	10,000		177,500	2750	10,000
Utility and Deposits	3,750 6,685	11,750 6,685	19,750	3,750	3,750 6,685
Unemployment Tax Additonal Provider Tax Payment	6,665	6,665		-	0,005
Management Company Expenses	16,282	16.282	16,282	16,282	16,282
Workers Compensation Insurance	11,798	11,890	11,798	11,798	11,798
Working Capital Interest	34,999	-	-	•	34,999
RMS Professionals	22,500	8.78	-	-	22,500
Billing/AR Processing	25,998	-			25,998
US Trustee fees	2,475	2,475	2,475	2,475	2,475
Professional fees	15,000	15,000 24,000	15,000	15,000	15,000
Mortgage Reserve (a) Mortgage Payment thur Realty Co.	-	115,000	:	-	-
(3) Total Disbursements	521,096	582,073	632,914	429,414	521,096
to some or the some standard standard standard	521,096	582,073	632,914	429,414	521,096
Cash Surplus/(Shortfall)	183,654	(75,823)	(92,164)	90,336	99,654
Private Pay not received	-	•	1.0	•	
Private Pay received	129,817	313,472	237,649	145,485	235,821
Beginning Cash Balance Ending Cash Balance	313,472	237,649	145,485	235,821	335,475
Linding Cash Dalance	010,472	201,040	140,400	200,021	000,170

^{&#}x27;(a) Notwithstanding anything in the Tenth Interim Order (a) approving sale of provider accounts receivable to Revenue Management Solutions, LLC pursuant to §§ 363(b) and (f) of the bankruptcy code; (b) granting first-priority security interests on purchased accounts; (c) authorizing indebtedness with administrative super-priority and secured by llens on and security interests in non-purchased accounts and on all other assets of the providers pursuant to §§ 364(c) and (d) of the bankruptcy code; (d) authorizing use of cash collateral and (e) granting related relief (the "Order") to the contrary, the Debtors shall pay this line item within ten (10) days of entry of the Order.

Affinity Health Care Mgt Cash Flow Schedule Actual

Blair Manor (1) Beginning Month Cash On-Hand	44410047	1010017	1/00/0017	01110017	
Weekly Fundings from Factor/Financing Company:	1/14/2017	1/21/2017	1/28/2017	2/4/2017	2/11/2017
Weekly	95,000	95,000	97,000	97,000	97,000
Total Weekly Funding	95,000	95,000	97,000	97,000	97,000
Insuarance Release 20% Release DSS Grant of Pending Claim	80,000	87,500	20,000	-	50,000
minus State of Ct Recoupments Total Funding from Factor/Finance Co.	175,000	182,500	117,000	97,000	147,000
Misc Deposits **Private Pay Applied Income	250 35,000 5,000	250 35,000	250 55,000 -	250 35,000 37,000	250 35,000 5,000
(2) Total Receipts	215,250	217,750	172,250	169,250	187,250
Cash Disbursements:					
Salary and Wages-Gross	78,000	78,000	83,000	83,000	78,000
Taxes	5,814	5,967	5,814	5,814	5,814
Accounts Payable Checks	15,000	15,000	18,000	12,500	15,000
Therapy Vendor	12,000	11,440	12,000	11,440	12,000
Pharmacy payments via wire	5,313	5,313	5,313	5,313	5,313
Provider tax	11,000	11,000	11,000	11,000	11,000
Health Insurance Utility and Deposits	10,000 1,875	1,875	87,500 9,875	1,875	10,000
Unemployment Tax	1,070	1,070	3,070	1,070	1,070
Additional Provider Tax Payment Management Company Expenses	5,446	5,446	5,446	5,446	5,446
Workers Compensation Insurance	3,473	3,565	3,473	3,473	3,473
Working Capital Interest	11,667	0,000	1		11,667
RMS Professionals	7,500		1		7,500
Billing/AR Processing	8,666		- 1		8,666
US Trustee fees	825	825	825	825	825
Professional fees	5,000	5,000	5,000	5,000	5,000
Mortgage Reserve (a)Mortgage Payment thur Realty Co.		5,700 27,000	1	1.51	
(3) Total Disbursements	181,578	176,130	247,245	145,685	181,578
Cash Surplus/(Shortfall)	33,672	41,620	(74,995)	23,565	5,672
Private Pay not received		211, 22-1		7.0	
Private Pay received	198			-	-
Beginning Cash Balance	103,912	137,584	179,204	104,209	127,774
Ending Cash Balance	137,584	179,204	104,209	127,774	133,445

^{&#}x27;(a) Notwithstanding anything in the Tenth Interim Order (a) approving sale of provider accounts receivable to Revenue Management Solutions, LLC pursuant to §§ 363(b) and (f) of the bankruptcy code; (b) granting first-priority security interests on purchased accounts; (c) authorizing indebtedness with administrative super-priority and secured by liens on and security interests in non-purchased accounts and on all other assets of the providers pursuant to §§ 364(c) and (d) of the bankruptcy code; (d) authorizing use of cash collateral and (e) granting related relief (the "Order") to the contrary, the Debtors shall pay this line item within ten (10) days of entry of the Order.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 101 of 102

Affinity Health Care Mgt Cash Flow Schedule Actual

Douglas Manor (1) Beginning Month Cash On-Hand	1/14/2017	1/21/2017	1/28/2017	2/4/2017	2/11/2017
Weekly Fundings from Factor/Financing Company:					
Weekly	103,000	103,000	103,000	103,000	103,000
Total Weekly Funding	103,000	103,000	103,000	103,000	103,000
Insuarance Release		-	30,000		-
20% Release	85,000			*	55,000
DSS Grant of Pending Claim					
minus State of Ct Recoupments					
Total Funding from Factor/Finance Co.	188,000	103,000	133,000	103,000	158,000
Misc Deposits	250	250	250	250	250
**Private Pay	60,000	50,000	50,000	50,000	60,000
Applied Income	5,000		-	17,000	5,000
(2) Total Receipts	253,250	153,250	183,250	170,250	223,250
Cash Disbursements:					
Salary and Wages-Gross	77,000	77,000	82,000	82,000	77,000
Taxes	5,891	5,891	5,891	5,891	5,891
Accounts Payable Checks	10,000	17,000	14,500	14,500	
Therapy Vendor	14,000	12,500	14,000	12,500	14,000
Pharmacy payments via wire	5,313	5,313	5,313	5,313	5,313
Provider tax	11,000	11,000	11,000	11,000	11,000
Health Insurance		-	45,000	-	-
Utility and Deposits			8,000	-	-
Unemployment Tax Additional Provider Tax Payment	100	6,685	-	120	
Management Company Expenses	5,001	5,001	5,001	5,001	5,001
Workers Compensation Insurance	3,750	3,750	3,750	3,750	3,750
Working Capital Interest	11,666	TALST C	1075	530000	11,666
RMS Professionals	7,500				7,500
Billing/AR Processing	8,666				8,666
US Trustee fees	825	825	825	825	825
Professional fees	5,000	5,000	5,000	5,000	5,000
Mortgage Reserve (a)Mortgage Payment thur Realty Co.		12,100 58,000	1		-
(3) Total Disbursements	165,611	220,064	200,279	145,779	165,611
Cash Surplus/(Shortfall)	87,639	(66,814)	(17,029)	24,471	57,639
Private Pay not received	station in the state of the sta			83 AM NASSO	
Private Pay received				-	
Beginning Cash Balance	45,167	132,805	65,991	48,962	73,433
Ending Cash Balance	132,805	65,991	48,962	73,433	131,071

^{&#}x27;(a) Notwithstanding anything in the Tenth Interim Order (a) approving sale of provider accounts receivable to Revenue Management Solutions, LLC pursuant to §§ 363(b) and (f) of the bankruptcy code; (b) granting first-priority security interests on purchased accounts; (c) authorizing indebtedness with administrative super-priority and secured by liens on and security interests in non-purchased accounts and on all other assets of the providers pursuant to §§ 364(c) and (d) of the bankruptcy code; (d) authorizing use of cash collateral and (e) granting related relief (the "Order") to the contrary, the Debtors shall pay this line item within ten (10) days of entry of the Order.

Case 16-30043 Doc 541 Filed 01/13/17 Entered 01/13/17 16:07:14 Desc Main Document Page 102 of 102

Affinity Health Care Mgt Cash Flow Schedule Actual

(1) Beginning Month Cash On-Hand	1/14/2017	1/21/2017	1/28/2017	2/4/2017	2/11/2017
Weekly Fundings from Factor/Financing Company:					
Weekly	116,000	120,000	120,000	120,000	120,000
Total Weekly Funding	116,000	120,000	120,000	120,000	120,000
Total VVeekly Fullding	110,000	120,000	120,000	120,000	120,000
Insuarance Release	-	-	50,000	-	-
20% Release	90,000		-	-	60,000
DSS Grant of Pending Claim					
minus State of Ct Recoupments	2 5 5 5 2 5 5 5 5 6 6 6 6 5 5 5 6 6 6 6	P// Per 100 (Per 10)	SECTION AND L	150000000000000000000000000000000000000	POWER WEST-BOOK
Total Funding from Factor/Finance Co.	206,000	120,000	170,000	120,000	180,000
Misc Deposits	250	250	250	250	250
**Private Pay	25,000	15,000	15,000	15,000	25,000
Applied Income	5,000	,		45,000	5,000
(2) Total Receipts	236,250	135,250	185,250	180,250	210,250
Cash Disbursements:					
Salary and Wages-Gross	78,000	81,000	78,000	78,000	78,000
Taxes	5,967	6,197	5,967	5,967	5,967
Accounts Payable Checks	9,000	9,500	10,000	9,000	9,000
Therapy Vendor	12,000	10,560	12,000	10,560	12,000
Pharmacy payments via wire	5,313	5,313	5,313	5,313	5,313
Provider tax	11,000	11,000	11,000	11,000	11,000
Health Insurance	1 075	0.075	45,000	4 075	1 075
Utility and Deposits Unemployment Tax	1,875 6,685	9,875	1,875	1,875	1,875 6,685
Additional Provider Tax Payment	0,000	-	_	_	0,000
Management Company Expenses	5,835	5,835	5.835	5,835	5,835
Workers Compensation Insurance	4,575	4,575	4,575	4,575	4,575
Working Capital Interest	11,666	50,700,000,000			11,666
RMS Professionals	7,500	- 1			7,500
Billing/AR Processing	8,666				8,666
US Trustee fees	825	825	825	825	825
Professional fees	5,000	5,000 6,200	5,000	5,000	5,000
Mortgage Reserve (a)Mortgage Payment thur Realty Co.	ı	30,000		-	
(3) Total Disbursements	173,906	185,879	185,389	137,949	173,906
Cash Surplus/(Shortfall)	62,344	(50,629)	(139)	42,301	36,344
Private Pay not received	7.	(00,020)			
Private Pay received		*	-	-	-
Beginning Cash Balance	(19,262)	43,082	(7,547)	(7,686)	34,615
Ending Cash Balance	43,082	(7,547)	(7,686)	34,615	70,959

^{&#}x27;(a) Notwithstanding anything in the Tenth Interim Order (a) approving sale of provider accounts receivable to Revenue Management Solutions, LLC pursuant to §§ 363(b) and (f) of the bankruptcy code; (b) granting first-priority security interests on purchased accounts; (c) authorizing indebtedness with administrative super-priority and secured by liens on and security interests in non-purchased accounts and on all other assets of the providers pursuant to §§ 364(c) and (d) of the bankruptcy code; (d) authorizing use of cash collateral and (e) granting related relief (the "Order") to the contrary, the Debtors shall pay this line item within ten (10) days of entry of the Order.